HOCHSCHILD AND CYCLIC HOMOLOGY FOR BORNOLOGICAL COARSE SPACES



DISSERTATION ZUR ERLANGUNG DES DOKTORGRADES DER NATURWISSENSCHAFTEN (DR. RER. NAT.) DER FAKULTÄT FÜR MATHEMATIK DER UNIVERSITÄT REGENSBURG

vorgelegt von

LUIGI CAPUTI

aus Mesagne (Italien)

im Jahr 2019

Promotions gesuch eingereicht am: 02.05.2019

Die Arbeit wurde angeleitet von: Prof. Dr. Ulrich Bunke

Prüfungsausschuss:	Vorsitzender:	Prof. Dr. Helmut Abels
	1. Gutachter:	Prof. Dr. Ulrich Bunke
	2. Gutachter:	Prof. Dr. Denis-Charles Cisinski
	weiterer Prüferin:	Prof. Dr. Clara Löh

Summary

The main goal of the thesis is to construct equivariant coarse versions of the classical Hochschild and cyclic homologies of algebras. These are lax symmetric monoidal functors from the category of equivariant bornological coarse spaces to the cocomplete stable ∞ -category of chain complexes and are called *equivariant coarse Hochschild and cyclic homology*. If k is a field, the evaluation at the one point bornological coarse space induces equivalences with the classical Hochschild and cyclic homology of (a canonical bornological coarse space associated to) the group G agrees with the classical Hochschild (cyclic) homology of the associated group algebra k[G].

The second aim of the thesis is the construction of natural transformations from equivariant coarse algebraic K-homology to equivariant coarse Hochschild and cyclic homology, and of natural transformations from equivariant coarse Hochschild and cyclic homology to equivariant coarse ordinary homology. This is achieved by using trace-like maps and gives a natural transformation from equivariant coarse algebraic K-homology to equivariant coarse ordinary homology.

We conclude the dissertation with two additional investigations: we give a comparison result between the forget-control map for equivariant coarse Hochschild homology and the associated generalized assembly map, and we show a Segal-type localization theorem for equivariant Hochschild and cyclic homology. iv

Contents

In	trod	uction	vii
1	1 Bornological coarse spaces and coarse homology theories		
	1.1	<i>G</i> -Equivariant bornological coarse spaces	2
	1.2	Equivariant coarse homology theories	9
	1.3	Forget-control and assembly maps	13
	1.4	Localization theorems for coarse homology theories	19
	1.5	Equivariant coarse ordinary homology	26
2	The	symmetric monoidal category of controlled objects	29
	2.1	The category of controlled objects	30
	2.2	The symmetric monoidal refinement of $V_{\mathbf{A}}^G$	36
	2.3	Equivariant coarse algebraic K-homology	44
3	3 A coarse version of Hochschild and cyclic homology		47
	3.1	The category of mixed complexes	48
	3.2	Keller's cyclic homology of exact categories	53
	3.3	The equivariant coarse homology theory $\widetilde{\mathcal{XC}}_{k}^{G}$	57
	3.4	Coarse Hochschild and coarse cyclic homology	63
	3.5	A symmetric monoidal refinement of $\mathcal{X}\mathrm{HH}_k^G$	68
4	A ti	ransformation to coarse ordinary homology and further properties	71
	4.1	Comparison results	72
	4.2	The forget-control map for coarse Hochschild homology	74
	4.3	A transformation to coarse ordinary homology	76
	4.4	A transformation from coarse algebraic K-homology	83
	4.5	A Segal-type localization theorem for coarse Hochschild homology	86
\mathbf{A}	ppen	dix A	89
	A.1	Additive categories	91
	A.2	Differential graded categories	94
	A.3	Symmetric monoidal structures	98

A.4 Cyclic objects and the additive cyclic nerve	
References	105
Acknowledgements	110

Introduction

Coarse geometry is the study of metric spaces from a large-scale point of view, meaning that one wants to capture the behavior at infinity of metric spaces, discrete groups or more general spaces. The general principle, in coarse geometry, is that two spaces are considered equivalent if they "look the same from great distances". This viewpoint has lead Roe to the abstract notion of *coarse spaces*, objects that encode the large-scale geometry and properties of metric spaces [Roe93, Roe96, Roe03].

In order to apply homological methods in coarse geometry and perform index theory on non-compact manifolds, a large-scale analogue of ordinary (co-)homology has also been introduced by Roe and has been called *coarse (co-)homology*. This coarse cohomology theory, together with a coarse version of topological K-theory, provided new invariants and have applications in index theory, homotopy theory, K-theory of C^* -algebras and, maybe most importantly, in studying assembly map conjectures [Roe93, Mit01, BEKW18]. Among the applications of coarse geometry to other mathematical fields there is, *e.g.*, Mostow's proof of his rigidity theorem [Mos73]. For further references and a survey on coarse geometry we refer to [Roe03].

A new axiomatic and homotopic approach to coarse geometry and coarse algebraic topology has been recently developed by Bunke and Engel [BE16], and then generalized to the equivariant framework by Bunke, Engel, Kasprowski and Winges [BEKW17]. In their set-up, the main objects are called *bornological coarse spaces*: these are triples $(X, \mathcal{C}, \mathcal{B})$ consisting of a set X, equipped with a bornology \mathcal{B} (describing the bounded subsets of X) and a coarse structure \mathcal{C} (a family of neighborhoods of the diagonal in $X \times X$ encoding the large-scale properties of X), with bornology and coarse structure being compatible in a suitable way [BE16, Def. 2.1]. Every metric space is a bornological coarse space in a canonical way, and, more generally, a coarse space, as defined by Roe, is a coarse space in this framework.

Let G be a group. Among different invariants of (G-equivariant) bornological coarse spaces, *i.e.*, bornological coarse spaces with a G-action by automorphisms, there are (equivariant) coarse homology theories, *i.e.*, functors

$E \colon G\mathbf{BornCoarse} \to \mathbf{C}$

from the category of G-bornological coarse spaces GBornCoarse to a cocomplete sta-

ble ∞ -category **C** (*e.g.*, the ∞ -categories of chain complexes or spectra), satisfying some additional axioms: coarse invariance, flasqueness, coarse excision and u-continuity [BEKW17, Def. 3.10]. Examples of coarse homology theories arise as coarsifications of locally finite homology theories [BE16]. Among others, there are coarse versions of ordinary homology and of topological K-theory [BE16], of equivariant algebraic Khomology and of topological Hochschild homology [BEKW17, BC17], and of Waldhausen's A-theory [BKW18].

This new homotopic approach to coarse geometry has brought new applications and insight (for example, the dualizing spectrum of a group is now known to be a coarse invariant [BE17b]), and it is now developing as a new field of research [BC17, BE17a, BE17b, BE17c, Bun18, BKW18].

Hochschild and cyclic homology for bornological coarse spaces

Classically, Hochschild and cyclic homology are homological invariants of algebras [Lod98], and have been extended to dg-algebras, schemes and, more generally, additive categories and exact categories [McC94, Kel99]. The goal of this thesis is to construct (equivariant) Hochschild and cyclic homology theories

$\mathcal{X}\mathrm{HH}_k^G : G\mathbf{BornCoarse} \to \mathbf{Ch}_\infty \quad \text{and} \quad \mathcal{X}\mathrm{HC}_k^G : G\mathbf{BornCoarse} \to \mathbf{Ch}_\infty$

(as described below) for bornological coarse spaces within the homotopic framework developed by Bunke and Engel [BE16, BEKW17].

Our main motivation comes from equivariant coarse algebraic K-homology [BEKW17, BC17], as we now explain. Various versions of algebraic K-theory come equipped with trace maps (e.g., the Dennis trace map from algebraic K-theory of rings to Hochschild homology, or the refined version, the cyclotomic trace, from the algebraic K-theory spectrum to the topological cyclic homology spectrum) and these trace maps have been of fundamental importance in the understanding of algebraic K-theory; see, for example, [DGM13]. A version of algebraic K-theory for bornological coarse spaces has recently also been defined [BEKW17]; hence, inspired by the classical case, our aim is to define coarse versions of Hochschild and cyclic homology and to define trace maps from equivariant coarse algebraic K-homology to equivariant coarse Hochschild and cyclic homology.

We now present the main result of the thesis. Let k be field and let G be a group. The symbols HH and HC refer to Hochschild and cyclic homology; we denote by C_*^{HH} and C_*^{HC} the chain complexes computing Hochschild homology and cyclic homology (of k-algebras) respectively. We deonte by $\{*\}$ the one point bornological coarse space; the G-equivariant bornological coarse space $G_{\text{can,min}}$ denotes a canonical equivariant bornological coarse space space associated to the group G (see Example 1.1.22).

The main result of the dissertation is the following theorem (see Definition 3.4.6,

Theorem 3.4.7, Proposition 4.1.3 and Proposition 4.1.4):

Theorem. There are functors

 $\mathcal{X}\mathrm{HH}_k^G \colon G\mathbf{BornCoarse} o \mathbf{Ch}_\infty \quad and \quad \mathcal{X}\mathrm{HC}_k^G \colon G\mathbf{BornCoarse} o \mathbf{Ch}_\infty$

from the category of G-equivariant bornological coarse spaces to the cocomplete stable ∞ -category \mathbf{Ch}_{∞} of chain complexes, satisfying the following properties:

- (i) $\mathcal{X}HH_k^G$ and $\mathcal{X}HC_k^G$ are G-equivariant coarse homology theories;
- (ii) there are equivalences of chain complexes

$$\mathcal{X}\mathrm{HH}_{k}^{G}(*) \simeq C_{*}^{\mathrm{HH}}(k) \quad and \quad \mathcal{X}\mathrm{HC}_{k}^{G}(*) \simeq C_{*}^{\mathrm{HC}}(k)$$

between the evaluations of $\mathcal{X}HH_k^G$ and $\mathcal{X}HC_k^G$ at the one point bornological coarse space $\{*\}$, endowed with the trivial G-action, and the chain complexes computing Hochschild and cyclic homology of k;

(iii) there are equivalences

$$\mathcal{X}\mathrm{HH}_{k}^{G}(G_{\mathrm{can,min}}) \simeq C^{\mathrm{HH}}_{*}(k[G];k) \quad and \quad \mathcal{X}\mathrm{HC}_{k}^{G}(G_{\mathrm{can,min}}) \simeq C^{\mathrm{HC}}_{*}(k[G];k)$$

of chain complexes between the evaluations at the G-bornological coarse space $G_{\text{can,min}}$ and the chain complexes computing Hochschild and cyclic homology of the k-algebra k[G].

The functors $\mathcal{X}HH_k^G$ and $\mathcal{X}HC_k^G$ are called *equivariant coarse Hochschild homology* and *equivariant coarse cyclic homology* respectively. The names are justified by the use of a suitable Hochschild homology (and cyclic homology) functor in their construction, and by the comparison results: the evaluation at the one point space and at the (canonical *G*-bornological coarse space associated to the) group *G* agree with the classical Hochschild and cyclic homologies of the base field *k* and of the group algebra k[G].

Our construction of the functors $\mathcal{X}HH_k^G$ and $\mathcal{X}HC_k^G$ uses a cyclic homology theory for additive exact categories that satisfies certain additive and localizing properties in the sense of Tabuada [Tab07]; see also [CT11]. This is *Keller's cone construction* for exact categories [Kel99]

$$C: \mathbf{Ex} \to \mathbf{Mix},\tag{0.0.1}$$

which is a functor from the category \mathbf{Ex} of small exact categories to Kassel's category **Mix** of mixed complexes [Kas87]. As shown by Kassel, Hochschild and cyclic homologies can be defined in terms of mixed complexes, consistently with the classical definitions for k-algebras [Lod98]. Hochschild and cyclic homologies of the Keller's mixed complex $C(\mathcal{A})$, associated to a small exact category \mathcal{A} , will be called *Keller's Hochschild and cyclic homology of* \mathcal{A} .

To every G-bornological coarse space X, we can associate a suitable k-linear category; this is the k-linear category $V_k^G(X)$ of G-equivariant X-controlled (finite dimensional) k-vector spaces ([BEKW17], or Definition 2.1.2). In fact, this correspondence is functorial in X, and describes a functor

$V_k^G : G\mathbf{BornCoarse} \to \mathbf{Cat}_k$

from the category of G-bornological coarse spaces to the category Cat_k of small k-linear categories [BEKW17]. The k-linear category $V_k^G(X)$ is then equipped with the exact structure given by the short split exact sequences. We define equivariant coarse Hochschild (and cyclic) homology $\mathcal{X}\operatorname{HH}_k^G$ and $\mathcal{X}\operatorname{HC}_k^G$, as Keller's Hochschild and cyclic homology of the k-linear category $V_k^G(X)$ equipped with the exact structure given by the short split exact sequences. In Theorem 3.4.2, we show that these compositions of functors satisfy the axioms describing a coarse homology theory.

Keller's functor C satisfies several properties, like an additivity and a localization property [Kel99]. We review them in Section 3.2. The localization property [Kel99, Thm. 1.5], in particular, is fundamental to us in order to prove coarse excision (Theorem 3.3.8), one of the axioms defining a coarse homology theory. A simpler definition of Hochschild and cyclic homologies for exact k-linear categories is due to McCarthy: his definition uses the *additive cyclic nerve* CN(\mathcal{A}) of the k-linear category \mathcal{A} [McC94]. However, the homology theories defined by McCarthy do not (a priori) satisfy the localization property [Kel99, Example 1.8, Example 1.9]: a cyclic homology functor satisfying localization has to take also negative values, and McCarthy's Hochschild and cyclic homologies are only positively graded. However, when restricted to additive categories with the split exact structure, Keller's and McCarthy's definitions of Hochschild and cyclic homologies are equivalent (see also Lemma 3.4.4 and Remark 3.4.5 for a discussion about it). This has led us to define two intermediate coarse homology theories

$$\mathcal{X}\mathrm{Mix}_k^G \colon G\mathbf{BornCoarse} \xrightarrow{V_k^G} \mathbf{Cat}_k \xrightarrow{\mathrm{Mix}} \mathbf{Mix} \xrightarrow{loc} \mathbf{Mix}_\infty$$

of Definition 3.4.1 and

$$\widetilde{\mathcal{XC}}_k^G : G \mathbf{BornCoarse} \xrightarrow{V_k^G} \mathbf{Cat}_k \longrightarrow \mathbf{Ex}_k \xrightarrow{C} \mathbf{Mix} \xrightarrow{loc} \mathbf{Mix}_{\infty}$$

of Definition 3.3.1; here, the category $\operatorname{Mix}_{\infty}$ is Kassel's ∞ -category of mixed complexes, the functor C is Keller's cone construction (0.0.1) and the functor Mix is the functor that associates a canonical mixed complex to the additive cyclic nerve of a k-linear category. Finally, the functor $\operatorname{Cat}_k \to \operatorname{Ex}_k$ sends a k-linear category to the same k-linear category equipped with the split exact structure.

The functors $\mathcal{X}\operatorname{Mix}_k^G$ and $\widetilde{\mathcal{X}C}_k^G$ provide equivalent coarse homology theories. Coarse

Hochschild and cyclic homology are then defined by post-composition of Hochschild and cyclic homology for mixed complexes with either $\mathcal{X}\operatorname{Mix}_k^G$ or $\widetilde{\mathcal{XC}}_k^G$; in this thesis, we decide to use the functor $\mathcal{X}\operatorname{Mix}_k^G$ (see Section 3.4), but we prove most of the theorems by using the functor $\widetilde{\mathcal{XC}}_k^G$ (see Remark 3.3.3).

Trace-like maps

The construction of equivariant coarse Hochschild homology follows the same ideas of [BEKW17], where the equivariant coarse algebraic K-homology $\mathcal{X}\mathbf{A}K^G$ (with values in an additive category \mathbf{A} with strict G-action) has been defined as the non-connective K-theory of the additive category $V^G_{\mathbf{A}}(X)$ of G-equivariant X-controlled \mathbf{A} -objects (see Definition 2.1.2). When \mathbf{A} is the category of finitely dimensional k-vector spaces, we denote by $K\mathcal{X}^G_k$ the corresponding G-equivariant coarse K-homology functor. By using an extension to exact categories of the classical Dennis trace map we get the following (see Proposition 4.4.1):

Theorem. There are natural transformations of equivariant coarse homology theories

$$\mathrm{K}\mathcal{X}_k^G o \mathcal{X}\mathrm{HH}_k^G \quad and \quad \mathrm{K}\mathcal{X}_k^G o \mathcal{X}\mathrm{HC}_k^G$$

induced by the classical trace maps from algebraic K-theory to Hochschild and cyclic homology.

Equivariant coarse ordinary homology

$$\mathcal{X}C^G \colon G\mathbf{BornCoarse} o \mathbf{Ch}_\infty$$

is defined in terms of equivariant locally finite controlled maps $X^{n+1} \to k$ (see Definition 1.5.1), where X is a G-bornological coarse space and k is the field of coefficients. We review its definition in Section 1.5.

The second main result of the thesis is the construction of a natural transformation from equivariant coarse Hochschild homology to equivariant coarse ordinary homology. We prove the following result (see Theorem 4.3.8 and Proposition 4.3.9):

Theorem. There is a natural transformation

$$\Phi_{\mathcal{X}\mathrm{HH}_{k}^{G}} \colon \mathcal{X}\mathrm{HH}_{k}^{G} \to \mathcal{X}C^{G}.$$

of G-equivariant \mathbf{Ch}_{∞} -valued coarse homology theories, which induces an equivalence of chain complexes when evaluated at the one point space $\{*\}$.

The transformation $\Phi_{\mathcal{X}HH_k^G} \colon \mathcal{X}HH_k^G \to \mathcal{X}C^G$ is defined by using trace-like maps and extends to coarse cyclic homology as well. The idea, developed in Section 4.3, is the following.

We start by considering the additive cyclic nerve of the k-linear category $V_k^G(X)$ of *G*-equivariant X-controlled finite dimensional k-vector spaces. The *n*-th component of the additive cyclic nerve of a k-linear category C is

$$CN_n(\mathcal{C}) = \bigoplus_{(C_0,\dots,C_n)} \left(\bigotimes_{i=0}^n Hom_{\mathcal{C}}(C_{i+1},C_i) \right)$$

where the index *i* runs cyclically in the ordered set $(0, \ldots, n)$ and the sum ranges over all the tuples (C_0, \ldots, C_n) of objects of \mathcal{C} . When \mathcal{C} is the *k*-linear category $V_k^G(X)$, we let $A_0 \otimes A_1 \otimes \cdots \otimes A_n$, with $A_i: M_{i+1} \to M_i$, be an element in the *n*-th component of the additive cyclic nerve $\operatorname{CN}_n(V_k^G(X))$ and (x_0, \ldots, x_n) be a point of X^{n+1} . There are well-defined evaluations $M_i(x_j)$ of the object $M_i \in V_k^G(X)$ at the point $x_j \in X$ and there are induced linear operators $A_i^{x_i, x_{i+1}}: M_i(x_i) \to M_{i+1}(x_{i+1})$ between finite dimensional *k*-vector spaces. We consider the following composition:

$$M_0(x_n) \xrightarrow{A_n^{x_n, x_{n-1}}} M_n(x_{n-1}) \xrightarrow{A_{n-1}^{x_{n-1}, x_{n-2}}} \dots \xrightarrow{A_1^{x_1, x_0}} M_1(x_0) \xrightarrow{A_0^{x_0, x_n}} M_0(x_n), \quad (0.0.2)$$

that is an endomorphism of $M_0(x_n)$ (a finite dimensional k-vector space). To every tuple (A_0, \ldots, A_n) and every (n + 1)-tuple (x_0, \ldots, x_n) of points of X, we can associate an element of the field k, that is the trace of the composition (0.0.2). We show that this describes an equivariant coarse n-chain of X and extends linearly to a chain map defined on the additive cyclic nerve $CN(V_k^G(X))$). It extends to the canonical mixed complex associated to $CN(V_k^G(X))$, and this extension yields a natural transformation $\Phi_{\mathcal{X}HH_k^G} : \mathcal{X}HH_k^G \to \mathcal{X}C^G$ (see Theorem 4.3.8). The computation at the one point space induces an equivalence of chain complexes and this implies that the transformation is not trivial (as the equivariant coarse ordinary homology of the point is non-zero).

By composition of these two natural transformations, we get a natural transformation

$$K\mathcal{X}_k^G \to \mathcal{X}\mathrm{HH}_k^G \to \mathcal{X}\mathrm{H}_k^G$$

from equivariant coarse algebraic K-homology to (the spectra-valued) equivariant coarse ordinary homology. We believe that the study of this transformation may be useful for the understanding and detection of coarse K-theory classes.

Assembly maps and localization results

One of the main applications of coarse algebraic topology is in the study of assembly map conjectures, like the Farrell-Jones and the Baum-Connes assembly map conjectures. If \mathcal{F} is a family of subgroups of a group G (meaning that it is a collection of subgroups of G closed under conjugation and under taking subgroups), and if GOrb denotes the orbit category (*i.e.*, the category of transitive G-sets and G-equivariant maps), then we can consider the full subcategory $GOrb_{\mathcal{F}}$ of GOrb given by the G-sets whose stabilizers belong to \mathcal{F} . The assembly map for an equivariant homology theory (assumed to be defined on GOrb), with respect to the family of subgroups \mathcal{F} , is the map induced (after taking colimits) in homology by the inclusion $GOrb_{\mathcal{F}} \to GOrb$; roughly, it approximates the starting functor with its restrictions in $GOrb_{\mathcal{F}}$ [Lue18]. The Farrell-Jones conjecture, for example, asserts that the assembly map for algebraic K-theory with respect to the family of virtually cyclic subgroups is an isomorphism, and this conjecture has many connections to other outstanding conjectures, like the Borel conjecture, the Kaplansky conjecture or the Novikov conjecture. We refer to [BLR07, RV17, Lue18] for recent surveys on the subject.

If $E: GBornCoarse \to C$ is a *G*-equivariant coarse homology theory, then the composition

$$E \circ \mathcal{O}_{hlg}^{\infty} \colon G\mathbf{Top} \to \mathbf{C}$$

with a suitable cone functor $\mathcal{O}_{hlg}^{\infty}$ [BEKW17], describes a *G*-equivariant homology theory. Furthermore, one can associate to *E* a fiber sequence of functors, called a *cone sequence* [BEKW17, Def. 11.9] whose boundary map is called the *forget-control* map. In the coarse approach to the assembly map conjectures, one studies the forget-control map of a coarse homology theory *E*, trying to infer properties of the associated *G*-homology theory $E \circ \mathcal{O}_{hlg}^{\infty}$. Under some mild assumptions, the assembly map for this *G*-homology theory is equivalent to the forget control map for *E* [BEKW17, BEKW18]; see [BKW18] for a recent application of these methods to the split-injectivity of the assembly map in Waldhausen *A*-theory.

We apply a comparison result (between the forget-control map and the assembly map [BEKW17, Thm. 11.16]) to equivariant coarse Hochschild and cyclic homology as well. Unfortunately, we do not know (and we do not expect) that coarse Hochschild and cyclic homologies are additive coarse homology theories, hence we cannot apply directly the more general results on the split-injectivity of the associated assembly map [BEKW18, Thm. 1.11]. However, we show (see Proposition 4.2.9) that, when restricted to the family **Fin** of finite subgroups of G, the forget-control map for G-equivariant coarse Hochschild (cyclic) homology agrees with the assembly map for the associated G-equivariant homology theory $\mathbf{HH}_k^G := \mathcal{X} \mathrm{HH}_k^G \circ \mathcal{O}_{\mathrm{hlg}}^{\infty}$:

Theorem. The forget-control map for $\mathcal{X}HH_k^G$ is equivalent to the assembly map for the *G*-homology theory \mathbf{HH}_k^G .

Our last application is inspired by Segal's Localization Theorem [Seg68a], that we briefly recall.

Atiyah gives a definition of G-equivariant K-theory K_G^* of a G-space X in terms of G-vector bundles [Ati64, Seg68a]. From this definition, the evaluation at the one point space $\{*\}$ induces an isomorphism of rings between $K_G^0(*)$ and the representation ring

(called *character ring* therein) R(G) of G; hence the G-equivariant K-theory $K_G^*(X)$ of a G-space X has the structure of an R(G)-algebra over the representation ring because every space has a natural map onto a point.

Segal's Localization Theorem [Seg68a, Prop. 4.1] says that, after localization at a certain prime ideal \mathfrak{p} of R(G) (that is associated to a conjugacy class γ of G), the inclusion $X^{\gamma} \to X$ of the subspace of γ -fixed points X^{γ} in X induces (for locally compact G-spaces) an isomorphism

$$K^*_G(X)_{\mathfrak{p}} \to K^*_G(X^{\gamma})_{\mathfrak{p}}$$

in equivariant K-theory.

Inspired by Segal's Localization Theorem, general localization results for equivariant coarse homology theories, with the goal to develop new tools for studying equivariant coarse algebraic K-theory, have been shown [BCb]. In this thesis, we apply these general coarse localization results to (the lax-symmetric monoidal refinements of) equivariant coarse Hochschild and cyclic homology, obtaining the following localization theorem for the associated G-equivariant Hochschild homology (see Corollary 4.5.7):

Theorem. Let G be a finite group and γ be a conjugacy class of G. Let W be a finite G CW-complex and let W^{γ} be the sub-complex of γ -fixed points. Then, the inclusion $W^{\gamma} \to W$ induces an equivalence

$$\mathbf{HH}^{G}_{(\gamma)}(W^{\gamma}) \to \mathbf{HH}^{G}_{(\gamma)}(W)$$

of chain complexes.

Organization of the thesis

The thesis is structured as follows.

In Chapter 1 we recall and study the category GBornCoarse of G-bornological coarse spaces and its invariants, the G-equivariant coarse homology theories. This is done in Section 1.1 and Section 1.2, where we also introduce the necessary notation and terminology for the subsequent sections and chapters. In Section 1.3, we define the forget-control map for a G-equivariant coarse homology theory. We introduce the notions of G-uniform bornological coarse spaces and of the cone functor; the cone functor in particular is important in the study of assembly map conjectures because it provides a bridge between G-homology theories and G-equivariant coarse homology theories. We then describe the general comparison result between the forget-control maps and the usual assembly maps [BEKW17]. In Section 1.4, we describe the general coarse localization theorems [BCb]. The last section, Section 1.5, is devoted to a first example of coarse homology theories: this is coarse ordinary homology.

In Chapter 2, we study in detail the symmetric monoidal category $V^G_{\mathbf{A}}(X)$ of *G*-equivariant *X*-controlled **A**-objects, where *X* is a *G*-bornological coarse space and **A** denotes an additive category with strict *G*-action. In Section 2.1, we analyze this category and the associated functor $V^G_{\mathbf{A}}: G\mathbf{BornCoarse} \to \mathbf{Add}$ from *G*-bornological coarse spaces to the category of small additive categories. Furthermore, we study the behavior of this functor under coarse equivalences, colimits and flasque spaces [BEKW17]. In Section 2.2, we discuss the behavior of the functor $V^G_{\mathbf{A}}$ under products and we see that it admits a lax symmetric monoidal refinement [BCa]. In the last section, Section 2.3, we recall the definition of equivariant coarse algebraic *K*-homology, which is defined as the non-connective *K*-theory of the category $V^G_{\mathbf{A}}(X)$ [BEKW17].

In *Chapter 3*, we construct coarse versions of the classical Hochschild and cyclic homologies of algebras as functors

$\mathcal{X}\mathrm{HH}_k^G : G\mathbf{BornCoarse} \to \mathbf{Ch}_\infty \quad \mathrm{and} \quad \mathcal{X}\mathrm{HC}_k^G : G\mathbf{BornCoarse} \to \mathbf{Ch}_\infty$

from the category of G-bornological coarse spaces to the cocomplete stable ∞ -category of chain complexes. In order to define these functors, we use Keller's cyclic homology [Kel99] of additive exact categories, which is recalled in Section 3.1 and Section 3.2. Section 3.3 and Section 3.4 are the core of the thesis, where we prove that, in complete analogy to the construction of coarse algebraic K-homology of [BEKW17], the composition of the functor V_k^G and Keller's cone functor provide a coarse homology theory. In Section 3.5, we prove that equivariant coarse Hochschild and cyclic homologies refine to lax symmetric monoidal functors.

In *Chapter 4*, we describe some features and properties of the coarse homology theories defined in Chapter 3. In Section 4.1, we see that coarse Hochschild and cyclic homologies do extend the classical Hochschild and cyclic homologies for algebras: we prove that the evaluation at the one point space is equivalent to the usual Hochschild homology of the

base field, and that the evaluation at (a canonical G-bornological coarse space associated to) G is equivalent to the classical Hochschild homology of the group algebra k[G]. In Section 4.2, we study the forget-control map for coarse Hochschild and cyclic homology, by using the general results reviewed in Section 1.3 and we see that this forget-controlling is equivalent to the assembly map for the G-equivariant homology theory associated to coarse Hochschild and cyclic homology. In Section 4.3 and Section 4.4, we construct the trace-like natural transformations from coarse algebraic K-homology to coarse Hochschild homology, and from coarse Hochschild homology to coarse ordinary homology. In the last section, Section 4.5, we apply the general localization theorems for equivariant coarse homology theories, reviewed in Section 1.4, to the case of coarse Hochschild and cyclic homology.

Finally, in the Appendix, we mainly recollect some useful facts concerning differential graded categories.

Conventions

In this thesis we freely employ the language of ∞ -categories. More precisely, we model ∞ -categories as quasi-categories [Cis, Lur09, Lur14]. Our main reference on dg-categories is Keller's ICM [Kel06].

The results of this work are stated for small categories. However, the categories of bornological coarse spaces, the category of dg-categories and dg-modules, etc., are not small. There are classical ways to overcome size problems; we assume the Grothendieck theory of nested universes.

In the thesis, G will usually denote a group, k a field and the symbol \otimes denotes tensor products over k. We assume that 0 belongs to the natural numbers and that $\{*\}$ is the one point space.

Wandrers Nachtlied

Über allen Gipfeln Ist Ruh; In allen Wipfeln Spürest du Kaum einen Hauch; Die Vögelein schweigen im Walde. Warte nur, balde Ruhest du auch.

Goethe

xviii

Chapter 1

Bornological coarse spaces and coarse homology theories

A bornological coarse space is a set endowed with a bornology and a coarse structure (see Definition 1.1.11). When a group G acts on such a space by automorphisms, we have the notion of a G-equivariant bornological coarse space.

Following our main references [BE16, BEKW17], in this chapter we define the category GBornCoarse of G-equivariant bornological coarse spaces and we recall the definition of equivariant coarse homology theories. These are functors

$E: GBornCoarse \to C$

from the category of G-bornological coarse spaces to some cocomplete stable ∞ -category **C** satisfying additional axioms, as reviewed in Definition 1.2.1. We then recollect some general properties and results about equivariant coarse homology theories, focusing in particular on forget-control and assembly maps [BEKW17, BEKW18] and on Segal-type localization theorems [Seg68a].

The chapter is organized as follows. In Section 1.1, we give the definitions of bornological coarse spaces and G-bornological coarse spaces, introducing notations and terminology for the subsequent sections and chapters. In Section 1.2 we study the (equivariant) coarse homology theories and we introduce the notions of strongness and continuity, necessary for studying coarse assembly map conjectures.

In Section 1.3 we focus on forget-control and coarse assembly maps for equivariant coarse homology theories. Following [BEKW17], we give a comparison result between these different maps, showing that in some special cases they are equivalent. We introduce the necessary background and the core definition of *cone functor* \mathcal{O}^{∞} . In Section 1.4, following [BCb], we focus on general localization theorems for coarse homology theories.

We conclude the chapter with a very first example of coarse homology theories: equivariant coarse ordinary homology.

1.1 *G*-Equivariant bornological coarse spaces

In this section we give the fundamental definition of G-bornological coarse spaces (see Definition 1.1.21). The main source of examples is given by metric spaces, for which the bornology is the family of bounded sets and the coarse structure is generated by the family of controlled sets (see Example 1.1.12).

We denote by (G)**BornCoarse** the category of (G-)bornological coarse spaces and (G-equivariant) proper controlled maps (see Definition 1.1.3 and Definition 1.1.6). After reviewing some basic definitions, namely Definition 1.1.24–1.1.27, we describe the canonical symmetric monoidal structure on G**BornCoarse**.

Let X be a set and let $\mathcal{P}(X)$ denote its power set.

Definition 1.1.1. [BE16, Def 2.1] A *bornology* on a set X is a subset $\mathcal{B} \subseteq \mathcal{P}(X)$ such that:

- *B* is closed under taking subsets;
- \mathcal{B} is closed under taking finite unions;
- the set X is covered by elements of \mathcal{B} : $X = \bigcup_{B \in \mathcal{B}} B$.

A bornological space is a pair (X, \mathcal{B}) where \mathcal{B} is a bornology on the set X.

The elements B in \mathcal{B} are called \mathcal{B} -bounded sets, or simply bounded sets when the given bornology is clear from the context. Intersection of bornologies on a set X is again a bornology on X; if A is a subset of X, the bornology generated by A is the minimal bornology on X containing A. This bornology is denoted by $\mathcal{B}\langle A \rangle$.

Notation 1.1.2. In order to emphasize the dependence of the bornology \mathcal{B} on the set X, the bornology \mathcal{B} on X will also be denoted by $\mathcal{B}(X)$.

Let (X, \mathcal{B}) and (X', \mathcal{B}') be two bornological spaces and let $f: X \to X'$ be a map between the underlying sets.

Definition 1.1.3. Let (X, \mathcal{B}) and (X', \mathcal{B}') be two bornological spaces and let $f: X \to X'$ be a map between the underlying sets. The map f is called *proper* if the pre-image of every \mathcal{B}' -bounded set is \mathcal{B} -bounded.

Let X be a set. Let U and U' be subsets of $X \times X$; we consider the following operations:

Inversion: $U^{-1} \coloneqq \{(x, y) \in X \times X \mid (y, x) \in U\};$

Composition: $U \circ U' \coloneqq \{(x, y) \in X \times X \mid \exists z \in X : (x, z) \in U \text{ and } (z, y) \in U'\}.$

We denote by $\Delta_X \coloneqq \{(x, x) \in X \times X \mid x \in X\}$ the diagonal of X in the product $X \times X$.

Definition 1.1.4. [BE16, Def. 2.3] A coarse structure on a set X is a subset $C \subseteq \mathcal{P}(X \times X)$ that contains the diagonal Δ_X and is closed under taking subsets, finite unions, inverses, and compositions.

The pair (X, \mathcal{C}) given by a set X and a coarse structure \mathcal{C} on X is called a *coarse* space; elements of \mathcal{C} are called *entourages*, or \mathcal{C} -controlled sets.

Notation 1.1.5. In order to emphasize the dependence on X, the coarse structure C on X will also be denoted by C(X).

Definition 1.1.6. Let (X, \mathcal{C}) and (X', \mathcal{C}') be two coarse spaces and let $f: X \to X'$ be a map between the underlying sets. The map $f: (X, \mathcal{C}) \to (X', \mathcal{C}')$ is called *controlled* (or *coarse*) if for every entourage U of C the set $(f \times f)(U)$ belongs to \mathcal{C}' .

Intersection of coarse structures on X is again a coarse structure on X. When a subset F of $\mathcal{P}(X \times X)$ is given, the coarse structure generated by F is the minimal coarse structure on X that contains F. It is denoted by $\mathcal{C}\langle F \rangle$.

- **Example 1.1.7.** (i) Let X be a set. The minimal bornology \mathcal{B}_{\min} on X consists of all the finite subsets of X; the maximal bornology \mathcal{B}_{\max} on X is $\mathcal{P}(X)$. The minimal coarse structure \mathcal{C}_{\min} is the set $\mathcal{C}_{\min} = \mathcal{P}(\Delta_X)$ generated by the diagonal Δ_X , and the maximal coarse structure \mathcal{C}_{\max} is $\mathcal{P}(X \times X)$.
- (ii) Let (X, d) be a metric space. There is a canonical bornology and a canonical coarse structure associated to (X, d), denoted by \mathcal{B}_d and \mathcal{C}_d respectively. The family

$$\mathcal{B}_d \coloneqq \mathcal{B}\langle \{B(x,r) \mid x \in X, \ r \ge 0\} \rangle$$

is the bornology generated by the *d*-bounded balls B(x,r) of (X,d). We observe that the bornology \mathcal{B}_d on X contains all the *d*-bounded subsets of X. The coarse structure \mathcal{C}_d is generated by the family of subsets of $X \times X$

$$U_r \coloneqq \{(x, y) \mid d(x, y) \le r)\}$$
(1.1.1)

for every $r \geq 0$.

By the previous example we can see that the definition of a bornology on a set X generalizes the notion of *bounded sets* of a metric space (X, d) and that the definition of a coarse structure generalizes the notion, for two points of X, of *being uniformly close*.

Let U be a subset of $X \times X$ and let B be a subset of X; the U-thickening of B is the subset of X defined as follows:

$$U[B] \coloneqq \{x \in X \mid \exists b \in B, \ (x,b) \in U\}$$

$$(1.1.2)$$

If U is an entourage of a coarse space (X, \mathcal{C}) , the U-thickening is also called a *controlled* thickening.

Example 1.1.8. Let (X, d) be a metric space. If $p \in X$ is a point of X and $U_r \in C_d$ is a controlled set as in (1.1.1), then the U_r -thickening of p is the ball of radius r. More generally, the U_r -thickening of a ball of radius s is contained in the ball of radius r + s.

Remark 1.1.9. Let X be a set endowed with a bornology and a coarse structure. For every entourage U and V in $\mathcal{C}(X)$ and bounded set B in $\mathcal{B}(X)$ we have an inclusion $U[V[B]] \subseteq (U \circ V)[B]$ of controlled thickenings.

Recall the definition of proper (Definition 1.1.3) and controlled maps (Definition 1.1.6).

Definition 1.1.10. A bornology \mathcal{B} and a coarse structure \mathcal{C} on a set X are *compatible* if every controlled thickening of every bounded set is bounded.

Definition 1.1.11. [BE16, Definition 2.6 & 2.10] A bornological coarse space is a triple $(X, \mathcal{C}, \mathcal{B})$ given by a set X, a bornology \mathcal{B} and a coarse structure \mathcal{C} on X such that \mathcal{B} and \mathcal{C} are compatible. We denote by **BornCoarse** the category of bornological coarse spaces and proper controlled maps.

When the bornology \mathcal{B} and the coarse structure \mathcal{C} on a set X are clear from the context, we will omit them and we will refer to a bornological coarse space just by its underlying set. Proper and controlled maps between bornological coarse spaces are also called *morphisms* of bornological coarse spaces.

Example 1.1.12. Examples of bornological coarse spaces arise from metric spaces (X, d), by equipping the underlying set X with the associated bornology \mathcal{B}_d and coarse structure \mathcal{C}_d . By Example 1.1.8, \mathcal{C}_d and \mathcal{B}_d are compatible and the space $X_d := (X, \mathcal{C}_d, \mathcal{B}_d)$ is a bornological coarse space.

Example 1.1.13. Let S be a set. We denote by $S_{\min,\max}$ the bornological coarse space $(S, \mathcal{C}_{\min}, \mathcal{B}_{\max})$ whose underlying set is S, equipped with the minimal coarse structure and the maximal bornology.

Example 1.1.14. Let $(X', \mathcal{C}', \mathcal{B}')$ be a bornological coarse space and let X be a set. A map $f: X \to X'$ of sets induces a bornological coarse structure on X. In fact, let

$$f^*\mathcal{C}' \coloneqq \mathcal{C}\langle \{(f \times f)^{-1}(U') \mid U' \in \mathcal{C}'\} \rangle$$

and

$$f^*\mathcal{B}' \coloneqq \mathcal{B}\langle \{f^{-1}(B') \mid B' \in \mathcal{B}'\}\rangle$$

be the coarse structure and bornology on X generated by the preimages. Then, the triple $(X, f^*\mathcal{C}, f^*\mathcal{B})$ is a bornological coarse space and $f: (X, f^*\mathcal{C}, f^*\mathcal{B}) \to (X', \mathcal{C}', \mathcal{B}')$ is a morphism of bornological coarse spaces. When Z is a subset of X' and f is the inclusion of Z in X', we refer to this construction as *inclusion* of bornological coarse spaces.

If X and Y are metric spaces, two maps $f, f' \colon X \to Y$ are close to each other if $d_Y(f(x), f'(x))$ is bounded, uniformly in X. In the context of bornological coarse spaces this leads to the following definition:

Definition 1.1.15. Let $f_0, f_1: (X, \mathcal{C}, \mathcal{B}) \to (X', \mathcal{C}', \mathcal{B}')$ be morphisms in **BornCoarse**. We say that f_0 and f_1 are *close* to each other if the image of the diagonal $(f_0, f_1)(\Delta_X)$ is an entourage of X'.

Observe that being close to each other is an equivalence relation on the set of morphisms between X and X'.

Definition 1.1.16. [BE16, Def. 3.14] A morphism $f: (X, \mathcal{C}, \mathcal{B}) \to (X', \mathcal{C}', \mathcal{B}')$ of bornological coarse spaces is an *equivalence* if there exists an inverse $g: (X', \mathcal{C}', \mathcal{B}') \to (X, \mathcal{C}, \mathcal{B})$ such that the compositions $g \circ f$ and $f \circ g$ are close to the identity maps. In this case, the spaces X and X' are called *coarsely equivalent*.

Example 1.1.17. A quasi-isometry between metric spaces induces a coarse equivalence between the associated bornological coarse spaces.

Definition 1.1.18. [BE16, Def. 3.21] A bornological coarse space X is called *flasque* if it admits a morphism $f: X \to X$ such that:

- (i) f is close to the identity map;
- (ii) for every entourage U of X, the union $\bigcup_{k \in \mathbb{N}} (f^k \times f^k)(U)$ is an entourage;
- (iii) for every bounded set B of X there exists a natural number k such that $f^k(X) \cap B = \emptyset$.

The definition says that the morphism f is equivalent to the identity, is uniformly controlled, together with all its powers, and that each bounded set has eventually no intersections with the image of f^n . An example of such a morphism on the bornological coarse space \mathbb{N} (*i.e.*, the bornological coarse space associated to the set of natural numbers endowed with the standard metric) is the function $\mathbb{N} \to \mathbb{N}$ sending the natural number nto n + 1. On the other hand, the bornological coarse space associated to the integers \mathbb{Z} is not flasque.

Definition 1.1.19. [BE16, Def. 3.2 & 3.5] Let $(X, \mathcal{C}, \mathcal{B})$ be a bornological coarse space.

1. A big family \mathcal{Y} on X is a filtered family $(Y_i)_{i \in I}$ of subsets of X satisfying the following:

 $\forall i \in I, \quad \forall U \in \mathcal{C}, \quad \exists j \in I \text{ such that } U[Y_i] \subseteq Y_j.$

2. A pair (Z, \mathcal{Y}) consisting of a subset Z of X and of a big family $\mathcal{Y} = (Y_i)_{i \in I}$ on X is called a *complementary pair* if there exists an index $i \in I$ for which $Z \cup Y_i = X$.

Example 1.1.20. Let $X = \mathbb{R}$, endowed with bornology and coarse structure induced by the standard euclidean metric on the real numbers. Let $Z = (-\infty, 0]$, and $Y_n = [-n, \infty)$, for $n \in \mathbb{N}$. Then, the pair (Z, \mathcal{Y}) , with $\mathcal{Y} = (Y_n)_{n \in \mathbb{N}}$, is a complementary pair on \mathbb{R} .

Let G be a group acting by automorphisms on a bornological coarse space X. The G-action on X induces a diagonal G-action on $X \times X$, hence a G-action on the power set $\mathcal{P}(X \times X)$ of X and on the set of entourages of X. Let \mathcal{C}^G be the partially ordered subset of \mathcal{C} consisting of the set-wise G-fixed entourages.

Definition 1.1.21. [BEKW17, Definition 2.1] A *G*-bornological coarse space is a bornological coarse space $(X, \mathcal{C}, \mathcal{B})$ equipped with a *G*-action by automorphisms such that the set of invariant entourages \mathcal{C}^G is cofinal in \mathcal{C} .

A morphism of G-bornological coarse spaces is a morphism of bornological coarse spaces that is also G-equivariant. We denote by GBornCoarse the category of G-bornological coarse spaces and G-equivariant, proper controlled maps.

For a subset U of $X \times X$ we set

$$GU\coloneqq \bigcup_{g\in G}(g\times g)(U)$$

The cofinality assumption of \mathcal{C}^G in \mathcal{C} is equivalent to requiring that, for every entourage U of \mathcal{C} , the set GU belongs to \mathcal{C} .

Example 1.1.22. We provide some examples of G-bornological coarse spaces:

- (i) If G acts on a metric space (X, d) by isometries, then $(X, \mathcal{C}_d, \mathcal{B}_d)$ is a G-bornological coarse space.
- (ii) Let G be a group, \mathcal{B}_{\min} be the minimal bornology on its underlying set, and let $\mathcal{C}_{\operatorname{can}} \coloneqq \langle \{G(B \times B) \mid B \in \mathcal{B}_{\min}\} \rangle$ be the coarse structure on G generated by the G-orbits. The space $G_{\operatorname{can,min}} \coloneqq (G, \mathcal{C}_{\operatorname{can}}, \mathcal{B}_{\min})$ is a G-bornological coarse space.
- (iii) Let G be a countable group equipped with a proper left invariant metric d; then, $(G, \mathcal{C}_d, \mathcal{B}_d)$ is the same as $(G, \mathcal{C}_{can}, \mathcal{B}_{min})$ and different choices of d induce the same G-bornological coarse space.
- (iv) Let X be a G-bornological coarse space and let Z be a G-invariant subset of X. As in Example 1.1.14, we define the induced coarse structure and bornology on Z by restriction: $\mathcal{C}_Z := \{(Z \times Z) \cap U \mid U \in \mathcal{C}\}$ and $\mathcal{B}_Z := \{Z \cap B \mid B \in \mathcal{B}\}$. Then, $Z_X := (Z, \mathcal{C}_Z, \mathcal{B}_Z)$ is a G-bornological coarse space and the inclusion $Z \hookrightarrow X$ is a morphism of G-bornological coarse spaces.
- (v) Let X be a G-bornological coarse space and let U be a G-invariant entourage of X. The induced coarse structure C_U is the coarse structure on X generated by U. This is compatible with the bornology \mathcal{B} and $X_U \coloneqq (X, \mathcal{C}_U, \mathcal{B})$ is a G-bornological coarse

space. Moreover, the identity map $X \to X_U$ is a morphism of G-bornological coarse spaces and, if $U \subseteq U'$, there is a natural induced morphism $X_U \to X_{U'}$.

Let X be a G-set. In general, a bornology \mathcal{B} on X is not closed under G-action. However, we can *complete* the bornology \mathcal{B} , obtaining a new G-bornological space with this property:

Definition 1.1.23. [BEKW17, Def. 2.12] Let $(X, \mathcal{C}, \mathcal{B})$ be a *G*-bornological coarse space. The *G*-completion of $(X, \mathcal{C}, \mathcal{B})$ is the *G*-bornological coarse space $B_G X := (X, \mathcal{C}, \mathcal{B}_G)$, where \mathcal{B}_G is the bornology generated by all the sets *GB*, with *B* in \mathcal{B} .

A bornological coarse space $X \in \mathbf{BornCoarse}$ is a *G*-bornological coarse space with trivial *G*-action and the category **BornCoarse** of bornological coarse spaces is a full subcategory of *G***BornCoarse**. The definitions 1.1.15, 1.1.16, 1.1.18 and 1.1.19, have natural extensions to *G*-bornological coarse spaces:

Definition 1.1.24. [BEKW17, Def. 3.1] Two morphisms between G-bornological coarse spaces are *close* to each other if they are close to each other as morphisms of bornological coarse spaces (see Definition 1.1.15). A morphism f of G-bornological coarse spaces is an equivalence if it admits an inverse g such that the two compositions are close to the identities.

Definition 1.1.25. [BEKW17, Def. 3.8] A *G*-bornological coarse space X is called *flasque* if it admits a morphism $f: X \to X$ such that:

- (i) f is close to the identity map;
- (ii) for every entourage U of X, the union $\bigcup_{k \in \mathbb{N}} (f^k \times f^k)(U)$ is again an entourage of X;

(iii) for every bounded set B of X there exists k such that $f^k(X) \cap GB = \emptyset$.

Definition 1.1.26. [BEKW17, Def. 3.5 & 3.7] Let X be a G-bornological coarse space.

- (i) An equivariant big family on X is a big family (see Definition 1.1.19) consisting of G-invariant subsets.
- (ii) An equivariant complementary pair (Z, \mathcal{Y}) on X is a complementary pair (see Definition 1.1.19) where Z is a G-invariant subset of X and \mathcal{Y} is an equivariant big family.

Example 1.1.27. Let X be a G-bornological coarse space and A a G-invariant subset of X. Then, the family

$$\{A\} \coloneqq (U[A])_{U \in \mathcal{C}^G}$$

is an equivariant big family on X (generated by A).

The categories **BornCoarse** and *G***BornCoarse** have natural symmetric monoidal structures (see Definition A.3.1). We conclude the section with a description of them (see [BE16, Example 2.30] and [BEKW17, Example 2.17]).

Let GSet be the category of G-sets and G-equivariant maps. Let

$\mathcal{F}: G\mathbf{BornCoarse} \to G\mathbf{Set}$

be the forgetful faithful functor, which associates to every G-bornological coarse space X its underlying G-set. The category G**Set** has the symmetric monoidal structure given by the cartesian product. The symmetric monoidal structure on G**BornCoarse** is then defined by pulling back the symmetric monoidal structure on G**Set**, as we now describe.

Let X and X' be G-bornological coarse spaces. Then, their tensor product

$$X \otimes_{G\mathbf{BornCoarse}} X'$$

is the G-bornological coarse space defined as follows:

- 1. The underlying G-set of $X \otimes_{GBornCoarse} X'$ is the cartesian product $X \times X'$ of the underlying G-sets, with the diagonal action.
- 2. The bornology on $X \times X'$ is generated by all the subsets $B \times B'$, with B and B' varying among all the bounded sets of X and X'.
- 3. The coarse structure on $X \times X'$ is generated by the entourages $U \times U'$ with U in $\mathcal{C}(X)$ and U' in $\mathcal{C}(X')$.

If $f: X \to Y$ and $f': X' \to Y'$ are morphisms of G-bornological coarse spaces, then their tensor product

 $f \otimes f' : X \otimes_{G\mathbf{BornCoarse}} Y \to X' \otimes_{G\mathbf{BornCoarse}} Y'$

is the equivariant map between the underlying G-sets $(x, y) \mapsto (f(x), f(y))$. This describes a bifunctor

 $-\otimes_{GBornCoarse} - : GBornCoarse \times GBornCoarse \rightarrow GBornCoarse,$ (1.1.3)

which agrees with the cartesian symmetric monoidal structure of GSet on the underlying G-sets. The tensor unit $1_{GBornCoarse}$ (Definition A.3.1.2) is given by the one-point space *.

The functor \mathcal{F} preserves the unit and the tensor product strictly, *i.e.*, the morphisms 1 and 2 in Definition A.3.4 are identities. The associator, unit and symmetry constraints are imported from those in **GSet** and are implemented by morphisms of *G*-bornological coarse spaces; the relations of Definition A.3.1 are then satisfied and this finishes the description of the symmetric monoidal structure on the category **GBornCoarse**.

Remark 1.1.28. The one-point space * is a commutative algebra object in *G*BornCoarse.

Notation 1.1.29. We will use the shorter notation $X \otimes X'$ for the tensor product $X \otimes_{GBornCoarse} X'$.

1.2 Equivariant coarse homology theories

The goal of this section is to recall the definition of equivariant coarse homology theories [BEKW17, Def. 3.10]. A coarse homology theory is a functor

$E: G\mathbf{BornCoarse} \to \mathbf{C}$

from the category of G-bornological coarse spaces to a cocomplete stable ∞ -category C (e.g., the ∞ -category of chain complexes \mathbf{Ch}_{∞} or the ∞ -category of spectra \mathbf{Sp}), satisfying additional axioms: coarse invariance, flasqueness, coarse excision and u-continuity. These are reviewed in Definition 1.2.1.

In order to study general properties of equivariant coarse homology theories, suitable categories $G\mathbf{Spc}\mathcal{X}$ of motivic coarse spaces and $G\mathbf{Sp}\mathcal{X}$ of motivic coarse spectra have been constructed [BE16, BEKW17]; in particular, the category $G\mathbf{Sp}\mathcal{X}$ is the target of a universal coarse homology theory

$\operatorname{Yo}_G^s \colon G\mathbf{BornCoarse} \to G\mathbf{Sp}\mathcal{X}$

with the property that every equivariant coarse homology theory $E: GBornCoarse \to C$ has an essentially unique factorization over Yo_G^s . After reviewing these facts, we see that equivariant coarse homology theories can be described as colimit-preserving functors on the category of motivic coarse spaces and we conclude the section with the definitions of a strong and of a continuous equivariant coarse homology theory; we will use these notions in Section 1.3 in relation with forget-control and assembly maps.

In the following, let **C** be a cocomplete stable ∞ -category, let $E: GBornCoarse \to C$ be a functor and let X be a G-bornological coarse space.

If $\mathcal{Y} = (Y_i)_{i \in I}$ is a filtered family of *G*-invariant subsets of *X*, we define the value of the functor *E* at the family \mathcal{Y} as the colimit in **C**

$$E(\mathcal{Y}) \coloneqq \operatorname{colim}_{i \in I} E(Y_i). \tag{1.2.1}$$

Here, the subsets Y_i of X are endowed with the G-bornological coarse structure induced from X (by Example 1.1.22 (iv)). Observe that there is a natural morphism $E(\mathcal{Y}) \to E(X)$ induced by all the inclusions $Y_i \to X$.

Let Z be a G-invariant subset of X, with the induced structures. If \mathcal{Y} is an equivariant big family on X (see Definition 1.1.26), then the intersection $Z \cap \mathcal{Y} := (Z \cap Y_i)_{i \in I}$ is an equivariant big family on Z and $E(Z \cap \mathcal{Y})$ denotes the value of E at this filtered family; moreover, we observe that there are natural morphisms $E(Z \cap \mathcal{Y}) \to E(Z)$ and $E(Z \cap \mathcal{Y}) \to E(\mathcal{Y})$.

Let $\{0, 1\}_{\max,\max}$ be the *G*-bornological coarse space whose underlying set is $\{0, 1\}$, endowed with the maximal coarse structure and the maximal bornology; the *G*-action is the trivial one. Recall that C^G denotes the family of *G*-invariant entourages of *X*. Recall Definition 1.1.25 of a flasque *G*-bornological coarse space and Definition 1.1.26 of equivariant complementary pair. We recall that the space X_U , for *U* a *G*-invariant entourage, is the *G*-bornological coarse space with coarse structure induced by *U* of Example 1.1.22 (v). We can now give the fundamental definition of an equivariant coarse homology theory:

Definition 1.2.1. [BEKW17, Definition 3.10] Let G be a group and let **GBornCoarse** be the category of G-bornological coarse spaces. Let **C** be a cocomplete stable ∞ -category. A *G*-equivariant **C**-valued coarse homology theory is a functor

$E\colon G\mathbf{BornCoarse} \longrightarrow \mathbf{C}$

with the following properties:

i. Coarse invariance: for all $X \in GBornCoarse$, the projection

$$\{0,1\}_{\max,\max}\otimes X\to X$$

is sent by E to an equivalence of \mathbf{C} ;

- ii. Flasqueness: if X is a flasque G-bornological coarse space, then $E(X) \simeq 0$;
- iii. Coarse excision: $E(\emptyset) \simeq 0$, and for every equivariant complementary pair (Z, \mathcal{Y}) on X, the diagram

$$E(Z \cap \mathcal{Y}) \longrightarrow E(Z)$$

$$\downarrow \qquad \qquad \downarrow$$

$$E(\mathcal{Y}) \longrightarrow E(X)$$

is a push-out square;

iv. **u-continuity:** for every *G*-bornological coarse space $(X, \mathcal{C}, \mathcal{B})$, the canonical morphisms $X_U \to X$ induce an equivalence $E(X) \simeq \operatorname{colim}_{U \in \mathcal{C}^G} E(X_U)$.

We recall that the product \otimes of *G*-bornological coarse spaces in Definition 1.2.1 is the symmetric monoidal product (1.1.3).

Remark 1.2.2. [BEKW17, Rem. 3.11] Coarse invariance in Definition 1.2.1 is equivalent to asking that the functor E sends equivalences of G-bornological coarse spaces (see Definition 1.1.24) to equivalences of \mathbf{C} .

1.2. EQUIVARIANT COARSE HOMOLOGY THEORIES

Examples of (equivariant) coarse homology theories are: coarse ordinary homology and coarse topological K-theory [BE16], coarse algebraic K-homology and coarse topological Hochschild homology [BEKW17, BC17], coarse A-theory [BKW18], coarse Hochschild and cyclic homology (see Theorem 3.4.7). We will recall the definitions of coarse ordinary homology in Section 1.5 and of coarse algebraic K-theory in Section 2.3.

Equivariant C-valued coarse homology theories can be equivalently seen as colimitpreserving functors $G\mathbf{Sp}\mathcal{X} \to \mathbf{C}$, where $G\mathbf{Sp}\mathcal{X}$ is a suitable category of motivic coarse spectra, as we now explain. In [BE16, Section 3.4] and [BEKW17, Section 4.1], ∞ categories $\mathbf{Spc}\mathcal{X}$ and $G\mathbf{Spc}\mathcal{X}$ of motivic coarse spaces and of *G*-equivariant motivic coarse spaces respectively, together with the corresponding stable versions $\mathbf{Sp}\mathcal{X}$ and $G\mathbf{Sp}\mathcal{X}$, have been constructed. This is implemented by first completing the category of bornological coarse spaces (by embedding it in the category of spaces-valued presheaves) and then by localizing (in the realm of ∞ -categories) the obtained ∞ -category at various sets of morphisms: these encode the properties of descent, coarse equivalences, vanishing on flasque *G*-bornological coarse spaces, and u-continuity. The ∞ -categories of motivic coarse spectra $\mathbf{Sp}\mathcal{X}$ and $G\mathbf{Sp}\mathcal{X}$ are defined as the stabilizations of the ∞ -categories $\mathbf{Spc}\mathcal{X}$ and $G\mathbf{Spc}\mathcal{X}$ of motivic coarse spaces [BE16, Sec. 4].

Every bornological coarse space X represents a motivic coarse space Yo(X) in $Spc\mathcal{X}$. In fact, this is described by *Yoneda functors*

$$Yo: BornCoarse \to Spc\mathcal{X}$$
(1.2.2)

and

$$Yo_G : GBornCoarse \to GSpc\mathcal{X},$$
 (1.2.3)

sending a bornological coarse space (or a G-bornological coarse space) to its (equivariant) coarse motivic space. The stable versions of these functors are denoted by

$$Yo^s : BornCoarse \to Sp\mathcal{X}$$
 (1.2.4)

and

$$\operatorname{Yo}_{G}^{s}: G\mathbf{BornCoarse} \to G\mathbf{Sp}\mathcal{X}.$$
 (1.2.5)

The category $G\mathbf{Sp}\mathcal{X}$ of motivic coarse spectra is a stable cocomplete ∞ -category and the functor Yo_G^s : $G\mathbf{BornCoarse} \to G\mathbf{Sp}\mathcal{X}$ is the universal *G*-equivariant coarse homology theory [BEKW17, Def. 4.9]: every equivariant coarse homology theory factorizes in an essentially unique way over it.

Let **C** be a cocomplete stable ∞ -category and let **CoarseHomologyTheories**_C be the full subcategory of **Fun**(**BornCoarse**, **C**) consisting of all functors satisfying coarse invariance, excision, continuity and vanishing on flasque spaces (as in Definition 1.2.1). Let **Fun**^{colim}(**Spc** \mathcal{X} , **C**) be the category of colimit-preserving functors from **Spc** \mathcal{X} to **C**. **Remark 1.2.3.** [BE16, Cor. 3.34] For every cocomplete ∞ -category **C**, we have an equivalence between the ∞ -categories **CoarseHomologyTheories**_C and **Fun**^{colim}(**Spc** \mathcal{X} , **C**).

Analogously, pre-composition with the functor Yo_G^s , yields an equivalence between the ∞ -category of *G*-equivariant coarse homology theories and the category of colimitpreserving functors $G\mathbf{Sp}\mathcal{X} \to \mathbf{C}$ [BEKW17, Cor. 4.10].

Notation 1.2.4. If X is in BornCoarse (or GBornCoarse) and E is a colimitpreserving functor in $\operatorname{Fun}^{\operatorname{colim}}(\operatorname{Spc}\mathcal{X}, \mathbb{C})$ (or in $\operatorname{Fun}^{\operatorname{colim}}(\operatorname{GSpc}\mathcal{X}, \mathbb{C})$), then we write E(X) instead of $E(\operatorname{Yo}(X))$ (or $E(\operatorname{Yo}_G(X))$).

The categories **BornCoarse** and *G***BornCoarse** have symmetric monoidal structures denoted by \otimes (1.1.3). In the same way, the categories **Spc** \mathcal{X} and *G***Spc** \mathcal{X} have symmetric monoidal structures, also denoted by \otimes , which are essentially uniquely determined by the requirement that the functors Yo and Yo_G refine to symmetric monoidal functors. For Xin *G***Spc** \mathcal{X} and Y in *G***BornCoarse** we will often write $Y \otimes X$ instead of Yo_G(Y) $\otimes X$.

Definition 1.2.5. Let $E: GBornCoarse \to C$ be a functor and let X be a G-bornological coarse space. The twist E_X of E by X is the functor

$E_X : G\mathbf{BornCoarse} \to \mathbf{C}$

defined by $E_X(Y) \coloneqq E(X \otimes Y)$.

Remark 1.2.6. Analogously, we can twist a coarse homology theory E by a coarse motivic space $\operatorname{Yo}_G(X)$ in $G\operatorname{Spc} \mathcal{X}$: $E_X(Y) \coloneqq E(\operatorname{Yo}_G(X) \otimes Y)$.

If E is a G-equivariant coarse homology theory, it is natural to ask whether the twisted version is also a coarse homology theory:

Lemma 1.2.7. [BEKW17, Lemma 4.17] If E is a G-equivariant coarse homology theory and X is a G-bornological coarse space, then the twist E_X is a G-equivariant coarse homology theory.

We conclude the section with the definitions of *strongness* and *continuity* for a (equivariant) coarse homology theory. These properties are important is the study of coarse assembly maps. We first need a more general definition of flasque spaces (recall Definition 1.1.25):

Definition 1.2.8. [BEKW17] A *G*-bornological coarse space X is called *weakly flasque* if it admits a morphism $f: X \to X$ such that:

- 1. $\operatorname{Yo}_{G}^{s}(f) \simeq \operatorname{id}_{\operatorname{Yo}_{G}^{s}}(X);$
- 2. for every entourage U of X, the union $\bigcup_{k \in \mathbb{N}} (f^k \times f^k)(U)$ is again an entourage of X;

3. for every bounded set B of X there exists k such that $f^k(X) \cap GB = \emptyset$.

Definition 1.2.9. [BEKW17, Def. 4.19] Let $E: GBornCoarse \to C$ be a *G*-equivariant coarse homology theory. Then, *E* is *strong* if $E(X) \simeq 0$ for all weakly flasque *G*-bornological coarse spaces *X*.

Remark 1.2.10. [BEKW17, Lemma 11.25] Let $E: GBornCoarse \to C$ be a *G*-equivariant coarse homology theory. Let X be a *G*-bornological coarse space. If E is strong, then also the twist E_X by X is strong equivariant coarse homology theory.

We now proceed with the definition of a continuous coarse homology theory. Let X be a G-bornological coarse space.

Definition 1.2.11. [BEKW17, Def. 5.1] Let F be a subset of X. Then, F is called *locally finite* if $B \cap F$ is finite for every bounded set $B \in \mathcal{B}(X)$.

Definition 1.2.12. [BEKW17, Def. 5.6] Let $\mathcal{Y} = (Y_i)_{i \in I}$ be a filtered family of invariant subsets of X. The family \mathcal{Y} is called a *trapping exhaustion* if, for every locally finite G-invariant subset F of X, there exists an index $i \in I$ for which F is contained in Y_i .

Let $E: GBornCoarse \to \mathbb{C}$ be a *G*-equivariant coarse homology theory. Recall that, for a given filtered family $\mathcal{Y} = (Y_i)_i$, we define $E(\mathcal{Y})$ as the colimit of the $E(Y_i)$ (1.2.1).

Definition 1.2.13. [BEKW17, Def. 5.15] A *G*-equivariant coarse homology theory *E* is *continuous* if for every trapping exhaustion \mathcal{Y} of a *G*-bornological coarse space *X*, the morphism

$$E(\mathcal{Y}) \to E(X)$$

is an equivalence.

1.3 Forget-control and assembly maps

One of the main applications of coarse geometry is in the study of (coarse) assembly map conjectures. We refer to [BLR07, RV17, Lue18] for recent surveys on assembly map conjectures and controlled methods. The coarse geometry approach to these conjectures consists in analyzing the so-called *forget-control* map and then, by comparison, in getting information about the assembly maps, as we shall see.

In this section, we recall the definitions of forget-control map [BEKW17, Def. 11.10] and of coarse assembly map [BEKW17, Def. 10.24] in the framework of [BEKW17]. We first need to introduce the category GUBC of G-uniform bornological coarse spaces (see Definition 1.3.3). Then, we define the so-called cone functors

$$\mathcal{O}, \mathcal{O}^{\infty} \colon GUBC \to GBornCoarse$$

that are functors from the category of G-uniform bornological coarse spaces to the category of G-bornological coarse spaces. From these functors, we get a fiber sequence

$$F^0 \to F \to F^\infty \to \Sigma F^0$$

of functors $GBornCoarse \rightarrow GSp\mathcal{X}$ (1.3.2). The forget-control map is then defined as the connecting morphism associated to this fiber sequence.

After giving the definition of coarse assembly map (Definition 1.3.19), that is related to the classical assembly map [BEKW17, BEKW18], we conclude the section with a comparison result [BEKW17, Thm. 11.16]: under some mild assumptions, the coarse assembly map and the forget-control map provide equivalent morphisms when restricted to the family of subgroups (see Definition 1.3.16) **Fin** of finite subgroups.

We start by recalling the definition of G-uniform bornological coarse space.

Definition 1.3.1. Let X be a G-set. A G-uniform structure on X is a non-empty subset $\mathcal{U} \subseteq \mathcal{P}(X \times X)$ with the following properties:

- every element of \mathcal{U} contains the diagonal Δ_X of X;
- \mathcal{U} is closed under inverses, compositions, finite intersections and supersets;
- for every $U \in \mathcal{U}$ there exists $V \in \mathcal{U}$ such that $V \circ V \subseteq U$;
- for every U in \mathcal{U} the set $\bigcap_{g \in G} gU$ belongs to \mathcal{U} .

A *G*-uniform space (X, \mathcal{U}) is a pair given by a *G*-set *X* and a *G*-uniform structure \mathcal{U} on *X*. If (X, \mathcal{U}) and (X', \mathcal{U}') are *G*-uniform spaces, a *G*-equivariant map $f: X \to X'$ is uniform if $(f \times f)^{-1}(U') \in \mathcal{U}$ for every $U' \in \mathcal{U}'$.

Definition 1.3.2. Let X be a G-set endowed with a coarse structure \mathcal{C} and a uniform structure \mathcal{U} . The structures \mathcal{U} and \mathcal{C} are *compatible* if $\mathcal{C}^G \cap \mathcal{U}^G$ is not empty.

We can define a category GUBC of G-uniform bornological coarse spaces:

Definition 1.3.3. [BEKW17, Def. 9.9] A *G*-uniform bornological coarse space is a G-bornological coarse space with a compatible G-uniform structure. Morphisms of G-uniform bornological coarse spaces are morphisms of G-bornological coarse spaces that are also uniform.

Remark 1.3.4. The category G**UBC** is symmetric monoidal. If X and Y are G-uniform bornological coarse spaces, the product $X \otimes Y$ is defined to be the G-bornological coarse space $X \otimes Y$ endowed with the (compatible) G-uniform structure generated by all the products $U_X \times U_Y$ of invariant uniform entourages of X and Y. **Example 1.3.5.** Let (X, d) be a metric space with *G*-action. The uniform structure \mathcal{U}_d on *X* associated to the metric *d* is generated by the sets

$$U_r = \{(x, y) \in X \times X \mid d(x, y) \le r\}$$

for all r in $(0, \infty)$. This is compatible with the coarse structure C_d of Example 1.1.22 (i). We denote by X_{du} the *G*-uniform bornological coarse space obtained in this way.

Let X be a G-uniform bornological coarse space and let $\mathcal{Y} = (Y_i)_{i \in I}$ be a G-equivariant filtered family on X. Consider an order-preserving function $\varphi \colon I \to \mathcal{P}(X \times X)^G$, where $\mathcal{P}(X \times X)^G$ is the family of G-invariant subsets of $X \times X$ ordered by the opposite of the inclusion relation. We say that φ is \mathcal{U}^G -admissible [BEKW17, Def. 9.14] if for every $U \in \mathcal{U}^G$ there exists *i* in *I* such that $\varphi(i) \subseteq U$.

For a \mathcal{U}^G -admissible function $\varphi \colon I \to \mathcal{P}(X \times X)^G$, define the *G*-invariant entourages

$$U_{\varphi} \coloneqq \{(x, y) \in X \times X \mid (\forall i \in I \mid (x, y) \in Y_i \times Y_i \text{ or } (x, y) \in \varphi(i))\}.$$

Assume also that $\mathcal{Y} = (Y_i)_{i \in I}$ is a *G*-equivariant big family on *X* (see Definition 1.1.26); then, the *hybrid coarse structure* \mathcal{C}_h is the coarse structure generated by the entourages $U \cap U_{\varphi}$ for all the *G*-invariant entourages *U* in \mathcal{C}^G and all the \mathcal{U}^G -admissible functions $\varphi \colon I \to \mathcal{P}(X \times X)^G$ [BEKW17, Def. 9.15].

Definition 1.3.6. [BEKW17, Def. 9.16] Let X be a G-uniform bornological coarse space. The hybrid space X_h is the G-bornological coarse space $X_h := (X, \mathcal{C}_h, \mathcal{B})$ with the hybrid coarse structure and the same bornology.

In order to define the forget-control map, we first need to define the cone functor $\mathcal{O}: GUBC \to GBornCoarse$ [BEKW17, Sec. 9.4] from the category of *G*-uniform bornological coarse spaces to the category of *G*-bornological coarse spaces. We now explain how to construct it.

One first considers the metric space $[0, \infty)$ endowed with the trivial *G*-action. By Example 1.3.5, we get the *G*-uniform bornological coarse space $[0, \infty)_{du}$. If *X* is a *G*-uniform bornological coarse space, the product $[0, \infty)_{du} \otimes X$ is still a *G*-uniform bornological coarse space by Remark 1.3.4. Let \mathcal{Y} be the *G*-equivariant big family $\mathcal{Y}(X) \coloneqq ([0, n] \times X)_{n \in \mathbb{N}}$ on $[0, \infty)_{du} \otimes X$.

Definition 1.3.7. [BEKW17, Def. 9.24] The cone functor

$\mathcal{O}: GUBC \to GBornCoarse$

sends a *G*-uniform bornological coarse space *X* to the hybrid *G*-bornological coarse space $([0, \infty)_{du} \otimes X)_h$. If $f: X \to X'$ is a morphism of *G*-uniform bornological coarse spaces, then $\mathcal{O}(f): \mathcal{O}(X) \to \mathcal{O}(X')$ is the morphism of *G*-bornological coarse spaces induced by $\mathrm{id}_{[0,\infty)} \times f$.

For further details on the functoriality of the cone functor \mathcal{O} , we refer to [BEKW17, Sec. 9.4].

Let \mathbb{R} be the set of real numbers endowed with the euclidean metric; hence, \mathbb{R} is a G-uniform bornological coarse space (with trivial G-action). For a G-uniform bornological coarse space X, the product $\mathbb{R} \otimes X$ is again a G-uniform bornological coarse space; let \mathcal{Y} be the equivariant big family $\mathcal{Y} := ((-\infty, n] \times X)_{n \in \mathbb{N}}$ on it. As above, these data describe a functor:

Definition 1.3.8. [BEKW18, Def. 4.9] We denote by

$\mathcal{O}_{\text{geom}}^{\infty}$: $GUBC \rightarrow GBornCoarse$

the functor that sends a *G*-uniform bornological coarse space to the hybrid *G*-bornological coarse space $\mathcal{O}_{\text{geom}}^{\infty}(X) := (\mathbb{R} \otimes X)_h$.

Let $\operatorname{Yo}_G^s : GBornCoarse \to GSp\mathcal{X}$ be the stable Yoneda functor (1.2.5) and let $GSp\mathcal{X}$ be the category of motivic coarse spectra.

Definition 1.3.9. [BEKW18, Def. 4.10] We denote by \mathcal{O}^{∞} the composition

$$\mathcal{O}^{\infty} \coloneqq \operatorname{Yo}_{G}^{s} \circ \mathcal{O}_{\operatorname{geom}}^{\infty} \colon GUBC \to GSp\mathcal{X}.$$

This composition is called *cone-at-infinity functor*.

Let $\mathcal{F}: GUBC \to GBornCoarse$ be the forgetful functor (which forgets the uniform structure). By [BEKW17, Cor. 9.30] (see also [BEKW18, Sec. 4]), we have a fibre sequence of functors

$$\dots \to \operatorname{Yo}_{G}^{s} \circ \mathcal{F} \to \operatorname{Yo}_{G}^{s} \circ \mathcal{O} \to \mathcal{O}^{\infty} \to \Sigma \operatorname{Yo}_{G}^{s} \circ \mathcal{F} \to \dots$$
(1.3.1)

called the *cone sequence*. The first map of the sequence is induced by the inclusion of X in the product $[0, \infty) \times X$, the second by the inclusion of $\mathcal{O}(X)$ in $\mathcal{O}_{geom}^{\infty}(X)$, and the third is described as follows: the spaces $\mathcal{O}_{geom}^{\infty}(X)$ and $\mathbb{R} \otimes \mathcal{F}(X)$ have the same underlying sets and the identity map induces a morphism $\operatorname{Yo}_{G}^{s}(\mathcal{O}_{geom}^{\infty}(X)) \to \operatorname{Yo}_{G}^{s}(\mathbb{R} \otimes \mathcal{F}(X))$. The boundary map $\mathcal{O}^{\infty} \to \Sigma \operatorname{Yo}_{G}^{s} \circ \mathcal{F}$ is then obtained after applying the equivalence $\operatorname{Yo}_{G}^{s}(\mathbb{R} \otimes \mathcal{F}(X)) \simeq \Sigma \operatorname{Yo}_{G}^{s}(\mathcal{F}(X))$ (that is obtained by using excision with respect to the complementary pair of Example 1.1.20).

Definition 1.3.10. Let X be a bornological coarse space and U an entourage of X. Let μ be a probability measure on the measurable space $(X, \mathcal{P}(X))$. Then, μ is called *finite* U-bounded if there is a finite subset of X such that $F \times F$ is contained in U and $\mu(F) = 1$. The support supp (μ) of μ is the smallest subset of X of μ -measure 1.

Definition 1.3.11. [BEKW17, Def. 11.1] Let X be a bornological coarse space and U an entourage of X. Then, we denote by $P_U(X)$ the set of finite U-bounded probability measures on X.

The set $P_U(X)$ is actually a *G*-simplicial complex [BEKW17, Sec. 11.1]. On the set $P_U(X)$, we consider the bornology generated by the sets $P_U(B)$ for all the bounded sets *B* of *X* and the coarse and uniform structures induced by the path metric. Then, $P_U(X)$, with these structures, is a *G*-uniform bornological coarse space [BEKW17, Def. 11.5] that we denote by $P_U(X)_u$. When equipped with only the bornology and the coarse structure (hence after forgetting the uniform structure), it is denoted by $P_U(X)_d$.

The construction of $P_U(X)$ can be made functorial (in pairs (X, U) of bornological coarse spaces and fixed entourage) as described in [BEKW18, Sec. 4].

Definition 1.3.12. [BEKW17, Def. 11.9] Let X be a G-bornological coarse space, U a G-invariant entourage and let $P_U(X)$ be the set of Definition 1.3.11 equipped with the bornology and the uniform and coarse structure described above. Then, we define:

$$F(X) \coloneqq \operatorname{colim}_{U \in \mathcal{C}^G(X)} \operatorname{Yo}_G^s(P_U(X)_u)$$
$$F^{\infty}(X) \coloneqq \operatorname{colim}_{U \in \mathcal{C}^G(X)} \mathcal{O}^{\infty}(P_U(X)_u)$$
$$F^0(X) \coloneqq \operatorname{colim}_{U \in \mathcal{C}^G(X)} \operatorname{Yo}_G^s(P_U(X)_d).$$

These refine to functors F^0, F, F^∞ : **GBornCoarse** $\to GSp\mathcal{X}$ [BEKW17, Rmk. 11.11]. Composition of the functor P with the fibre sequence (1.3.1) induces a fibre sequence of functors **GBornCoarse** $\to GSp\mathcal{X}$

$$F^0 \to F \to F^\infty \to \Sigma F^0 \tag{1.3.2}$$

(see [BEKW17, Def. 11.9] or also [BEKW18, Def. 4.15]). We are particularly interested in the boundary map $F^{\infty} \to \Sigma F^0$ induced by the cone boundary:

Definition 1.3.13. [BEKW17, Def. 11.10] Let X be a G-bornological coarse space. Then, the map

$$\beta_X \colon F^\infty(X) \to \Sigma F^0(X)$$

in the cone sequence (1.3.2) is called the *forget-control* map.

Remark 1.3.14. The functors F^0, F^∞ are not coarse homology theories, hence β is not a transformation of coarse homology theories; however, if E is a strong equivariant coarse homology theory, then the induced forget-control map

$$\beta \colon E \circ F^{\infty} \to \Sigma E \circ F^0$$

is a transformation of equivariant coarse homology theories [BEKW17, Cor. 11.26].

We now introduce the necessary terminology for defining the coarse assembly map.

For a given group G, let GOrb be the category of transitive G-sets and G-equivariant maps. Let GTop be the category of topological spaces (compactly generated weakly Hausdorff spaces) endowed with a continuous G-action.

The cone functor \mathcal{O}^{∞} of Definition 1.3.9 induces a new functor [BEKW17, Def. 10.10]

$$\mathcal{O}_{hlg}^{\infty} \colon G\mathbf{Top} \to G\mathbf{Sp}\mathcal{X}$$
 (1.3.3)

that is an equivariant homology theory [BEKW17, Prop. 10.11]. By Elmendorf's Theorem, equivariant homology theories with values is a cocomplete stable ∞ -category **C** are the same as functors G**Orb** \rightarrow **C** (see [Blu17, Thm. 1.3.8] for a recent reformulation of the classical Elmendorf's Theorem [Elm83]). The equivariant homology theory $\mathcal{O}_{hlg}^{\infty}$ is essentially uniquely characterized by the natural equivalence

$$\mathcal{O}^{\infty}_{\text{hlg}}(S) \simeq \Sigma \operatorname{Yo}^{s}_{G}(S_{min,max}) \tag{1.3.4}$$

for S in GOrb, where $S_{\min,\max}$ denotes the G-bornological coarse space S with the minimal coarse structure and the maximal bornology (see [BEKW17, Prop. 9.35] and the text after [BEKW18, Rmk. 8.17]). We refer to [BEKW18, Def. 8.16] for a construction of the functor $\mathcal{O}_{hlg}^{\infty}$.

Let **C** be a stable cocomplete ∞ -category and let *E* be a *G*-equivariant **C**-valued coarse homology theory, seen as a colimit preserving functor on $G\mathbf{Sp}\mathcal{X}$.

Remark 1.3.15. [BEKW17, Prop. 9.35] The composition $E \circ \mathcal{O}_{hlg}^{\infty}$: G**Top** \to **C** is a G-equivariant homology theory. By the equivalence (1.3.4), if E_L is the twist of E by a G-bornological coarse space L, then we get the equivalences

$$E_L \circ \mathcal{O}_{hla}^{\infty}(S) \simeq \Sigma E_L(\operatorname{Yo}_G^s(S_{min,max}))$$

for S in GOrb.

Definition 1.3.16. Let \mathcal{F} be a non-empty set of subgroups of G. The set \mathcal{F} is called a *family of subgroups* if it is closed under conjugation and under taking subgroups.

Example 1.3.17. Let G be a group. The trivial family $\{*\}$, the set **Fin** of all finite subgroups of G and the set **All** of all subgroups of G are families of subgroups.

Let G be a discrete group. We recall that a classifying space for a family of subgroups \mathcal{F} of G is a G-CW-complex $E_{\mathcal{F}}G$ (unique up to G-homotopy) with the property that, for every H in \mathcal{F} , the set of H-fixed points of $E_{\mathcal{F}}G$ is contractible, and is empty otherwise.

Definition 1.3.18. We denote by $E_{\mathcal{F}}G^{cw}$ the choice of a model of the classifying space for the family of subgroups \mathcal{F} of G.

The projection $E_{\mathcal{F}}G^{cw} \to \{*\}$ to the one point space is a morphism in G**Top**. By applying an equivariant homology theory F we get a map $F(E_{\mathcal{F}}G^{cw}) \to F(\{*\})$, that is the assembly map for F and the family of subgroups \mathcal{F} . When F is the homology theory $\mathcal{O}_{hlg}^{\infty}: G$ **Top** $\to G$ **Sp** \mathcal{X} , we get an assembly map that is called the universal assembly map [BEKW17, Def. 10.21].
For a *G*-topological space *X*, the projection $X \to \{*\}$ to the one point space induces a morphism $\alpha_X \colon \mathcal{O}^{\infty}_{hlg}(X) \to \mathcal{O}^{\infty}_{hlg}(*)$.

Definition 1.3.19. [BEKW17, Def. 10.24] Let Q be a G-bornological coarse space such that the G-action on the underlying G-set is free and let X be a G-topological space. Then, the morphism

 $\alpha_{X,Q} \coloneqq \alpha_X \otimes \operatorname{Yo}^s(Q) \colon \mathcal{O}^\infty_{\operatorname{hlg}}(X) \otimes \operatorname{Yo}^s(Q) \to \Sigma \operatorname{Yo}^s_G(Q)$

is called the *motivic assembly map* with twist Q.

We refer to [BEKW17, Sec. 10.3] for further details. The main example for us is given by $Q = G_{\text{can,min}}$ of Example 1.1.22(ii) and for X by the choice of a model of $E_{\text{Fin}}G$. In fact, with these assumptions, we get the assembly map for the equivariant homology theory $E \circ \mathcal{O}_{\text{hlg}}^{\infty}$.

We also consider twists $\beta_{X,Q}$ of the forget-control map β_X (Definition 1.3.13).

We conclude the section with a comparison result between the coarse assembly map $\alpha_{E_{\mathbf{Fin}}G,G_{\mathrm{can,min}}}$ and the forget-control map $\beta_{G_{\mathrm{can,min}},G_{\mathrm{max,max}}}$:

Theorem 1.3.20. [BEKW17, Thm. 11.16] Let $E: GBornCoarse \to C$ be a continuous G-equivariant coarse homology theory. Then, the coarse assembly map $\alpha_{E_{\mathbf{Fin}}G,G_{\mathrm{can,min}}}$ and the forget-control map $\beta_{G_{\mathrm{can,min}},G_{\mathrm{max,max}}}$ induce equivalent morphisms $E(\alpha_{E_{\mathbf{Fin}}G,G_{\mathrm{can,min}}})$ and $E(\beta_{G_{\mathrm{can,min}},G_{\mathrm{max,max}}})$.

New developments and applications to split-injectivity results of the assembly maps can be found in [BEKW18].

1.4 Localization theorems for coarse homology theories

Let G be a discrete group and denote by R(G) its representation ring [Seg68b]. Let γ be a conjugacy class of G and denote by \mathfrak{p} the (prime) ideal of R(G) consisting of representations in R(G) with trace 0. Segal proves a localization theorem [Seg68a, Prop. 4.1] saying that, after localization at the prime ideal \mathfrak{p} , the map

$$K^*_G(X) \to K^*_G(X^{\gamma})$$

in equivariant K-theory induced from the inclusion of the γ -fixed points X^{γ} in X (for a locally compact G-space X) is an isomorphism.

Inspired by Segal's localization theorem, with the aim to develop new tools for studying equivariant coarse algebraic K-homology, a general localization result for equivariant coarse homology theories has been proved [BCb]. In this section we recall and state the principal results of [BCb], in particular the coarse localization theorems Theorem 1.4.15 and Theorem 1.4.21. We will apply these theorems in Section 4.5, where a localization theorem for G-equivariant coarse Hochschild homology and cyclic homology is given.

Definition 1.4.1. [Mac71, Exercise IX.6.3] Let \mathbf{I} be an ordinary category. The *twisted* arrow category $\mathbf{Tw}(\mathbf{I})$ is the category defined as follows:

- 1. objects: the objects of $\mathbf{Tw}(\mathbf{I})$ are arrows $i \to j$ in \mathbf{I} .
- 2. morphisms: a morphism $(i \to j) \to (i' \to j')$ is a commuting diagram



with the natural compositions.

Observe that the twisted arrow category of I comes equipped with a functor

$$\pi : \mathbf{Tw}(\mathbf{I}) \to \mathbf{I}^{op} \times \mathbf{I} , \quad (i \to j) \mapsto (i, j) .$$

Definition 1.4.2. [GHN17, Def. 2.5] Let **I** be a category and let **C** be a cocomplete ∞ -category. For a functor $F : \mathbf{I}^{op} \times \mathbf{I} \to \mathbf{C}$, we define its *coend*

$$\int^{\mathbf{I}} F := \operatorname{colim}_{\mathbf{Tw}(\mathbf{I})} F \circ \pi.$$

as the colimit over the twisted arrow category $\mathbf{Tw}(\mathbf{I})$ of $F \circ \pi$.

Let G be a discrete group, GOrb the category of transitive G-sets and G-equivariant maps and let C be a stable cocomplete ∞ -category. Recall that, by Remark 1.2.3, a (non-equivariant) coarse homology theory is the same as a functor of $Fun^{colim}(Spc\mathcal{X}, C)$.

Consider a functor

$$E: GOrb \to Fun^{colum}(Spc\mathcal{X}, C)$$

with values in non-equivariant coarse homology theories and let X be in

$$\mathbf{Fun}(G\mathbf{Orb}^{\mathrm{op}}, \mathbf{Spc}\mathcal{X}).$$

Then, we can consider the induced functor

$$E \circ X : GOrb \times GOrb^{op} \to C$$
(1.4.1)

defined by sending a pair (S, T) to the object E(S)(X(T)) of **C**. We can now apply the coend formula of Definition 1.4.2:

Definition 1.4.3. [BCb, Def. 3.13] As above, let $E: GOrb \to Fun^{colim}(Spc\mathcal{X}, \mathbf{C})$ be a functor with values in a cocomplete stable ∞ -category \mathbf{C} . We define the functor E^G associated to E by

$$E^G(X) \coloneqq \int^{G\mathbf{Orb}} E \circ X.$$

where the index category is GOrb and X is a functor $X: GOrb^{op} \to Spc\mathcal{X}$.

A transformation of functors $X \to X'$ induces a transformation of the pairing $E \circ X \to E \circ X'$ (that uses the functoriality of E(S) as well), hence a transformation between the coend formulas.

Remark 1.4.4. We now briefly explain the motivation behind Definition 1.4.3. Let \mathbf{C} be a cocomplete stable ∞ -category. In equivariant homotopy theory, as a consequence of Elmendorf's theorem [Elm83], a \mathbf{C} -valued equivariant homology theory is essentially uniquely determined by a functor $H: \mathbf{GOrb} \to \mathbf{C}$. There is an equivalence of ∞ -categories between \mathbf{C} and the colimit-preserving functors from the ∞ -category of spaces \mathbf{Spc} to \mathbf{C} , and by using this equivalence, we can interpret H as a functor

$$H: GOrb \to Fun^{colim}(Spc, C).$$

If X is G-space, there is a well-defined pairing $H \circ X$ defined by sending the pair $(S,T) \in GOrb \times GOrb^{op}$ to the object $H(S)(X(T)) \in C$ and its evaluation can be described in terms of a coend formula [BCb, Rmk. 3.11]. This interpretation can be used in order to get abstract localization results [BCb]. In the case of equivariant coarse homology theories and of G-bornological coarse spaces, the ∞ -category **Spc** of spaces is replaced by the ∞ -category **Spc** \mathcal{X} of coarse motivic spaces, which leads to the Definition 1.4.3.

The functor E^G is a coarse homology theory when pre-composed with a suitable orbit functor, as we now explain.

Let G be a group and H a subgroup of G and denote by $W_G(H) := N_G(H)/H$ the associated Weyl group. If X is a G-bornological coarse space, we consider the subset X^H of H-fixed points, which has an induced $W_G(H)$ -action. In order to endow X^H with a compatible coarse structure and bornology, we embed it in the G-completion $B_G X$ of X (see Definition 1.1.23); the set X^H , endowed with the bornology and coarse structure induced by the inclusion $X^H \to B_G X$ (see Example 1.1.14), is a $W_G(H)$ -bornological coarse space. A morphism $f: X \to Y$ of G-bornological coarse spaces restricts to a morphism $f^H: X^H \to Y^H$ of $W_G(H)$ -bornological coarse spaces. Then, we get an H-fixed points functor

$$(-)^H : GBornCoarse \to W_G(H)BornCoarse$$

sending X to the *H*-fixed points X^H of X. The composition with the Yoneda functor $\operatorname{Yo}_{W_G(H)} : W_G(H)$ BornCoarse $\to W_G(H)$ Spc \mathcal{X} (1.2.3) describes a new functor

$$G$$
BornCoarse $\rightarrow W_G(H)$ Spc \mathcal{X}

that can be refined to a *motivic orbit functor* [BCb, Lemma 3.6 & Def. 3.8]

$$Y: GBornCoarse \to Fun(GOrb^{op}, Spc\mathcal{X})$$
(1.4.2)

(roughly) defined by sending a G-bornological coarse space X to the functor which sends the G -set G/H to the motivic coarse space $\operatorname{Yo}(X^H)$ of the bornological coarse space of H-fixed points X^H . In this way, we take the dependence on the subgroup H into account properly; we refer to [BCb, Sec. 3.3] for further details on the definition of the motivic orbit functor.

Let E^G be the functor constructed in Definition 1.4.3. Then, we get a *G*-equivariant coarse homology theory:

Lemma 1.4.5. [BCb, Cor. 3.15] Let $E: GOrb \to Fun^{colim}(Spc\mathcal{X}, \mathbb{C})$ be a functor with values in a stable cocomplete ∞ -category \mathbb{C} . Then, the functor

$$E^G \circ \tilde{Y} \colon G\mathbf{BornCoarse} \to \mathbf{C}$$

is a G-equivariant coarse homology theory.

We do not know whether all the **C**-valued equivariant coarse homology theories are of the form E^G for some functor $E: GOrb \to Fun^{colim}(Spc\mathcal{X}, \mathbb{C})$ [BCb, Rmk. 3.16]. Such functors will be called of Bredon-style:

Definition 1.4.6. [BCb, Def. 3.17] Let $E: GOrb \to Fun^{colim}(Spc\mathcal{X}, \mathbf{C})$ as above. A *G*-equivariant coarse homology theory of the form E^G is called a *Bredon-style equivariant* coarse homology theory.

Bredon-style equivariant coarse homology theories are important to us because of the coarse localization theorems Theorem 1.4.15 and Theorem 1.4.21; in fact, these theorems are stated for Bredon-style equivariant coarse homology theories. However, every equivariant coarse homology theory E can be approximated by a Bredon-style equivariant coarse homology theory E^{Bredon} as follows.

Remark 1.4.7. Let **C** be a stable ∞ -category and let $E \in \operatorname{Fun}^{\operatorname{colim}}(G\operatorname{Spc}\mathcal{X}.\mathbf{C})$ be a functor corresponding to an equivariant coarse homology theory $E \circ \operatorname{Yo}_G$. Consider the functor

$$\underline{E}: GOrb \to Fun^{colum}(Spc\mathcal{X}, C)$$

defined as $\underline{E}(S)(X) := E(S_{\min,\max} \otimes X)$. We can then apply the coend formula to \underline{E} and $X \in \mathbf{Fun}(G\mathbf{Orb}^{\mathrm{op}}, \mathbf{Spc}\mathcal{X})$ as in Definition 1.4.3 obtaining a new functor \underline{E}^G .

We call $Y: GSpc\mathcal{X} \to Fun(GOrb^{op}, Spc\mathcal{X})$ the factorization of the motivic orbit functor (1.4.2) through the Yoneda functor Yo_G.

Definition 1.4.8. [BCb, Def. 3.29] The Bredon-style equivariant coarse homology theory E^{Bredon} associated to a functor $E: GOrb \to Fun^{colim}(Spc\mathcal{X}, \mathbb{C})$ is the composition

$$E^{Bredon} \coloneqq \underline{E}^G \circ Y \colon G\mathbf{Spc}\mathcal{X} \to \mathbf{C}.$$

Consider the following categories:

Definition 1.4.9. [BCb, Def. 3.34] Let $\mathbf{Sp}\mathcal{X}_{bd}$ denote the full subcategory of $\mathbf{Sp}\mathcal{X}$ generated under colimits by the objects $\mathrm{Yo}^{s}(X)$, with X a bounded bornological coarse space.

Definition 1.4.10. [BCb, Def. 3.46] Let

$GSp\mathcal{X}(GOrb\otimes Sp\mathcal{X}_{bd})$

be the full subcategory of $GSp\mathcal{X}$ generated under colimits by the motives $S_{min,max} \otimes X$ for S in GOrb and X in $Sp\mathcal{X}_{bd}$.

Let $E: GOrb \to Fun^{colim}(Spc\mathcal{X}, \mathbb{C})$ be a functor and E^{Bredon} be its associated Beredon-style coarse homology theory. There is a comparison map

$$E^{Bredon} \to E$$
 (1.4.3)

[BCb, Sec. 3.5]. One may ask under which conditions on the functor E (and on the bornological coarse space X) the comparison map (1.4.3) provides an equivalence. By [BCb, Cor. 3.43], this is true if the group G is finite and X belongs to the category $G\mathbf{Sp}\mathcal{X}\langle G\mathbf{Orb}\otimes\mathbf{Sp}\mathcal{X}\rangle$, or, by [BCb, Cor. 3.47], if E is hyperexcisive [BCb, Def. 3.32] and X belongs to the category $G\mathbf{Sp}\mathcal{X}\langle G\mathbf{Orb}\otimes\mathbf{Sp}\mathcal{X}_{bd}\rangle$.

We do not give the definition of hyperexcisive coarse homology theory here; however, if E is a coarse homology theory and $E_{G_{\text{can,min}}}$ denotes its twist by the G-bornological coarse space $G_{\text{can,min}}$, then the following is true:

Remark 1.4.11. [BCb, Lemma 3.37] The twist $E_{G_{\text{can,min}}}$ of a continuous equivariant coarse homology theory E is an hyperexcisive coarse homology theory.

Let G be a discrete group and let \mathcal{F} be a set of subgroups of G that is invariant under conjugation. It determines the full subcategory $G_{\mathcal{F}}\mathbf{Orb}$ of $G\mathbf{Orb}$ of transitive G-sets with stabilizers in \mathcal{F} . Moreover, the inclusion of $G_{\mathcal{F}}\mathbf{Orb}$ in $G\mathbf{Orb}$ induces adjunctions:

$$\operatorname{Ind}_{\mathcal{F}}: \operatorname{Fun}(G_{\mathcal{F}}\operatorname{Orb}^{op}, \mathbf{D}) \leftrightarrows \operatorname{Fun}(G\operatorname{Orb}^{op}, \mathbf{D}): \operatorname{Res}_{\mathcal{F}}$$

and

$$\operatorname{Res}_{\mathcal{F}} : \operatorname{Fun}(G\operatorname{Orb}^{op}, \mathbf{D}) \leftrightarrows \operatorname{Fun}(G_{\mathcal{F}}\operatorname{Orb}^{op}, \mathbf{D}) : \operatorname{Coind}_{\mathcal{F}}$$

for every presentable ∞ -category **D** [Lur09, Corollary 5.5.2.9].

Definition 1.4.12. Let \mathcal{F} be a conjugation invariant set of subgroups of G. Then, $\mathcal{F}^{\perp} := \operatorname{All} \setminus \mathcal{F}$ denotes the complement of \mathcal{F} in the set All of all subgroups of G.

Definition 1.4.13. Let X be in $\mathbf{Fun}(G\mathbf{Orb}^{\mathrm{op}}, \mathbf{D})$, where **D** is a presentable ∞ -category. We define

$$X^{\mathcal{F}} := \operatorname{Ind}_{\mathcal{F}^{\perp}} \operatorname{Res}_{\mathcal{F}^{\perp}} X \tag{1.4.4}$$

and we denote by $X^{\mathcal{F}} \to X$ the morphism induced by the counit of the adjunction (1.4).

Definition 1.4.14. [BCb, Def. 2.6] Let $E: GOrb \to C$ be a functor, with C a stable ∞ -category, and let \mathcal{F} be a conjugation invariant set of subgroups of G. We say that the functor E vanishes on \mathcal{F} if $E(S) \simeq 0$ for all S in $G_{\mathcal{F}}Orb$.

We can now formulate the Coarse Abstract Localization Theorem I. Let \mathcal{F} be a family of subgroups of G and let X be an object of $\operatorname{Fun}(G\operatorname{Orb}^{op}, \operatorname{Spc}\mathcal{X})$. Let $E: G\operatorname{Orb} \to$ $\operatorname{Fun}^{\operatorname{colim}}(\operatorname{Spc}\mathcal{X}, \mathbf{C})$ be a functor with \mathbf{C} a stable cocomplete ∞ -category.

Theorem 1.4.15 (Coarse Abstract Localization Theorem I). [BCb, Thm. 4.1] If E vanishes on \mathcal{F} , then the map $X^{\mathcal{F}} \to X$ induces an equivalence

$$E^G(X^{\mathcal{F}}) \to E^G(X)$$

where E^G is the functor associated to E in Definition 1.4.3.

For a second version of the localization theorem, we first need to fix a conjugacy class γ of G and to restrict to a class of *nice* bornological coarse spaces, as we shall see.

Definition 1.4.16. [BCb, Def. 4.2] Let X be a G-bornological coarse space and γ be a conjugacy class of the group G. Define the G-bornological coarse space of γ -fixed points X^{γ} as the G-invariant subset of X

$$X^{\gamma} \coloneqq \bigcup_{g \in \gamma} X^g$$

with the induced bornology and coarse structures.

A G-invariant subset A of X generates a big family $\{A\}$ (see Example 1.1.27). If F is a functor from GBornCoarse to a cocomplete category, then we let $F(\{A\})$ be the colimit

$$F(\{A\}) \coloneqq \operatorname{colim}_{A' \in \{A\}} F(A').$$

Recall the Yoneda functor $\operatorname{Yo}_G : GBornCoarse \to GSpc\mathcal{X}$ (1.2.3).

Definition 1.4.17. [BCb, Def. 4.3] We say that a G-invariant subset A of X is a *nice* subset if the natural morphism

$$\operatorname{Yo}_G(A) \to \operatorname{Yo}_G(\{A\})$$

is an equivalence.

Recall the construction of the functor $\mathcal{O}_{\text{geom}}^{\infty}$: $GUBC \to GBornCoarse$ of Definition 1.3.8.

Example 1.4.18. [BCb] Let W be a finite G-simplicial complex and γ a conjugacy class of G. Then, W is a uniform G-bornological coarse space with the uniform structure induced from the path metric. The subset W^{γ} of γ -fixed points is a uniform G-bornological coarse space. The cones $\mathcal{O}_{\text{geom}}^{\infty}(W)$ and $\mathcal{O}_{\text{geom}}^{\infty}(W^{\gamma})$ are G-bornological coarse spaces. By [BCb, Lemma 3.4], $\mathcal{O}_{\text{geom}}^{\infty}(W^{\gamma})$ is equivalent to $\mathcal{O}_{\text{geom}}^{\infty}(W)^{\gamma}$; moreover, $\mathcal{O}_{\text{geom}}^{\infty}(W)^{\gamma}$ is a nice subset of $\mathcal{O}_{\text{geom}}^{\infty}(W)$.

Definition 1.4.19. Let γ be a conjugacy class of the group G. Let $\mathcal{F}(\gamma)$ be the set of subgroups of G

$$\mathcal{F}(\gamma) \coloneqq \{ H < G \mid H \cap \gamma = \emptyset \}$$

that do not intersect the conjugacy class γ .

Remark 1.4.20. The set $\mathcal{F}(\gamma)$ is a family of subgroups.

We can finally state the Coarse Abstract Localization Theorem II.

Consider a functor $E: GOrb \to Fun^{colim}(Spc\mathcal{X}, \mathbf{C})$, with \mathbf{C} a stable cocomplete ∞ -category. Let X be a G-bornological coarse space and γ a conjugacy class of G. Let \tilde{Y} be the motivic orbit functor (1.4.2).

Theorem 1.4.21 (Coarse Abstract Localization Theorem II). [BCb, Thm. 4.4] Assume that:

- 1. X^{γ} is a nice subset of X;
- 2. E vanishes on $\mathcal{F}(\gamma)$;

then, the map

$$E^G(\tilde{Y}(X^\gamma)) \to E^G(\tilde{Y}(X))$$

induced by the inclusion $X^{\gamma} \to X$ is an equivalence.

Let E be a **C**-valued equivariant coarse homology theory seen as a functor in $\mathbf{Fun}^{\text{colim}}(G\mathbf{Spc}\mathcal{X}, \mathbf{C})$ and \underline{E} the functor of Remark 1.4.7. We have the following corollary of the coarse localization theorem:

Corollary 1.4.22. [BCb, Cor. 4.9] Let X be a G-bornological coarse space and γ a conjugacy class of G. Let X^{γ} be a nice subset of X and assume that \underline{E} vanishes on $\mathcal{F}(\gamma)$. Then, the morphism

$$E^{Bredon}(\operatorname{Yo}_G(X^{\gamma})) \to E^{Bredon}(\operatorname{Yo}_G(X))$$

induced by the inclusion is an equivalence.

1.5 Equivariant coarse ordinary homology

A first example of equivariant coarse homology theories is given by the coarse version of ordinary homology

 $\mathcal{X}\mathrm{H}^{G}$: GBornCoarse \rightarrow Sp,

that is a spectra-valued equivariant coarse homology theory [BE16, BEKW17]. The goal of this section is to recall its construction.

Let X be a bornological coarse space, $n \in \mathbb{N}$ a natural number, B a bounded set of X, and $x = (x_0, \ldots, x_n)$ a point of X^{n+1} . We say that x meets B if there exists an index $i \in \{0, \ldots, n\}$ such that x_i belongs to B. If U is an entourage of X, we say that x is U-controlled if, for each i and j in $\{0, \ldots, n\}$, the pair (x_i, x_j) belongs to U.

An *n*-chain c on X is a function $c: X^{n+1} \to \mathbb{Z}$; its $support \operatorname{supp}(c)$ is defined as the set of points for which the function c is non-zero:

$$supp(c) = \{ x \in X^{n+1} \mid c(x) \neq 0 \}.$$
(1.5.1)

We say that an *n*-chain *c* is *U*-controlled if every point *x* of $\operatorname{supp}(c)$ is *U*-controlled. The chain *c* is *locally finite* if, for every bounded set *B*, the set of points in $\operatorname{supp}(c)$ which meet *B* is finite. An *n*-chain *c*: $X^{n+1} \to \mathbb{Z}$ is controlled if it is locally finite and *U*-controlled for some entourage *U* of *X*.

Definition 1.5.1. Let X be a bornological coarse space and $n \in \mathbb{N}$ a natural number. Then, $\mathcal{X}C_n(X)$ denotes the free abelian group generated by the locally finite controlled *n*-chains on X.

We will also represent n-chains as formal sums

$$\sum_{x \in X^{n+1}} c(x)x,$$

that are locally finite and U-controlled. The boundary map $\partial \colon \mathcal{X}C_n(X) \to \mathcal{X}C_{n-1}(X)$ is defined as the alternating sum $\partial \coloneqq \sum_i (-1)^i \partial_i$ of the face maps:

$$\partial_i(x_0,\ldots,x_n) \coloneqq (x_0,\ldots,\hat{x_i},\ldots,x_n).$$

As the *n*-chains of $\mathcal{X}C_n(X)$ are locally finite and controlled, then ∂_i extends linearly to a map $\partial_i : \mathcal{X}C_n(X) \to \mathcal{X}C_{n-1}(X)$.

Lemma 1.5.2. [BE16, Sec. 6.3] Let X be a bornological coarse space. The graded abelian group $\mathcal{X}C_*(X)$, endowed with the boundary operator ∂ extended linearly to $\mathcal{X}C_*(X)$, is a chain complex.

When X is a G-bornological coarse space, we let $\mathcal{X}C_n^G(X)$ be the subgroup of $\mathcal{X}C_n(X)$ given by the locally finite controlled *n*-chains that are also G-invariant. The boundary operator restricts to $\mathcal{X}C_*^G(X)$, and $(\mathcal{X}C_*^G(X), \partial)$ is a subcomplex of $(\mathcal{X}C_*(X), \partial)$.

It is easy to see that the chain complexes $\mathcal{X}C_*(X)$ and $\mathcal{X}C^G_*(X)$ are functorial in X; namely, if $f: X \to Y$ is a morphism of G-bornological coarse spaces, then we consider the map on the products $X^n \to Y^n$ sending (x_0, \ldots, x_n) to $(f(x_0), \ldots, f(x_n))$. This extends linearly to a map

$$\mathcal{X}C^G(f) \colon \mathcal{X}C_n^G(X) \to \mathcal{X}C_n^G(Y)$$
 (1.5.2)

that is well-defined because the map f is proper and controlled and because the chains are locally finite. In fact, the map $\mathcal{X}C^G(f)$ involves sums over the pre-images by f, that are finite. Moreover, $\mathcal{X}C^G(f)$ sends controlled *n*-chains to controlled *n*-chains because fis controlled.

We have described a functor

$$\mathcal{X}C^G : GBornCoarse \to Ch$$
 (1.5.3)

with values in the category \mathbf{Ch} of chain complexes over the integers \mathbb{Z} .

The ∞ -category \mathbf{Ch}_{∞} of chain complexes is defined as the localization (in the realm of ∞ -categories) of the nerve of the category \mathbf{Ch} at the class W of quasi-isomorphism of chain complexes:

$$\mathbf{Ch}_{\infty} \coloneqq \mathrm{N}(\mathbf{Ch})[W^{-1}].$$
 (1.5.4)

as described in [Lur14, Sec. 1.3.4].

There is a standard way to go from chain complexes to spectra that uses the equivalence between the ∞ -category of chain complexes \mathbf{Ch}_{∞} and $H\mathbb{Z} - \mathbf{Mod}$ of $H\mathbb{Z}$ -modules [Shi07, Thm. 1.1]. This is the Eilenberg-MacLane correspondence

$$\mathcal{EM}: \mathbf{Ch} \xrightarrow{loc} \mathbf{Ch}_{\infty} \xrightarrow{\simeq} H\mathbb{Z} - \mathbf{Mod} \to \mathbf{Sp}$$
 (1.5.5)

between the ∞ -category of chain complexes and the ∞ -category of spectra, where *loc* is the localization functor (see [BE16, Sec. 6.3]); among the properties of the functor \mathcal{EM} , it is remarkable to say that it sends chain homotopic maps to equivalences, short exact sequences of chain complexes to spectra and it preserves filtered colimits. By post-composing the functor \mathcal{XC}^G with the functor \mathcal{EM} , we get a coarse version of equivariant coarse ordinary homology:

$$\mathcal{X}\mathrm{H}^G \coloneqq \mathcal{E}\mathcal{M} \circ \mathcal{X}C^G \colon G\mathbf{BornCoarse} \to \mathbf{Sp}.$$
 (1.5.6)

Theorem 1.5.3 ([BE16, BEKW17]). The functor $\mathcal{X}H^G$ is a G-equivariant **Sp**-valued coarse homology theory.

Notation 1.5.4. When we want to specify the ring of coefficients, we denote by $\mathcal{X}H^G_B(X)$

the G-equivariant coarse ordinary homology with R coefficients.

Example 1.5.5. Assume that $X = \{*\}$ is the one point space. Then, the chain complex $\mathcal{X}C_*(X)$ has one free generator in each dimension, and the boundary maps are either the null map or the identity, depending on the degree. The coarse homology groups are 0 in positive and negative degree and \mathbb{Z} otherwise. Hence, $\mathcal{X}H(*) \simeq H\mathbb{Z}$.

Example 1.5.6. If $X \in GBornCoarse$ has a trivial *G*-action, then $\mathcal{X}H^G(X) \simeq \mathcal{X}H(X)$.

Example 1.5.7 (Proposition 7.5 [BEKW17]). Let G be a group, and consider the Gbornological coarse space $G_{can,mn}$ endowed with the canonical coarse structure and the minimal bornology as described in Example 1.1.22 (ii); then the homology of the chain complex $\mathcal{X}C^G(G_{can,min})$ is isomorphic to the group homology of G:

 $\mathrm{H}_*(\mathcal{X}C^G(G_{\mathrm{can},\min})) \cong \mathrm{H}_*(G;\mathbb{Z}).$

Chapter 2

The symmetric monoidal category of controlled objects

Let X be a G-bornological coarse space (Definition 1.1.21) and let **A** be an additive category with a strict G-action (Definition A.1.3). From these data we can define a category of objects that are "parametrized" on X: this is the category $V_{\mathbf{A}}^{G}(X)$ of Gequivariant X-controlled **A**-objects [BEKW17, Def. 8.3]. The definition of this category is of fundamental importance for studying properties of bornological coarse spaces; for example, it is a flasque category provided the G-bornological coarse space X is flasque (Lemma 2.1.15), and coarse equivalences induce equivalent categories (Lemma 2.1.14).

The correspondence $X \mapsto V_{\mathbf{A}}^G(X)$ is functorial in the *G*-bornological coarse space X and yields a functor

$V^G_{\mathbf{A}} \colon G\mathbf{BornCoarse} \to \mathbf{Add}$

from the category of G-bornological coarse spaces to the category of small additive categories Add. This functor, when considered with values in the ∞ -category Add $_{\infty}$ of additive categories (see Definition A.3.7), is lax symmetric monoidal [BCa, Thm. 3.26] and this fact allows us to refine many equivariant coarse homology theories to lax symmetric monoidal functors. An example of equivariant coarse homology theory is G-equivariant coarse K-homology $KA\mathcal{X}^G$ [BEKW17, Def. 8.8], defined as the (nonconnective) K-theory of the additive category $V^G_A(X)$ of G-equivariant X-controlled A-objects:

$$K\mathbf{A}\mathcal{X}^G \coloneqq K \circ V^G_\mathbf{A} \colon G\mathbf{BornCoarse} \to \mathbf{Sp}.$$

In Section 2.1, we recall the definition of the category $V_{\mathbf{A}}^G(X)$, we construct the functor $V_{\mathbf{A}}^G: G\mathbf{BornCoarse} \to \mathbf{Add}$ and we review its main properties. In Section 2.2, we we see how to refine the functor $V_{\mathbf{A}}^G$ to a lax symmetric monoidal one. In the last section we recall the definition of equivariant coarse algebraic K-homology as defined in [BEKW17].

2.1 The category of controlled objects

The goal of this section is to recall the definition of the additive category $V_{\mathbf{A}}^G(X)$ of *G*-equivariant *X*-controlled **A**-objects [BEKW17, Def. 8.3] and of the functor

 $V^G_{\mathbf{A}} : G\mathbf{BornCoarse} \to \mathbf{Add}.$

sending a *G*-bornological coarse space to the category $V_{\mathbf{A}}^G(X)$. We will show that close morphisms of bornological coarse spaces (see Definition 1.1.24) induce naturally isomorphic functors between the associated categories of controlled objects (see Lemma 2.1.14). Moreover, we will show that the category $V_{\mathbf{A}}^G(X)$ is a flasque category (see Definition A.1.4) provided the *G*-bornological coarse space X is also flasque (see Lemma 2.1.15).

Our main reference is [BEKW17, Sec. 8.2] where the main results of this section are given.

Let G be a group and let X be a G-bornological coarse space.

Remark 2.1.1. The bornology $\mathcal{B}(X)$ on X defines a poset with the partial order induced by subset inclusion; hence, $\mathcal{B}(X)$ can be seen as a category.

Let \mathbf{A} be an additive category with strict G-action (Definition A.1.3). For every element g in G and every functor $F: \mathcal{B}(X) \to \mathbf{A}$, let $gF: \mathcal{B}(X) \to \mathbf{A}$ denote the functor sending a bounded set B in $\mathcal{B}(X)$ to the \mathbf{A} -object $g(F(g^{-1}(B)))$ (and defined on morphisms $B \subseteq B'$ as the induced morphism of \mathbf{A} $(gF)(B \subseteq B'): gF(g^{-1}(B)) \to gF(g^{-1}(B')))$. If $\eta: F \to F'$ is a natural transformation between two functors $F, F': \mathcal{B}(X) \to \mathbf{A}$, we denote by $g\eta: gF \to gF'$ the induced natural transformation between gF and gF'.

Definition 2.1.2. [BEKW17, Def. 8.3] Let G be a group, X a G-bornological coarse space and **A** an additive category with strict G-action. A G-equivariant X-controlled **A**object is a pair (M, ρ) consisting of a functor $M : \mathcal{B}(X) \to \mathbf{A}$ and a family $\rho = (\rho(g))_{g \in G}$ of natural isomorphisms $\rho(g) : M \to gM$, satisfying the following conditions:

- 1. $M(\emptyset) \cong 0;$
- 2. for all B, B' in $\mathcal{B}(X)$, the commutative diagram

$$\begin{array}{cccc}
M(B \cap B') & \longrightarrow & M(B) \\
\downarrow & & \downarrow \\
M(B') & \longrightarrow & M(B \cup B')
\end{array}$$

is a push-out;

3. for all B in $\mathcal{B}(X)$ there exists a finite subset F of B such that the inclusion induces an isomorphism $M(F) \xrightarrow{\cong} M(B)$; 4. for all elements g, g' in G we have the relation $\rho(gg') = g\rho(g') \circ \rho(g)$, where $g\rho(g')$ is the natural transformation from gM to gg'M induced by $\rho(g')$.

Notation 2.1.3. If (M, ρ) is an X-controlled A-object and x is an element of X, we will often write M(x) istead of $M(\{x\})$ for the value of the functor M at the bounded set $\{x\}$ of X.

The following lemma is a consequence of the above definition:

Lemma 2.1.4. [BEKW17, Lemma 8.4] In the same hypothesis of Definition 2.1.2, let (M, ρ) be a G-equivariant X-controlled **A**-object. Then,

- (i) the canonical morphism $\sum_{x \in F} (M(\{x\} \hookrightarrow F))$: $\bigoplus_{x \in F} M(\{x\}) \to M(F)$ is an isomorphism for every finite subset F of X;
- (ii) for all finite subsets $F \subseteq F'$ of X the following diagram is commutative:

(iii) for each bounded set B of X, there exists a unique minimal finite subset $F_B \subseteq B$ for which the induced morphism $M(F_B) \to M(B)$ is an isomorphism. Moreover, for every subset B' of X with $F_B \subseteq B' \subseteq B$, the morphisms $M(F_B) \to M(B')$ and $M(B') \to M(B)$ are isomorphisms.

Proof. The items (i) and (ii) are clear from Definition 2.1.2.

We prove (iii). Let B be a bounded set of X and suppose that F and F' are two finite subsets of B such that the morphisms $M(F) \to M(B)$ and $M(F') \to M(B)$ induced by the inclusions $F \subseteq B$ and $F' \subseteq B$ are isomorphisms. The square

$$\begin{array}{ccc} M(F \cap F') & \longrightarrow & M(F) \\ & & & \downarrow \cong \\ M(F') & \stackrel{\cong}{\longrightarrow} & M(B) \end{array}$$

is a push-out square and all the maps are inclusions of direct summands. This implies that the morphism $M(F \cap F') \to M(B)$ is an isomorphism; hence, there exists a minimal finite subset F_B of B for which the morphism $M(F_B) \to M(B)$ is an isomorphism.

In a similar way one proves that, for every subset B' of X such that $F_B \subseteq B' \subseteq B$, the morphism $M(F_B) \to M(B')$ is an isomorphism, hence $M(B') \to M(B)$ is an isomorphism as well.

Definition 2.1.5. Let (M, ρ) be an equivariant X-controlled **A**-object. The function σ which sends a bounded set B in $\mathcal{B}(X)$ to the minimal finite subset F_B of Lemma 2.1.4 is called the *support function* of (M, ρ) .

Let X be a G-bornological coarse space. Recall that for every bounded set B of X and every entourage U we can consider the U-thickening U[B] (1.1.2), which is again bounded due to the compatibility of the bornology and coarse structure on X (see Definition 1.1.10). As U preserves the inclusions of bounded sets, this describes a functor

$$U[-]: \mathcal{B}(X) \to \mathcal{B}(X),$$

where $\mathcal{B}(X)$ is the family of bounded sets of X seen as a category (see Remark 2.1.1). Observe that, if U is a G-invariant entourage of X, then U[gB] = g(U[B]) for every g in G.

Definition 2.1.6. [BEKW17, Def. 8.6] Let X be a G-bornological coarse space, let (M, ρ) and (M', ρ') be G-equivariant X-controlled **A**-objects and let $U \in C^G(X)$ be a G-invariant entourage of X. A G-equivariant U-controlled morphism $\varphi \colon (M, \rho) \to (M', \rho')$ is a natural transformation

$$\varphi \colon M(-) \to M' \circ U[-]$$

such that $\rho'(g) \circ \varphi = (g\varphi) \circ \rho(g)$ for all g in G.

The set of G-equivariant U-controlled morphisms $\varphi \colon (M, \rho) \to (M', \rho')$ is denoted by $\operatorname{Mor}_U((M, \rho), (M', \rho')).$

Remark 2.1.7. The set $Mor_U((M, \rho), (M', \rho'))$ is an abelian group with operation induced by **A**.

Let X be a G-bornological coarse space and let U and U' be G-invariant entourages of X with $U \subseteq U'$. For every bounded set B of X, the inclusion $U \subseteq U'$ induces an inclusion $U[B] \subseteq U'[B]$; this yields a natural transformation of functors $M' \circ U[-] \to M' \circ U'[-]$, hence a map

 $\operatorname{Mor}_U((M,\rho), (M',\rho')) \to \operatorname{Mor}_{U'}((M,\rho), (M',\rho'))$

by post-composition.

By using these structure maps we define the abelian group of *G*-equivariant controlled morphisms from (M, ρ) to (M', ρ') as the colimit

$$\operatorname{Hom}_{V^G_{\mathbf{A}}(X)}((M,\rho),(M',\rho')) \coloneqq \operatorname{colim}_{U \in \mathcal{C}^G} \operatorname{Mor}_U((M,\rho),(M',\rho')).$$

We now describe the composition. Consider two morphisms in $\operatorname{Hom}_{V^G_{\mathbf{A}}(X)}((M, \rho), (M', \rho'))$ and $\operatorname{Hom}_{V^G_{\mathbf{A}}(X)}((M', \rho'), (M'', \rho''))$ respectively and let $\varphi \colon M \to M' \circ U[-]$ and $\varphi' \colon M' \to M'' \circ U'[-]$ represent them. By using Remark 1.1.9, we see that φ and φ' induce the following composition of functors

$$M \to M' \circ U[-] \to M'' \circ (U \circ U')[-]$$

where $U \circ U'$ denotes the composition of entourages. The composition of the two morphisms is defined as the morphism of $\operatorname{Hom}_{V^G_{\mathbf{A}}(X)}((M, \rho), (M'', \rho''))$ represented by the composition $M \to M' \circ U[-] \to M'' \circ (U \circ U')[-].$

Definition 2.1.8. Let X be a G-bornological coarse space and let **A** be an additive category with strict G-action. The category $V_{\mathbf{A}}^G(X)$ is the category of G-equivariant X-controlled **A**-objects and G-equivariant controlled morphisms.

Notation 2.1.9. When **A** is the category of finitely generated free *R*-modules, with *R* a commutative ring, then we denote by $V_R^G(X)$ the associated category of *G*-equivariant *X*-controlled (finitely generated free) *R*-modules.

Recall that an additive category is a category enriched on abelian groups, with a zero object and all finite biproducts (Definition A.1.1).

Lemma 2.1.10. [BEKW17, Lemma 8.7] The category $V_{\mathbf{A}}^G(X)$ of G-equivariant X-controlled **A**-objects is an additive category.

Proof. Let X be a G-bornological coarse space with bornology \mathcal{B} . Let (M, ρ) and (M', ρ') be two equivariant controlled objects in $V^G_{\mathbf{A}}(X)$. The category \mathbf{A} is an additive category and it induces a direct sum operation in the functor category $\mathbf{Fun}(\mathcal{B}, \mathbf{A})$. Define the functor $M \oplus M' \colon \mathcal{B} \to \mathbf{A}$ as the direct sum in $\mathbf{Fun}(\mathcal{B}, \mathbf{A})$. In the same way, we define $\rho \oplus \rho'(g) \coloneqq \rho(g) \oplus \rho'(g)$ for every g in G.

The pair $(M \oplus M', \rho \oplus \rho')$ satisfies the conditions of Definition 2.1.2 and is a *G*-equivariant *X*-controlled **A**-object. There are isomorphisms of *G*-sets

 $\operatorname{Hom}_{\operatorname{\mathbf{Fun}}(\mathcal{B},\mathbf{A})}(M \oplus M', C \circ U[-]) \cong \operatorname{Hom}_{\operatorname{\mathbf{Fun}}(\mathcal{B},\mathbf{A})}(M, C \circ U[-]) \times \operatorname{Hom}_{\operatorname{\mathbf{Fun}}(\mathcal{B},\mathbf{A})}(M', C \circ U[-])$

 $\operatorname{Hom}_{\operatorname{\mathbf{Fun}}(\mathcal{B}, \mathbf{A})}(C, M \oplus M' \circ U[-]) \cong \operatorname{Hom}_{\operatorname{\mathbf{Fun}}(\mathcal{B}, \mathbf{A})}(C, M \circ U[-]) \times \operatorname{Hom}_{\operatorname{\mathbf{Fun}}(\mathcal{B}, \mathbf{A})}(C, M' \circ U[-])$

for every G-invariant entourage U of X and G-equivariant X-controlled **A**-object (C, η) . This implies that $(M \oplus M', \rho \oplus \rho')$ is also a biproduct in the category $V_{\mathbf{A}}^{G}(X)$.

The biproduct of morphisms of $V^G_{\mathbf{A}}(X)$ is defined analogously and is induced by the direct sum operation of morphisms in the functor category.

Remark 2.1.11. If the category **A** is a k-linear category with strict G-action, then the category of equivariant X-controlled **A**-objects $V_{\mathbf{A}}^G(X)$ is a k-linear category. For example, the category $V_k^G(X)$ of G-equivariant X-controlled finite dimensional k-vector spaces is a k-linear category.

We now describe the functoriality of $V_{\mathbf{A}}^G(X)$ with respect to X.

Let $f: (X, \mathcal{C}, \mathcal{B}) \to (X', \mathcal{C}', \mathcal{B}')$ be a morphism of *G*-bornological coarse spaces. If (M, ρ) is a *G*-equivariant *X*-controlled **A**-object, we consider the functor

$$f_*M: \mathcal{B}' \to \mathbf{A}$$

defined by $f_*M(B') := M(f^{-1}(B'))$ for every bounded set B' in \mathcal{B}' and defined on morphisms in the canonical way; observe that the functor is well-defined as f is proper. For every g in G, the family of transformation $f_*\rho = ((f_*\rho)(g))_{g\in G}$ is given by the natural isomorphisms $(f_*\rho)(g): f_*M \to g(f_*M)$ with

$$((f_*\rho)(g))(B') \coloneqq \rho(g)(f^{-1}(B')).$$

The pair $f_*(M,\rho) \coloneqq (f_*M, f_*\rho)$ defined in this way is a *G*-equivariant X'-controlled **A**-object [BEKW17, Sec. 8.2]. Assume also that *U* is an invariant entourage of *X* and that $\varphi \colon (M,\rho) \to (M',\rho')$ is an equivariant *U*-controlled morphism (Definition 2.1.6). Then, the set $V := (f \times f)(U)$ is a *G*-invariant entourage of X' as *f* is controlled; moreover, the relation $U[f^{-1}(B')] \subseteq f^{-1}(V[B'])$ is true for all bounded subsets *B'* of X'. Therefore, we obtain an induced V-controlled morphism on X':

$$f_*\varphi \coloneqq \left(f_*M(B') \xrightarrow{\varphi_{f^{-1}(B')}} M(U[f^{-1}(B')]) \to f_*M(V[B'])\right)_{B' \in \mathcal{B}'}.$$
 (2.1.1)

We have just described a functor

$$f_* := V^G_{\mathbf{A}}(f) \colon V^G_{\mathbf{A}}(X) \to V^G_{\mathbf{A}}(X') \tag{2.1.2}$$

between the additive categories of equivariant controlled objects; observe that f_* is also exact.

We denote by

$$V_{\mathbf{A}}^G : G\mathbf{BornCoarse} \to \mathbf{Add}.$$
 (2.1.3)

the functor from the category of G-bornological coarse spaces to the category of small additive categories obtained in this way.

Remark 2.1.12. If **A** is a *k*-linear category, then the functor $V_{\mathbf{A}}^G: G\mathbf{BornCoarse} \to \mathbf{Add}$ refines to a functor $V_{\mathbf{A}}^G: G\mathbf{BornCoarse} \to \mathbf{Cat}_k$ from the category of *G*-bornological coarse spaces to the category of small *k*-linear categories.

We conclude the section we some further properties of the functor $V_{\mathbf{A}}^{G}$.

Remark 2.1.13. Let $(X, \mathcal{C}, \mathcal{B})$ be a *G*-bornological coarse space and $U \in \mathcal{C}^G$ be a *G*invariant entourage of *X*. Then, $X_U \coloneqq (X, \mathcal{C}_U, \mathcal{B})$ is a *G*-bornological coarse space by restriction (see Example 1.1.22 (v)). The natural map $X_U \to X$ induces an additive functor $\Phi_U: V^G_{\mathbf{A}}(X_U) \to V^G_{\mathbf{A}}(X)$ which is the identity on objects as the definition of equivariant X controlled **A**-objects does not depend on the coarse structure. Moreover, the category $V^G_{\mathbf{A}}(X_U)$ can be seen as a subcategory of $V^G_{\mathbf{A}}(X)$. On the other hand, every controlled morphism in $V^G_{\mathbf{A}}(X)$ is U-controlled for some entourage U in \mathcal{C}^G . Therefore, the category $V^G_{\mathbf{A}}(X)$ is the filtered colimit

$$V_{\mathbf{A}}^G(X) \cong \operatorname{colim}_{U \in \mathcal{C}^G} V_{\mathbf{A}}^G(X_U)$$

indexed on the poset of G-invariant entourages of X.

Recall the definition of closeness for morphisms of G-bornological coarse spaces Definition 1.1.24.

Lemma 2.1.14. [BEKW17, Lemma 8.11] Let $f, g: X \to X'$ be two morphisms of Gbornological coarse spaces. If f and g are close to each other, then they induce naturally isomorphic functors $f_* \cong g_*: V^G_{\mathbf{A}}(X) \to V^G_{\mathbf{A}}(X')$.

Proof. As f and g are close to each other, there exists an entourage U' of X' such that $(f,g)(\Delta_X) \subseteq U'$. We can assume that U' is also symmetric (*i.e.*, $U' = (U')^{-1}$). Then, if B' is a bounded set of X', then $f^{-1}(B') \subseteq g^{-1}(U'[B'])$ and $g^{-1}(B') \subseteq f^{-1}(U'[B'])$.

Let (M, ρ) be a G-equivariant X-controlled **A**-object. The morphisms

$$M(f^{-1}(B')) \to M(g^{-1}(U'[B']))$$

and

$$M(g^{-1}(B')) \to M(f^{-1}(U'[B']))$$

induced by the inclusions define natural transformations $\eta: f_*M \to g_*M$ and $\eta': g_*M \to f_*M$. The composition $\eta' \circ \eta$ is the identity $\eta' \circ \eta = \mathrm{id}_{f_*M}$; in fact, it is given by the transformation

$$\left(M(f^{-1}(B') \subseteq f^{-1}((U')^2[B'])) \colon f_*M \to f_*M \circ (U')^2[-]\right)_{B'}$$

where B' runs among the bounded sets of X'. We conclude that η is a natural isomorphism with inverse of the same form.

Recall the notion of flasque category Definition A.1.4.

Lemma 2.1.15. [BEKW17, Lemma 8.13] If X is a flasque G-bornological coarse space, then the category $V_{\mathbf{A}}^G(X)$ of G-equivariant X-controlled **A**-objects is a flasque category.

Proof. The space X is flasque, hence there exists a morphism $f: X \to X$ as required in Definition 1.1.25. Consider the following functor:

$$S \coloneqq \bigoplus_{n \in \mathbb{N}} (f^n)_* \colon V^G_\mathbf{A}(X) \to V^G_\mathbf{A}(X).$$

For every equivariant controlled object (M, ρ) , the direct sum $\bigoplus_n (f^n)_* M$ is well-defined because for every bounded set B there exists an index n for which $(f^n)^{-1}(B) = \emptyset$. The same is true for the family ρ . For a U-controlled morphism $\varphi \colon (M, \rho) \to (M', \rho')$ let $V \coloneqq \bigcup_n (f \times f)^n(U)$; this is an entourage of X by the assumptions on f of Definition 1.1.25. Hence, the map $\bigoplus_n (f^n)_* \varphi$ is V-controlled and S describes in fact an endofunctor.

In order to conclude, observe that, as f is close to the identity $\operatorname{id}_{V_{\mathbf{A}}^G(X)}$, then the functor $f_* \circ S$ is naturally isomorphic to S. Hence, $\operatorname{id}_{V_{\mathbf{A}}^G(X)} \oplus S \cong \operatorname{id}_{V_{\mathbf{A}}^G(X)} \oplus (f_* \circ S) \cong S$. \Box

2.2 The symmetric monoidal refinement of $V_{\mathbf{A}}^{G}$

The goal of this section is to explain how to refine the functor

$$V^G_{\mathbf{A}} \colon G\mathbf{BornCoarse} o \mathbf{Add}$$

of (2.1.3) to a lax symmetric monoidal functor (see Definition A.3.7). This is [BCa, Thm. 3.26] and allow us to refine several equivariant coarse homology theories to lax symmetric monoidal functors; for the sake of completeness, we report the entire construction. In order to describe this lax symmetric monoidal refinement, we will also describe the Grothendieck construction associated to the functor $V_{\mathbf{A}}^G$.

Let \mathbf{A} be a small additive category with a strict *G*-action (Definition A.1.3).

Assumption 2.2.1. For the rest of the section we assume that the additive category \mathbf{A} has a symmetric monoidal structure and that the strict action of G on \mathbf{A} has a refinement to an action by symmetric monoidal functors.

Let **C** be a small category and let $F: \mathbf{C} \to \mathbf{Cat}$ be a functor from **C** to the category of small categories **Cat**. We recall that the Grothendieck construction for F provides a category \mathcal{F} and a projection (a cocartesian fibration) $\pi_F: \mathcal{F} \to \mathbf{C}$. We spell this out in the case in which **C** is the category of G-bornological coarse spaces and F is the functor $V_{\mathbf{A}}^G: G\mathbf{BornCoarse} \to \mathbf{Add}$ (2.1.3) viewed as a functor from $G\mathbf{BornCoarse}$ to the category of small categories **Cat**:

Definition 2.2.2. The *Grothendieck construction* for the functor $V_{\mathbf{A}}^G : G\mathbf{BornCoarse} \to \mathbf{Cat}$ is the category $\mathcal{V}_{\mathbf{A}}^G$ defined as follows:

- 1. The objects of $\mathcal{V}_{\mathbf{A}}^{G}$ are pairs $(X, (M, \rho))$ where X is a G-bornological coarse space in **GBornCoarse** and (M, ρ) is a G-equivariant X-controlled **A**-object in $V_{\mathbf{A}}^{G}(X)$.
- 2. A morphism $(f, \varphi) : (X, (M, \rho)) \to (X', (M', \rho'))$ in $\mathcal{V}^G_{\mathbf{A}}$ consists of a morphism $f : X \to X'$ in **GBornCoarse** and of a morphism $\varphi : f_*(M, \rho) \to (M', \rho')$ in $V^G_{\mathbf{A}}(X')$.

3. The composition of morphisms is given by

$$(f',\varphi')\circ(f,\varphi):=(f'\circ f,\varphi'\circ f'_*(\varphi))$$
.

The projection

$$\pi_{V^G_{\mathbf{A}}}: \mathcal{V}^G_{\mathbf{A}} \to G\mathbf{BornCoarse} \ , \quad (X, (M, \rho)) \mapsto X$$

is the functor that forgets the second component.

Let **C** be a small category and let $F: \mathbf{C} \to \mathbf{Cat}$ be a functor to the category of small categories **Cat**. Let $\pi_F: \mathcal{F} \to \mathbf{C}$ be the associated projection (via the Grothendieck construction). Assume that the categories **C** and \mathcal{F} have symmetric monoidal structures and that the projection π_F preserves the tensor product strictly:

$$\pi_F((X,A)\otimes_{\mathcal{F}} (Y,B)) = X \otimes_{\mathbf{C}} X'$$

Then, the tensor product $(X, A) \otimes_{\mathcal{F}} (Y, B)$ can be written as $(X \otimes_{\mathbf{C}} X', A \boxtimes_{X,X'} B)$ for some object $A \boxtimes_{X,X'} B$ of $F(X \otimes_{\mathbf{C}} X')$. For every pair of objects X and X' of **C**, we get a bifunctor

$$\boxtimes_{X,X'} \colon F(X) \times F(X') \to F(X \otimes_{\mathbf{C}} X')$$

and for every pair of morphisms $f: X \to X'$ and $g: Y \to Y'$, an induced morphism

$$F(f \otimes_{\mathbf{C}} g)(A \boxtimes_{X,Y} B) \to F(f)(A) \boxtimes_{X',Y'} F(g)(B)$$
(2.2.1)

where A is in F(X) and B in F(Y) (see [BCa, Sec. 2.2] for further details).

Assume now that the functor F is actually a functor $F: \mathbb{C} \to \mathbf{Add}$ from a symmetric monoidal category \mathbb{C} to the category of small additive categories \mathbf{Add} endowed with the symmetric monoidal structure of Definition A.3.2. Denote by $\mathbf{Add}_{\infty}^{\otimes}$ the resulting symmetric monoidal ∞ -category of additive categories (see Definition A.3.7 and the text below). Assume that also the Grothendieck construction \mathcal{F} of F has a symmetric monoidal structure. Then, with some mild assumptions, the functor F can be refined to a symmetric monoidal functor in the sense of Definition A.3.7:

Theorem 2.2.3. [BCa, Thm. 2.3] Let **C** be a symmetric monoidal category and let $F: \mathbf{C} \to \mathbf{Add}$ be a functor to the category of small additive categories **Add**. Assume that:

- 1. the Grothendieck construction \mathcal{F} for the functor F has a symmetric monoidal structure;
- 2. the projection $\pi_F \colon \mathcal{F} \to \mathbf{C}$ preserves the tensor product, the tensor unit as well as the associator, unit, and symmetry transformations.

- 3. The functors $\boxtimes_{X,X'}$ are bi-additive for all X, X' in **C**.
- 4. For objects (X, A) and (Y, B) in \mathcal{F} and morphisms $f : X \to X'$ and $g : Y \to Y'$ in \mathbf{C} , the morphism (2.2.1)

$$F(f \otimes_{\mathbf{C}} g)(A \boxtimes_{X,Y} B) \to F(f)(A) \boxtimes_{X',Y'} F(g)(B)$$

is an isomorphism.

Then, the data provide a lax symmetric monoidal refinement (Definition A.3.7)

$$F^{\otimes}: \mathbf{N}(\mathbf{C}^{\otimes}) \to \mathbf{Add}_{\infty}^{\otimes}$$

of the functor F.

Before explaining how to refine the functor $V_{\mathbf{A}}^G: G\mathbf{BornCoarse} \to \mathbf{Add}$ to a lax symmetric monoidal functor by using the theorem above, we need a more explicit description of the morphisms in the category $V_{\mathbf{A}}^G(X)$ [BCa, Sec. 3.3].

We use the following convention:

Convention 2.2.4. Let \mathbf{A} be an additive category. If $(A_i)_{i \in I}$ is a family of objects of \mathbf{A} with at most finitely many non-zero members, then we use the symbol $\bigoplus_{i \in I} A_i$ in order to denote a choice of an object of \mathbf{A} that comes together with a family of morphisms $(A_j \to \bigoplus_{i \in I} A_i)_{j \in I}$ and represents the coproduct of the family.

Since in an additive category finite coproducts and products coincide, for every j in I we furthermore have a canonical projection

$$\bigoplus_{i \in I} A_i \to A_j$$

such that the diagram

$$A_j \xrightarrow{\operatorname{id}_{A_j}} A_i \xrightarrow{} A_j$$

commutes.

If $(A'_{i'})_{i' \in I'}$ is a second family of this type and $(\varphi_{i,i'} : A'_{i'} \to A_i)_{(i',i) \in I' \times I}$ is a family of morphisms in **A**, then we have a unique morphism $\oplus \varphi_{i,i'}$ such that the squares

$$\begin{array}{c} A'_{i'} \xrightarrow{\varphi_{i,i'}} A_i \\ \downarrow & \downarrow \\ \bigoplus_{i' \in I'} A'_{i'} \xrightarrow{\oplus \varphi_{i,i'}} \bigoplus_{i \in I} A_i \end{array}$$

commute for every i' in I' and i in I (where the vertical morphisms are the inclusions of summands).

Let X be a G-bornological coarse space and let (M, ρ) be an equivariant X-controlled **A**-object. Let B be a bounded set of X and let x be a point in B. The inclusion $\{x\} \to B$ induces a morphism $M(\{x\}) \to M(B)$ of **A**. The conditions of Definition 2.1.2 imply that $M(\{x\}) = 0$ for all but finitely many points of B and that the canonical morphism (induced by the universal property of the coproduct in **A**)

$$\bigoplus_{x \in B} M(\{x\}) \xrightarrow{\cong} M(B)$$
(2.2.2)

is an isomorphism.

Let U be an invariant entourage of X and let $\varphi : (M, \rho) \to (M', \rho')$ be an equivariant U-controlled morphism, *i.e.*, a natural transformation of functors $\varphi : M \to M' \circ U[-]$ satisfying an equivariance condition (see Definition 2.1.6). For every point x of X, the transformation φ induces a morphism

$$M(\lbrace x \rbrace) \xrightarrow{\varphi_{\lbrace x \rbrace}} M'(U[\lbrace x \rbrace]) \stackrel{(2.2.2)}{\cong} \bigoplus_{x' \in U[\lbrace x \rbrace]} M(\lbrace x' \rbrace) .$$

$$(2.2.3)$$

We let

$$\varphi_{x',x}: M(\{x\}) \to M'(\{x'\})$$
 (2.2.4)

denote the composition of (2.2.3) with the projection onto the summand corresponding to x'. For every pair of points x, x' of X we get a morphism $\varphi_{x',x}$, hence a family of morphisms $(\varphi_{x',x})_{x',x\in X}$ in **A**. In a similar way, for g in G, the transformation $\rho(g): M \to gM$ gives rise to a family of morphisms

$$\left(\rho(g)_x : M(\{x\}) \to gM(\{g^{-1}x\})\right)_{x \in X}$$
 (2.2.5)

By construction the family $(\varphi_{x',x})_{x',x\in X}$ satisfies the following conditions.

- 1. For all x, x' in X the condition $\varphi_{x',x} \neq 0$ implies that $(x', x) \in U$.
- 2. We have $\rho'(g)_{x'} \circ \varphi_{x',x} = (g\varphi)_{q^{-1}x',q^{-1}x} \circ \rho(g)_x$ for all x, x' in X and g in G.

Lemma 2.2.5. [BCa, Lemma 3.14] Let X be a G-bornological coarse space and let U be a G-invariant entourage of X. The, we have a bijection between the equivariant U-controlled morphisms $\varphi : (M, \rho) \to (M', \rho')$ and the families $(\varphi_{x',x})_{x',x \in X}$ of morphisms as in (2.2.4) satisfying the conditions 1 and 2.

Let $f : X_0 \to X_1$ be a morphism of *G*-bornological coarse spaces and (M_i, ρ_i) be objects of $\mathcal{V}^G_{\mathbf{A}}(X_i)$ for i = 0, 1. Then, a morphism

$$\varphi: f_*(M_0, \rho_0) \to (M_1, \rho_0)$$
 (2.2.6)

induces a matrix

$$\left(\varphi_{x_1,x_0}^f: M_0(\{x_0\}) \to M_1(\{x_1\})\right)_{x_0 \in X_0, x_1 \in X_1} . \tag{2.2.7}$$

In fact, we observe that

$$(f_*M_0)(\{x_1'\}) = M_0(f^{-1}(\{x_1'\})) \stackrel{(2.2.2)}{\cong} \bigoplus_{x_0 \in f^{-1}(\{x_1'\})} M_0(\{x_0\})$$

so that, for x_1, x'_1 in X', the induced morphism $\varphi_{x_1,x'_1} \colon (f_*M_0)(\{x'_1\}) \to M_1(x_1)$ is written as sums $\bigoplus_{x_0 \in f^{-1}(\{x'_1\})} \varphi^f_{x_0,x_1} \colon M_0(f^{-1}(\{x'_1\})) \cong \bigoplus_{x_0 \in f^{-1}(\{x'_1\})} M_0(\{x_0\}) \to M_1(\{x_1\})$ and

$$\left(\varphi_{x_1,x_1'} := \bigoplus_{x_0 \in f^{-1}(\{x_1'\})} \varphi_{x_0,x_1}^f : M_0(f^{-1}(\{x_1'\})) \cong \bigoplus_{x_0 \in f^{-1}(\{x_1'\})} M_0(\{x_0\}) \to M_1(\{x_1\})\right)_{x_1',x_1 \in X_1}$$

is the matrix representing φ according to Lemma 2.2.5. The following, is a consequence of Lemma 2.2.5:

Corollary 2.2.6. [BCa, Cor. 3.15] A matrix (2.2.7) represents a morphism (2.2.6) iff the following conditions are satisfied:

- 1. There exists an entourage U_1 in $\mathcal{C}(X_1)$ such that for every x_0 in X_0 and x_1 in X_1 the condition $\varphi_{x_1,x_0}^f \neq 0$ implies that $(x_1, f(x_0)) \in U_1$.
- 2. For every g in G we have the equality

$$\rho_1(g)_{x_1} \circ \varphi^f_{x_1, x_0} = (g\varphi^f)_{g^{-1}x_1, g^{-1}x_0} \circ \rho(g)_{x_0} .$$

With this explicit description of the morphisms of $V_{\mathbf{A}}^G(X)$, we can now define the symmetric monoidal structure on the Grothendieck construction $\mathcal{V}_{\mathbf{A}}^G$ for the functor $V_{\mathbf{A}}^G: G\mathbf{BornCoarse} \to \mathbf{Cat}$ (se Definition 2.2.2). First, we describe the bifunctor

$$-\otimes_{\mathcal{V}_{\mathbf{A}}^{G}} : \mathcal{V}_{\mathbf{A}}^{G} \times \mathcal{V}_{\mathbf{A}}^{G} \to \mathcal{V}_{\mathbf{A}}^{G}$$

$$(2.2.8)$$

on objects.

Recall that the category *G***BornCoarse** of *G*-bornological coarse spaces has a symmetric monoidal structure as reviewed in Section 1.1. Let $(X, (M, \rho))$ and $(X', (M', \rho'))$ be objects in $\mathcal{V}^G_{\mathbf{A}}$ and define the functor

$$M \boxtimes M' : \mathcal{B}(X \otimes X') \to \mathbf{A}$$
(2.2.9)

(where $\mathcal{B}(X \otimes X')$ denotes the (category associated to the) bornology of the product of *G*-bornological coarse spaces $X \otimes X'$) as follows:

1. For every B in $\mathcal{B}(X \otimes X')$ we set (see Convention 2.2.4)

$$(M \boxtimes M')(B) := \bigoplus_{(x,x') \in B} M(\{x\}) \otimes_{\mathbf{A}} M'(\{x'\}) .$$
 (2.2.10)

Note that the sum has only finitely many non-zero summands because of Definition 2.1.2.

2. If B' is in $\mathcal{B}(X \otimes X')$ such that $B' \subseteq B$, then the morphism

 $(M\boxtimes M')(B'\subseteq B)\colon (M\boxtimes M')(B')\to (M\boxtimes M')(B)$

is given by the canonical map

$$\bigoplus_{(x,x')\in B'} M(\{x\}) \otimes_{\mathcal{A}} M'(\{x'\}) \to \bigoplus_{(x,x')\in B} M(\{x\}) \otimes_{\mathcal{A}} M'(\{x'\})$$

as described in Convention 2.2.4.

By using our Convention 2.2.4 and the universal property of the direct sum, one easily checks that this describes a functor satisfying the first three conditions of Definition 2.1.2.

The family $\rho \boxtimes \rho'$ is analogously defined as the sum $\bigoplus_{(x,x')\in B} \rho(g)_x \otimes \rho'(g)_{x'}$:

$$(\rho \boxtimes \rho')(g)_B \colon \bigoplus_{(x,x') \in B} M(\{x\}) \otimes_{\mathcal{A}} M'(\{x'\}) \to \bigoplus_{(x,x') \in B} gM(\{g^{-1}x\}) \otimes_{\mathcal{A}} gM'(\{g^{-1}x'\})$$

where we use the notation (2.2.5). For every object $B \in \mathcal{B}(X \otimes X')$ the map $(\rho \boxtimes \rho')(g)_B$ provides an isomorphism as $\rho(g)_x$ and $\rho'(g)_{x'}$ are isomorphisms for every $(x, x') \in B$. By using our Convention 2.2.4 and the universal property of the direct sum, it is easy to see that the assignment is functorial in B and this provides a family of natural isomorphisms.

Remark 2.2.7. The pair $(M \boxtimes M', \rho \boxtimes \rho')$ satisfies the remaining condition of Definition 2.1.2 and therefore belongs to $V_{\mathbf{A}}^G(X \otimes X')$.

We now describe the bifunctor $\otimes_{\mathcal{V}_{\mathbf{A}}^G}$ on morphisms. Let $(f, \varphi) : (X_0, (M_0, \rho_0)) \to (X_1, (M_1, \rho_1))$ be a morphism in $\mathcal{V}_{\mathbf{A}}^G$ (see Definition 2.2.2 (2)). Then, we define the morphism

$$(g,\psi) := (f,\varphi) \otimes_{\mathcal{V}_{\mathbf{A}}^G} (X', (M', \rho')) : (X_0 \otimes X', (M_0 \boxtimes M', \rho_0 \boxtimes \rho')) \to (X_1 \otimes X', (M_1 \boxtimes M', \rho_1 \boxtimes \rho'))$$

as follows.

1. We set $g := f \otimes id_{X'} : X_0 \otimes X' \to X_1 \otimes X'$ by using the tensor product in **GBornCoarse**.

2. In order to describe the morphism

$$\psi: (f \otimes \operatorname{id}_{X'})_*(M_0 \boxtimes M', \rho_0 \boxtimes \rho') \to (M_1 \boxtimes M', \rho_1 \boxtimes \rho')$$

we use Corollary 2.2.6, hence we have to describe the matrix

$$(\psi_{(x_1,y'),(x_0,x')}^{f \otimes \operatorname{id}_{X'}})_{(x_0,x') \in X_0 \times X',(x_1,y') \in X_1 \times X'}$$

By definition of $M_0 \boxtimes M'$, for x_0 in X_0 and x' in X', we have

$$(M_0 \boxtimes M')((x_0, x')) \cong M(\{x_0\}) \otimes_{\mathcal{A}} M'(\{x'\})$$

so that we can set

$$\psi_{(x_1,y'),(x_0,x')}^{f \otimes \operatorname{id}_{X'}} := \varphi_{x_1,x_0}^f \otimes_{\mathcal{A}} (\operatorname{id}_{(M',\rho')})_{y',x'} : M_0(\{x_0\}) \otimes_{\mathcal{A}} M'(\{x'\}) \to M_1(\{x_1\}) \otimes_{\mathcal{A}} M'(\{y'\})$$

This matrix satisfies the conditions listed in Corollary 2.2.6 and represents the desired morphism.

In a similar way, we define $(X, (M, \rho)) \otimes (f', \varphi')$ for a morphism $(f', \varphi') : (X'_0, (M'_0, \rho'_0)) \to (X'_1, (M'_1, \rho_1))$. Moreover, this description is compatible with compositions.

Definition 2.2.8. [BCa, Def. 3.18 & 3.19] Let $\mathcal{V}^G_{\mathbf{A}}$ be the Grothendieck construction of Definition 2.2.2. The bifunctor $\otimes_{\mathcal{V}^G_{\mathbf{A}}} (2.2.8)$ is defined on objects $(X, (M, \rho)), (X', (M', \rho'))$ of $\mathcal{V}^G_{\mathbf{A}}$ by

 $(X, (M, \rho)) \otimes_{\mathcal{V}^G_{\bullet}} (X', (M', \rho')) := (X \otimes X', (M \boxtimes M', \rho \boxtimes \rho'))$

where \otimes denotes the tensor product of *GBornCoarse* and on morphisms by the preceding description.

We refer to [BCa] for the descriptions of the tensor unit $1_{\mathcal{V}_{\mathbf{A}}^G}$, of the associative constrain $\alpha^{\mathcal{V}_{\mathbf{A}}^G}$, of the unit constraint $\eta^{\mathcal{V}_{\mathbf{A}}^G}$, the symmetry constraint $\sigma^{\mathcal{V}_{\mathbf{A}}^G}$ (see Definition A.3.1), as they are defined by the same reasoning.

Proposition 2.2.9. [BCa, Prop. 3.23] Let $\mathcal{V}_{\mathbf{A}}^G$ be the Grothendieck construction for the functor $V_{\mathbf{A}}^G$: GBornCoarse \rightarrow Cat and let $\pi : \mathcal{V}_{\mathbf{A}}^G \rightarrow$ GBornCoarse be the associated projection. Then:

- 1. the functor $\otimes_{\mathcal{V}_{\mathbf{A}}^{G}} and$ the object $1_{\mathcal{V}_{\mathbf{A}}^{G}}$ together with the natural isomorphisms $\alpha^{\mathcal{V}_{\mathbf{A}}^{G}}$, $\eta^{\mathcal{V}_{\mathbf{A}}^{G}}$ and $\sigma^{\mathcal{V}_{\mathbf{A}}^{G}}$ define a symmetric monoidal structure on $\mathcal{V}_{\mathbf{A}}^{G}$;
- 2. the functor $\pi: \mathcal{V}_{\mathbf{A}}^G \to G\mathbf{BornCoarse}$ preserves the tensor product strictly and the tensor unit as well as the associator, unit, and symmetry transformations.

The categories $\mathcal{V}^{G}_{\mathbf{A}}$ and **GBornCoarse** have symmetric monoidal structures such that the projection π preserves the tensor product strictly. This implies that, for all

G-bornological coarse spaces X and X', we obtain a bifunctor

$$\boxtimes_{X,X'} : V^G_{\mathbf{A}}(X) \times V^G_{\mathbf{A}}(X') \to V^G_{\mathbf{A}}(X \otimes X')$$
(2.2.11)

that preserves also the linear structure:

Proposition 2.2.10. [BCa, Prop. 3.24] Let X, X' be G-bornological coarse spaces. Then, the functor

$$\boxtimes_{X,X'}: V^G_{\mathbf{A}}(X) \times V^G_{\mathbf{A}}(X') \to V^G_{\mathbf{A}}(X \otimes X')$$

is additive in both variables.

Proof. Let (M_i, ρ_i) be in $V^G_{\mathbf{A}}(X)$ for i = 0, 1 and (M', ρ) be in $V^G_{\mathbf{A}}(X')$. In view of the symmetry it suffices to show that the canonical morphism

$$(M_0 \boxtimes_{X,X'} M') \oplus (M_1 \boxtimes_{X,X'} M') \to (M_0 \oplus M_1) \boxtimes_{X,X'} M'$$

is an isomorphism. It suffices to show that

$$\left[(M_0 \boxtimes_{X,X'} M') \oplus (M_1 \boxtimes_{X,X'} M') \right] (\{(x,x')\}) \to \left[(M_0 \oplus M_1) \boxtimes_{X,X'} M' \right] (\{(x,x')\})$$

is an isomorphism for every point (x, x') in $X \times X'$. By inserting the definitions we see that this morphism is the same as

$$(M_0(\{x\}) \otimes_{\mathbf{A}} M'(\{x'\})) \oplus (M_1(\{x\}) \otimes_{\mathbf{A}} M'(\{x'\})) \to (M_0(\{x\}) \oplus M_1(\{x\})) \otimes_{\mathbf{A}} M'(\{x'\}) .$$

But this last morphism is an isomorphism since the tensor product in \mathbf{A} is additive in the first argument.

Remark 2.2.11. If the category **A** is also k-linear, then bifunctor $\boxtimes_{X,X'}$ preserves the k-linear structure as well.

We can now apply Theorem 2.2.3 to the functor $V_{\mathbf{A}}^G: G\mathbf{BornCoarse} \to \mathbf{Cat}$. In fact, the Grothendieck construction $\mathcal{V}_{\mathbf{A}}^G$ for the functor $V_{\mathbf{A}}^G$, seen as a functor to the category of small categories, has a symmetric monoidal structure by Proposition 2.2.9; the projection $\pi: \mathcal{V}_{\mathbf{A}}^G \to G\mathbf{BornCoarse}$ preserves the tensor product strictly and the tensor unit as well as the associator, unit, and symmetry transformations. By Proposition 2.2.10, $\boxtimes_{X,X'}: V_{\mathbf{A}}^G(X) \times V_{\mathbf{A}}^G(X') \to V_{\mathbf{A}}^G(X \otimes X')$ is bi-additive for all $X, X' \in G\mathbf{BornCoarse}$. By [BCa, Lemma 3.25], Theorem 2.2.3 (2.2.1) is also satisfied and the data provide a lax symmetric monoidal refinement as in Definition A.3.7:

Theorem 2.2.12. [BCa, Thm. 3.26] If **A** is a symmetric monoidal additive category with a strict action of G by symmetric monoidal functors, then the functor $V_{\mathbf{A}}^G: \mathbf{GBornCoarse} \to \mathbf{Add}$ admits a refinement to a symmetric monoidal functor. **Remark 2.2.13.** If **A** is a symmetric monoidal k-linear category with a strict action of G by symmetric monoidal functors, then the functor $V_{\mathbf{A}}^G : G\mathbf{BornCoarse} \to \mathbf{Cat}_k$ takes values in small additive categories that are in addition k-linear. Hence, by Theorem 2.2.12 and [Lur14, Prop. 2.2.1.1], the functor $V_{\mathbf{A}}^G$ admits a refinement to a symmetric monoidal functor.

2.3 Equivariant coarse algebraic *K*-homology

In this section we recall the definition of equivariant coarse algebraic K-homology as defined in [BEKW17]. This is a coarse homology theory

$$KA\mathcal{X}^G : GBornCoarse \to Sp.$$

with values in the ∞ -category **Sp** of spectra and is defined as the (non-connective) Ktheory of the category $V_{\mathbf{A}}^G(X)$ of G-equivariant X-controlled **A**-objects of Definition 2.1.2. We remark that the equivariant coarse K-homology has been studied in several works [BE16, BEKW17, BC17, BEKW18]. In this section we briefly give its definition and we state the most important properties for us, referring to those papers for further details.

Let \mathbf{Add} be the category of small additive categories and exact functors (see Definition A.1.1). Consider the non-connective version of K-theory of additive categories

$$K: \mathbf{Add} \to \mathbf{Sp};$$

a construction (for more general exact categories) of this functor has been given by Schlichting [Sch06]. We list some of its properties:

- if R is a ring, then the K-theory of the additive category of finitely generated free R-modules is equivalent to the non-connective K-theory of the ring R;
- it sends isomorphic exact functors to equivalences of spectra;
- it commutes with filtered colimits of categories;
- it satisfies additivity and is a localizing invariant (see Section 3.2 for a description of these properties in the case of Hochschild homology);
- it sends flasque categories to zero;
- it is a lax symmetric monoidal functor.

Recall Definition A.1.3 of an additive category with strict G-action.

Definition 2.3.1. [BEKW17, Def. 8.8] Let G be a group and let \mathbf{A} be an additive category with strict G-action. The G-equivariant coarse algebraic K-homology associated

to **A** is the *K*-theory of the additive category of **A**-controlled objects:

$$KA\mathcal{X}^G \coloneqq K \circ V^G_A \colon GBornCoarse \to Sp.$$

Notation 2.3.2. When A is the category of finitely generated free *R*-modules, we denote by $K\mathcal{X}_R^G$ the associated *G*-equivariant *K*-homology.

The above properties of the K-theory functor are used in order to prove that the composition $K\mathbf{A}\mathcal{X}^G$ provides a coarse homology theory:

Theorem 2.3.3. [BEKW17, Thm. 8.9] Let G be a group and let **A** be an additive category with strict G-action. Then, the functor KAX^G is a G-equivariant **Sp**-valued coarse homology theory.

By Theorem 2.2.12, the functor $V_{\mathbf{A}}^G : G\mathbf{BornCoarse} \to \mathbf{Add}$ admits a lax symmetric monoidal refinement. As a consequence, coarse equivariant K-homology is a lax symmetric monoidal functor as well:

Theorem 2.3.4. [BCa, Thm. 3.27] Let G be a group and let \mathbf{A} be an additive category with strict G-action. The functor $K\mathbf{A}\mathcal{X}^G$ admits a lax symmetric monoidal refinement.

For a proof of this theorem we refer to [BCa, Sec. 3.5].

In Farrell-Jones type questions, the twist (see Definition 1.2.5) of coarse algebraic K-homology by the motivic coarse space $\operatorname{Yo}_G^s(G_{\operatorname{can,min}})$ associated to a group G (see Example 1.1.22 (ii)) is of fundamental importance. In order to state this computation, we use the following category:

Definition 2.3.5. [BR07, Def. 2.1] Let **A** be an additive category with strict *G*-action. Let *X* be a *G*-set. The additive category $\mathbf{A} *_G X$ is described as follows:

Objects: an object of $\mathbf{A} *_G X$ is a family $A = (A_x)_{x \in X}$ of objects of \mathbf{A} such that the set $\{x \in X \mid A_x \neq 0\}$ is finite;

Morphisms: a morphism $\varphi: A \to B$ between A and B is a family of morphisms $\varphi = (\varphi_{x,g})_{(x,g) \in X \times G}$ where $\varphi_{x,g}: A_x \to g(B_{g^{-1}x})$ is a morphism in \mathbf{A} and the set of pairs (x,g) for which $\varphi_{x,g} \neq 0$ is finite.

The composition is defined as the convolution product and the addition component-wise.

We recall that every G-set X, endowed with the minimal coarse structure and the maximal bornology, belongs to GBornCoarse (Example 1.1.13).

Proposition 2.3.6. [BEKW17, Prop. 8.24] For every G-set X we have an equivalence

 $V_{\mathbf{A}}^G(X_{min,max} \otimes G_{\operatorname{can,min}}) \simeq \mathbf{A} *_G X.$

We refer to [BEKW17, Prop. 8.24] for further details and the proof. As a consequence of this proposition, we have:

Remark 2.3.7. Let G be a group and H a subgroup of G. Let G/H be endowed with the minimal coarse structure and the maximal bornology; then the previous proposition yields the following computation:

$$K\mathbf{A}\mathcal{X}^G((G/H)_{\min,\max}\otimes G_{\operatorname{can},\min})\simeq K(\mathbf{A}*_G(G/H))$$

When **A** is the additive category of finitely generated free *R*-modules, *R* a ring, the category $\mathbf{A} *_G (G/H)$ is equivalent to the category of finitely generated free R[H]-modules.

Chapter 3

A coarse version of Hochschild and cyclic homology

The goal of this chapter is to define equivariant coarse versions

 $\mathcal{X} \mathrm{HH}_k^G \colon G\mathbf{BornCoarse} o \mathbf{Ch}_\infty$

and

 $\mathcal{X}\mathrm{HC}_k^G\colon G\mathbf{BornCoarse} \to \mathbf{Ch}_\infty$

of the classical Hochschild and cyclic homology for k-algebras [Lod98, Sec. 1.1 & 2.1], where k is a field.

Our definition of equivariant coarse Hochschild and cyclic homology is analogous to the definition of equivariant coarse algebraic K-homology [BEKW17, BC17] recalled in Definition 2.3.1. In fact, we first consider the functor $V_k^G : GBornCoarse \to Cat_k$ (see Remark 2.1.12) sending a G-bornological coarse space X to the k-linear category $V_k^G(X)$ of G-equivariant X-controlled finite dimensional k-vector spaces (see Definition 2.1.2 and Notation 2.1.9). Then, we define equivariant coarse Hochschild (and cyclic) homology as Keller's Hochschild (and cyclic) homology [Kel99] of the k-linear category $V_k^G(X)$ (see Definition 3.4.6).

The chapter is structured as follows. In Section 3.2 we recall Keller's definition of Hochschild and cyclic homology for k-linear exact categories; as this definition uses Kassel's approach to cyclic homology via mixed complexes [Kas87], we start the chapter in Section 3.1 by reviewing the main properties of the (∞ -)category of mixed complexes. In Section 3.3, we prove our main result Theorem 3.4.2, where we show that the composition of Keller's Hochschild homology with the functor V_k^G yields a *G*-equivariant coarse homology theory. In Section 3.4, we define coarse Hochschild and cyclic homology by using the results of Section 3.3. We conclude the chapter with Section 3.5, where, by using the results of Section 2.2, we refine equivariant coarse Hochschild $\mathcal{X}HH_k^G$ and equivariant coarse cyclic homology $\mathcal{X}HC_k^G$ to lax symmetric monoidal functors. We will freely use the language of additive and exact categories, and of differential graded categories, as reviewed in the Appendix.

In the following, as set in our conventions, we will always assume that k is a field and tensor products \otimes are tensor products over k. However, most of the proofs of this chapter are also true for a general commutative ring. We choose to work with coefficients in a field k instead of a ring because several properties of these coarse homology theories (as discussed in Chapter 4) depend on the choice of a base field (see, *e.g.*, the construction of the natural transformation to equivariant coarse ordinary homology of Section 4.3).

3.1 The category of mixed complexes

In this section we describe the (cocomplete stable ∞ -)category of (unbounded) mixed complexes; we mainly follow Kassel's viewpoint [Kas87].

We start with Kassel's definition of mixed complexes:

Definition 3.1.1. [Kas87, §1] A mixed complex (C, b, B) is a triple consisting of a \mathbb{Z} -graded k-module $C = \{C_p\}_{p \in \mathbb{Z}}$ together with differentials b and B

$$b = (b_p \colon C_p \to C_{p-1})_{p \in \mathbb{Z}}$$
 and $B = (B_p \colon C_p \to C_{p+1})_{p \in \mathbb{Z}}$

of degree -1 and 1 respectively, satisfying the following identities:

$$b^2 = 0$$
, $B^2 = 0$, $bB + Bb = 0$.

Morphisms $f: (C, b, B) \to (C', b', B')$ of mixed complexes are given by sequences $f = (f_p: C_p \to C'_p)_{p \in \mathbb{Z}}$ of maps commuting with both the differentials b and B. The category of mixed complexes and morphisms of mixed complexes is denoted by **Mix**.

When the differentials are clear from the context, we refer to a mixed complex (C, b, B) by C.

In the category of mixed complexes there is a notion of *shifted mixed complexes* and of *cone* of a morphism of mixed complexes:

- the shifted mixed complex $(C[1], b_{C[1]}, B_{C[1]})$ of a mixed complex (C, b, B) is defined by $(C[1])_p \coloneqq C_{p-1}$, for all p, with differentials $b_{C[1]} \coloneqq -b_C$ and $B_{C[1]} \coloneqq -B_C$.
- Let $f: (C, b, B) \to (C', b', B')$ be a morphism of mixed complexes. The cone of f is defined as the mixed complex

$$\operatorname{cone}(f) \coloneqq \left(C' \oplus C[1], \begin{bmatrix} b_{C'} & f \\ 0 & -b_C \end{bmatrix}, \begin{bmatrix} B_{C'} & 0 \\ 0 & -B_C \end{bmatrix}\right)$$

where C[1] is the shifted mixed complex of C.

Recall the definitions of dg-algebras and dg-modules over a dg-algebra, Definition A.2.1. Let Λ be the dg-algebra over the field k

$$\Lambda \coloneqq \dots \to 0 \to k\epsilon \xrightarrow{0} k \to 0 \to \dots$$
(3.1.1)

generated by an indeterminate ϵ of degree 1, with $\epsilon^2 = 0$ and differential (of degree -1) $d(\epsilon) = 0$. Then, mixed complexes are nothing but dg-modules over the dg-algebra Λ :

Remark 3.1.2. [Kas87] The category **Mix** of mixed complexes is equivalent (in fact, isomorphic) to the category of left dg Λ -modules, which we denote by Λ -**Mod**, as we now explain. A mixed complex (C, b, B) yields a differential graded left Λ -module whose underlying differential graded module is (C, b) and where the multiplication $\epsilon \cdot c$ is defined by the differential B, *i.e.*, $\epsilon \cdot c \coloneqq B(c)$. Vice versa, left Λ -dg-modules (M, d) correspond to mixed complexes (C, b, B) by defining $C_p \coloneqq M_p$, $b(c) \coloneqq d(c)$ and $B(c) \coloneqq \epsilon \cdot c$ for every $c \in C_p$. The identity $B^2 = 0$ corresponds to the relation $\epsilon^2 = 0$, and by using the equality $d(\epsilon) = 0$ one gets the identity bB + Bb = 0. Observe that also morphisms of mixed complexes correspond one-to-one to morphisms of dg- Λ -modules (see Definition A.2.1 and the text thereafter) via this correspondence. We denote by $L: \mathbf{Mix} \to \Lambda$ -**Mod** the induced functor sending a mixed complex to the associated Λ -dg-module and by $R: \Lambda$ -**Mod** \to **Mix** its inverse sending a Λ -dg-module to its associated mixed complex.

A mixed complex (C, b, B) functorially determines a double chain complex $\mathcal{B}C$ [Lod98, § 2.5.10] by means of the differentials b and B:

$$\mathcal{B}C \coloneqq \left(\dots \xleftarrow{0} (C,b) \xleftarrow{B} (C[-1], b_{C[-1]}) \xleftarrow{B} \dots \xleftarrow{B} (C[-n], b_{C[-n]}) \xleftarrow{B} \dots \right); \quad (3.1.2)$$

here, the chain complex (C, b) is placed in bi-degree (0, *), *i.e.*, $\mathcal{B}C_{(0,*)} = (C_*, b)$, and the chain complex $(C[-n], b_{C[-n]})$, placed in bi-degree (n, *), is the chain complex (C, b)shifted by -n, hence $\mathcal{B}C_{(p,q)} = C_{q-p}$ for $p \ge 0$ and $\mathcal{B}C_{(p,q)} = 0$ for p < 0. The total chain complex $\text{Tot}(\mathcal{B}C)$, functorially associated to the double chain complex $\mathcal{B}C$, is the chain complex defined in degree n by $\text{Tot}_n(\mathcal{B}C) = \bigoplus_{i\ge 0} C_{n-2i}$ with differential d acting as follows:

$$d(c_n, c_{n-2}, \dots) \coloneqq (bc_n + Bc_{n-2}, \dots).$$

Let \mathbf{Ch} be the category of chain complexes over k. Consider the forgetful functor

forget:
$$\mathbf{Mix} \to \mathbf{Ch}$$
 (3.1.3)

sending a mixed complex (M, b, B) to its underlying chain complex (M, b), and the functor

$$\operatorname{Tot}(\mathcal{B}-)\colon \operatorname{Mix} \to \operatorname{Ch}$$
 (3.1.4)

just described above. We observe that the chain complex $\operatorname{Tot}(\mathcal{B}C)$ associated to the mixed complex (C, b, B) represents the derived tensor product $C \otimes_{\Lambda}^{\mathbb{L}} k$.

Hochschild homology (and all its cyclic variants: cyclic, negative and periodic) can be defined as functors from the category of mixed complexes:

Definition 3.1.3. [Kas87, Sec. 1] Let (C, b, B) be a mixed complex. The Hochschild homology $HH_*(C)$ of (C, b, B) is the homology of the underlying chain complex (C, b). Its cyclic homology $HC_*(C)$ is the homology of the associated chain complex $Tot(\mathcal{B}C)$.

Let M be a cyclic module (see Definition A.4.3) and let d_i and s_i denote the *i*-th face and *i*-th degeneracy maps of M respectively. Let t_{n+1} be the cyclic operator in degree nand let $b: M_n \to M_{n-1}$ be the alternating sum

$$b \coloneqq \sum_{i=0}^{n} (-1)^{i} d_{i} \tag{3.1.5}$$

of face maps . Let $N \coloneqq \sum_{i=0}^{n} t_{n+1}^{i}$ be the sum of the powers of the cyclic operator t_{n+1} and define the cochain map $B \colon M_n \to M_{n+1}$ as the composition

$$B \coloneqq (-1)^{n+1} (1 - t_{n+1}) s N. \tag{3.1.6}$$

Here s denotes the extra degeneracy $s = (-1)^{n+1} t_{n+1} s_n \colon M_n \to M_{n+1}$.

Remark 3.1.4. Let M be a cyclic module. Let b be the differential $b \coloneqq \sum_{i=0}^{n} (-1)^{i} d_{i}$ (3.1.5) and B the differential $B \coloneqq (-1)^{n+1}(1 - t_{n+1})sN$ (3.1.6). Then, (M, b, B) is a mixed complex. Morphisms of cyclic modules commute with face and degeneracy maps, and with the cyclic operators as well; hence yield morphisms of mixed complexes. This describes a functor from the category of cyclic modules to the category of mixed complexes.

Example 3.1.5. If A is a k-algebra (that is associative and unital), then by Example A.4.5, we get a cyclic module $Z_*(A)$. The classical Hochschild homology HH(A) of A is defined as the homology (of the underlying chain complex) of the cyclic module $Z_*(A)$. If C(A) is the mixed complex associated to the cyclic module $Z_*(A)$ as described in Remark 3.1.4, then the Hochschild homology of the mixed complex C(A) of Definition 3.1.3 coincides with the Hochschild homology HH(A). Furthermore, the cyclic homology of the mixed complex C(A) is the classical cyclic homology of the k-algebra A [Kas87, Prop. 1.3].

The category of dg-modules over k admits a combinatorial model structure (the projective model structure), whose weak equivalences are the objects-wise quasi-isomorphisms of dg-modules (see Definition A.2.10 and Remark A.2.11). In the language of mixed complexes, the definition of quasi-isomorphism of dg-modules translate as follows:

Definition 3.1.6. A morphism $(C, b, B) \to (C', b', B')$ of mixed complexes is a *quasi-isomorphism* if the underlying *b*-complexes are quasi-isomorphic via the induced chain map $(C, b) \to (C', b')$.

Remark 3.1.7. Consider the functors $L: \operatorname{Mix} \to \Lambda\operatorname{-Mod}$ sending a mixed complex to the associated Λ -dg-module and $R: \Lambda\operatorname{-Mod} \to \operatorname{Mix}$ sending a Λ -dg-module to its associated mixed complex, as defined in Remark 3.1.2. Then, quasi-isomorphisms of mixed complexes correspond by the functor L to quasi-isomorphisms of Λ -dg-modules and vice versa, *i.e.*, the functors L and R preserve quasi-isomorphisms.

Remark 3.1.8. As observed by Keller, the correspondence $M \mapsto (M, b, B)$ from cyclic modules to mixed complexes, using the differentials b (3.1.5) and B (3.1.6), is not functorial when one considers morphisms of cyclic modules that do not commute with degeneracies (as opposed to morphisms of cyclic modules that commute with both faces and degeneracies) [Kel98, Sec. 2.1]. However, Keller defines a functor from the category of precyclic modules (*i.e.*, cyclic modules without degeneracies) to the category of mixed complexes; for the sake of completeness we report also this approach [Kel98, Sec. 2.1]. Let C be a precyclic module, let d_i denote the *i*-th face map and set $b' \coloneqq \sum_{i=0}^{n-1} d_i$ and $b \coloneqq \sum_{i=0}^{n} d_i$. Let t be the cyclic operator and let N be the sum of its powers. Keller defines the mixed complex \tilde{M} as follows:

$$\tilde{M} \coloneqq \left(C \oplus C[1], \begin{bmatrix} b & (1-t) \\ 0 & -b' \end{bmatrix}, \begin{bmatrix} 0 & 0 \\ N & 0 \end{bmatrix} \right)$$

This correspondence yields a functor from precyclic modules to mixed complexes. Moreover, there is a canonical morphism (induced by [1, (1-t)s]) between this mixed complex and the mixed complex constructed in Remark 3.1.4 by using the differentials (3.1.5) and (3.1.6), and this canonical morphism of mixed complexes is a quasi-isomorphism of mixed complexes [Kel98, Sec. 2.1].

To every small dg-category \mathcal{A} , one can functorially associate a cyclic module, called the *additive cyclic nerve* CN(\mathcal{A}) as described in Definition A.4.7; hence, a mixed complex by Remark 3.1.4. The composition of these two functors describes then a functor from the category of small dg-categories to the category of mixed complexes, that we call Mix:

Definition 3.1.9. [Kel99, Def. 1.3] Let

Mix: $\operatorname{dgcat}_k \to \operatorname{Mix}$

be the functor from the category of small dg-categories over k to the category of mixed complexes defined as composition of the additive cyclic nerve of Definition A.4.7 and of the functor of Remark 3.1.4.

We now introduce the ∞ -category $\operatorname{Mix}_{\infty}$ of mixed complexes. We recall that, if **C** is an ordinary category and W denotes a collection of morphisms of **C**, then $\operatorname{N}(\mathbf{C})[W^{-1}]$ is the ∞ -category obtained by the nerve $\operatorname{N}(\mathbf{C})$ of **C** by inverting the set of morphisms W[Cis, Def. 7.1.2 & Prop. 7.1.3], [Lur14, Def. 1.3.4.1].

The ∞ -category \mathbf{Mix}_{∞} of mixed complexes is then defined as the localization of the nerve of the category \mathbf{Mix} at the class W_{mix} of quasi-isomorphisms of mixed complexes

(Definition 3.1.6):

$$\mathbf{Mix}_{\infty} \coloneqq \mathbf{N}(\mathbf{Mix})[W_{mix}^{-1}]. \tag{3.1.7}$$

Analogously, the ∞ -category Λ -Mod $_{\infty}$ is defined as the localization of the category Λ -Mod of dg- Λ -modules at the class W of quasi-isomorphisms of dg- Λ -modules (Definition A.2.10):

$$\Lambda \operatorname{-Mod}_{\infty} \coloneqq \mathrm{N}(\Lambda \operatorname{-Mod})[W^{-1}]. \tag{3.1.8}$$

Proposition 3.1.10. The ∞ -category Λ -Mod $_{\infty}$ is a cocomplete stable ∞ -category.

Proof. The category Λ -Mod is a (pre-triangulated) dg-category. By applying the dg-nerve functor N_{dg} [Lur14, Constr. 1.3.1.6] we obtain an ∞ -category N_{dg}(Λ -Mod) [Lur14, Prop. 1.3.1.10]. The dg-nerve functor sends pre-triangulated dg-categories to stable ∞ -categories [Fao17, Thm. 4.3.1], [Lur14, Prop. 1.3.1.10]. The ∞ -category N_{dg}(Λ -Mod) is a stable ∞ -category and its homotopy category can be identified (as triangulated category) with the derived category $\mathcal{D}(\Lambda)$ associated to the dg-algebra Λ (see Definition A.2.14).

The category Λ -Mod is equipped with a combinatorial simplicial model structure by Remark A.2.11. By [Lur14, Prop. 1.3.1.17] and by the fact that the simplicial nerve of the simplicial category associated to Λ -Mod is equivalent to the localization $N(\Lambda$ -Mod)[W^{-1}] (by [Lur14, Rem. 1.3.4.16 & Thm. 1.3.4.20] where we also use that the model category Λ -Mod is combinatorial, hence admits functorial factorizations), the two constructions $N(\Lambda$ -Mod)[W^{-1}] and $N_{dg}(\Lambda$ -Mod) present equivalent ∞ -categories. Hence, the ∞ -category Λ -Mod $_{\infty}$ is a stable ∞ -category.

The ∞ -category Λ -Mod $_{\infty}$ is also cocomplete by [Lur14, Prop. 1.3.4.22] because the model category Λ -Mod is combinatorial.

Proposition 3.1.11. The ∞ -category Mix $_{\infty}$ is a cocomplete stable ∞ -category.

Proof. The categories **Mix** and **A-Mod** are isomorphic by Remark 3.1.2 and the functor $L: \mathbf{Mix} \to \Lambda - \mathbf{Mod}$ and its inverse $R: \Lambda - \mathbf{Mod} \to \mathbf{Mix}$ preserve quasi-isomorphisms by Remark 3.1.7. This yields an equivalence of ∞ -categories

$$N(\mathbf{Mix})[W_{mix}^{-1}] \to N(\Lambda - \mathbf{Mod})[W^{-1}].$$

The statement is then a consequence of this equivalence and of Proposition 3.1.10.

We observe that one can prove that the ∞ -category \mathbf{Mix}_{∞} is a cocomplete stable ∞ -category by following the same arguments of [Fao17, Thm. 4.3.1]. In fact, the category **Mix** is cocomplete, has a 0-object, and the cone of a morphism of mixed complexes and the shift mixed complexes can be used for proving that \mathbf{Mix}_{∞} is also stable.

Remark 3.1.12. The homotopy category of the stable ∞ -category $\operatorname{Mix}_{\infty}$ is canonically equivalent to the derived category $\mathcal{D}(\Lambda)$ of the dg-algebra Λ (by Proposition 3.1.10 and Proposition 3.1.11).

3.2 Keller's cyclic homology of exact categories

In this section we recall Keller's construction of Hochschild and cyclic homology for exact k-linear categories [Kel99]. In fact, Keller constructs a functor

 $C \colon \mathbf{Ex}_k \to \mathbf{Mix}$

from the category \mathbf{Ex}_k of small exact k-linear categories to the category **Mix** of Kassel's mixed complexes. The functor C is also called the *cone functor* because its definition uses the mapping cone for morphisms of mixed complexes. Hochschild and cyclic homology are homological invariants of mixed complexes, as explained in Definition 3.1.3. Hence, Keller's Hochschild (or cyclic) homology of a k-linear exact category \mathcal{A} is the Hochschild (or cyclic) homology of the mixed complex $C(\mathcal{A})$ associated to \mathcal{A} by the functor C.

Among the various properties of Keller's Hochschild homology, we have the following [Kel99]:

Agreement: the Hochschild homology of the exact category of finitely generated projective modules over a (unital) algebra agrees with the Hochschild homology of the algebra.

Invariance: Hochschild homology is preserved by exact functors inducing equivalences in the bounded derived categories.

Localization: suitable sequences of exact categories (roughly, inducing short exact sequences of bounded derived categories) are sent to triangles of the triangulated category $\mathcal{D}(\Lambda)$ in a sense to be made precise.

Additivity: Hochschild homology is additive.

Trace maps: there are trace maps linking the K-theory of an exact category to its Hochschild homology.

We review and explain these properties below in Theorem 3.2.5 and Theorem 3.2.7. The relation with K-theory is discussed in Proposition 4.4.1. The most important property for us is the localization property: it allows us to deduce Mayer-Vietoris type conclusions (see Theorem 3.3.8) in the same way as it has been already done for other localizing invariants [BEKW17, BC17].

We start with the definition of the functor $C \colon \mathbf{Ex}_k \to \mathbf{Mix}$.

Let \mathcal{E} be a small k-linear exact category (see Definition A.1.5). The category $\operatorname{Ch}^{b}(\mathcal{E})$ of bounded chain complexes in \mathcal{E} and the sub-category $\operatorname{Acy}^{b}(\mathcal{E})$ of bounded acyclic chain complexes in \mathcal{E} have the structure of dg-categories, as described in Example A.2.6. We denote these dg-categories by $\operatorname{Ch}^{b}_{dg}(\mathcal{E})$ and $\operatorname{Acy}^{b}_{dg}(\mathcal{E})$ respectively.

For a small dg-category \mathcal{A} , let $Mix(\mathcal{A})$ be the mixed complex associated to \mathcal{A} as described in Definition 3.1.9.

Definition 3.2.1. [Kel99, Sec. 1.4] Let \mathcal{E} be an exact k-linear category. The mixed complex $C(\mathcal{E})$ associated to \mathcal{E} is the cone

$$C(\mathcal{E}) \coloneqq \operatorname{cone}(\operatorname{Mix}(\operatorname{Acy}_{\operatorname{dg}}^{b}(\mathcal{E}))) \to \operatorname{Mix}(\operatorname{Ch}_{\operatorname{dg}}^{b}(\mathcal{E})))$$
(3.2.1)

of the morphism of mixed complexes $\operatorname{Mix}(\operatorname{Acy}_{\operatorname{dg}}^b(\mathcal{E})) \to \operatorname{Mix}(\operatorname{Ch}_{\operatorname{dg}}^b(\mathcal{E}))$ induced by the inclusion $\operatorname{Acy}_{\operatorname{dg}}^b(\mathcal{E}) \to \operatorname{Ch}_{\operatorname{dg}}^b(\mathcal{E})$ of dg-categories.

Remark 3.2.2. [Kel99, Sec. 1.4] The cone construction C of Definition 3.2.1 is functorial with respect to (k-linear) exact functors between (k-linear) exact categories.

The cone construction $C: \mathbf{Ex}_k \to \mathbf{Mix}$ is a functor from the category \mathbf{Ex}_k of small exact k-linear categories to the category \mathbf{Mix} of mixed complexes. We can describe the functor as follows.

For a category \mathbf{C} , we denote by \mathbf{C}^{Δ^1} the arrow category $\mathbf{Fun}(\{0 \to 1\}, \mathbf{C})$. Then, the cone functor of Definition 3.2.1 can be written as the composition:

$$C: \mathbf{Ex}_k \longrightarrow \mathbf{dgcat}_k^{\Delta^1} \xrightarrow{\mathrm{Mix}} \mathbf{Mix}^{\Delta^1} \xrightarrow{\mathrm{cone}} \mathbf{Mix}$$
(3.2.2)

where the first functor associates to a k-linear exact category \mathcal{E} the morphism of dgcategories $\operatorname{Acy}_{\operatorname{dg}}^{b}(\mathcal{E}) \to \operatorname{Ch}_{\operatorname{dg}}^{b}(\mathcal{E})$ in $\operatorname{dgcat}^{\Delta^{1}}$ (and on morphisms it is defined in the canonical way), the second functor uses Definition 3.1.9 and the third functor is the cone in the category **Mix**.

Definition 3.2.3. Let \mathcal{E} be a small k-linear exact category. Keller's Hochschild (cyclic) homology of \mathcal{E} is defined as the Hochschild (cyclic) homology of the mixed complex $C(\mathcal{E})$ associated to \mathcal{E} by the functor C of Definition 3.1.3.

Before describing the properties of the functor $C \colon \mathbf{Ex}_k \to \mathbf{Mix}$, we need some more terminology:

Definition 3.2.4. [Kel99, Sec. 1.5] Let \mathcal{A}, \mathcal{B} be additive categories and let $\mathcal{T}, \mathcal{T}'$ and \mathcal{T}'' be triangulated categories.

- (i) A factor-dense subcategory \mathcal{A}' of \mathcal{A} is a full subcategory such that each object of \mathcal{A} is a direct factor of a finite direct sum of objects of \mathcal{A}' .
- (ii) An equivalence up to factors is an additive functor $\mathcal{A} \to \mathcal{B}$ that induces an equivalence onto a factor-dense subcategory of \mathcal{B} .
- (iii) A sequence

 $\mathcal{T}' \to \mathcal{T} \to \mathcal{T}''$

of triangulated categories is *exact up to factors* if the composition is zero, the functor $\mathcal{T}' \to \mathcal{T}$ is fully faithful and the induced functor $\mathcal{T}/\mathcal{T}' \to \mathcal{T}''$ from the Verdier quotient to \mathcal{T}'' is an equivalence up to factors.
Let \mathcal{E} be a small exact k-linear category and let $\mathcal{D}^b(\mathcal{E})$ be the derived category of \mathcal{E} (see Definition A.1.10). We recall that, by Example 3.1.5, one can associate a mixed complex C(A) to every k-algebra A. Recall also that $\mathcal{D}(\Lambda)$ denotes the derived category of the dg-algebra Λ .

Theorem 3.2.5. [Kel99, Theorem 1.5] Let k be a commutative ring.

1. If A is a k-algebra, there is a natural isomorphism

$$C(A) \to C(\operatorname{proj} A)$$

in $\mathcal{D}(\Lambda)$, where projA is the exact category of finitely generated projective modules.

2. If $F: \mathcal{A} \to \mathcal{B}$ is an exact functor between exact categories that induces an equivalence up to factors $\mathcal{D}^{b}(\mathcal{A}) \to \mathcal{D}^{b}(\mathcal{B})$, then F induces an isomorphism

$$C(\mathcal{A}) \to C(\mathcal{B})$$

in $\mathcal{D}(\Lambda)$.

3. If $F: \mathcal{A}' \to \mathcal{A}$ and $G: \mathcal{A} \to \mathcal{A}''$ are exact functors between exact categories such that the sequence

$$\mathcal{D}^b(\mathcal{A}') \to \mathcal{D}^b(\mathcal{A}) \to \mathcal{D}^b(\mathcal{A}'')$$

is exact up to factors, then there is a canonical morphism $\partial(F,G)$ such that the sequence

$$C(\mathcal{A}') \longrightarrow C(\mathcal{A}) \longrightarrow C(\mathcal{A}'') \xrightarrow{\partial(F,G)} C(\mathcal{A}')[1]$$

is a triangle in $\mathcal{D}(\Lambda)$.

Remark 3.2.6. After application of the functor C, the inclusion of an exact k-linear category in its idempotent completion induces an isomorphism in $\mathcal{D}(\Lambda)$.

If \mathcal{E} is a small k-linear exact category, let $\operatorname{Ex} \mathcal{E}$ denote the category of admissible short exact sequences of \mathcal{E} (which is again exact, provided that short exact sequences are defined component-wise). Consider the following exact functors:

 $I: \mathcal{E} \to \operatorname{Ex} \mathcal{E} \quad A \mapsto (A \xrightarrow{1} A \to 0)$ $R: \operatorname{Ex} \mathcal{E} \to \mathcal{E} \quad (A \to B \to C) \mapsto A$ $P: \operatorname{Ex} \mathcal{E} \to \mathcal{E} \quad (A \to B \to C) \mapsto C$ $S: \mathcal{E} \to \operatorname{Ex} \mathcal{E} \quad C \mapsto (0 \to C \xrightarrow{1} C)$

The functors I and P are left adjoint to the functors R and S, respectively. Keller proves that Hochschild and cyclic homology are additive invariants:

Theorem 3.2.7 (Theorem 1.12 [Kel99]). Let \mathcal{E} be a small k-linear exact category. Then, the functors P and R induce an isomorphism

$$C(\operatorname{Ex} \mathcal{E}) \xrightarrow{\sim} C(\mathcal{E}) \oplus C(\mathcal{E})$$

in $\mathcal{D}(\Lambda)$.

We now interpret Keller's results in the context of ∞ -categories.

Let \mathbf{dgcat}_k be the category of small dg-categories over k. The category \mathbf{dgcat}_k is endowed with the Morita model structure of Theorem A.2.16; we denote by $\mathbf{dgcat}_{k,\infty}$ the underlying ∞ -category:

$$\mathbf{dgcat}_{k,\infty} \coloneqq \mathrm{N}(\mathbf{dgcat}_k)[W_{\mathrm{Morita}}^{-1}] \tag{3.2.3}$$

where W_{Morita} is the class of Morita equivalences between dg-categories.

To every dg-category we can associate a cyclic module, *i.e.*, its additive cyclic nerve of Definition A.4.7, hence a mixed complex by Remark 3.1.4. Let

Mix: $\mathbf{dgcat}_k \to \mathbf{Mix}$

be the functor of Definition 3.1.9. By Theorem 3.2.5 (2), the functor Mix sends Morita equivalences of dg-categories to quasi-isomorphisms of mixed complexes. A morphism in an ∞ -category is an equivalence if its image in the homotopy category is an isomorphism, hence, the functor Mix descends to a functor

$$\begin{array}{ccc} \mathbf{dgcat}_k & \xrightarrow{\mathrm{Mix}} & \mathbf{Mix} \\ & & & \downarrow_{\mathrm{loc}} & & \downarrow_{\mathrm{loc}} \\ \mathbf{dgcat}_{k,\infty} & \xrightarrow{\mathrm{Mix}_{\infty}} & \mathbf{Mix}_{\infty} \end{array}$$
(3.2.4)

on the localizations. Observe that the functor Mix preserves filtered colimits and that quasiisomorphisms of mixed complexes commute with filtered colimits of mixed complexes. Hence, $loc \circ Mix$ preserves filtered colimits because the localization preserves filtered colimits as well.

Let \mathbf{Ex}_k be the category of small exact k-linear categories. Let W_{ex} be the class of exact functors between exact k-linear categories inducing equivalences up to factors between the associated bounded derived categories (Definition 3.2.4). Let $N(\mathbf{Ex}_k)[W_{ex}^{-1}]$ be the ∞ -category associated to \mathbf{Ex}_k by inverting the morphisms of W_{ex} [Lur14, Def. 1.3.4.1]. By Theorem 3.2.5 (2), the cone functor $C: \mathbf{Ex}_k \to \mathbf{Mix}$ (3.2.2) sends morphisms in W_{ex} to quasi-isomorphisms of mixed complexes, hence it descends to a functor between the associated ∞ -categories



The functor C commutes with filtered colimits (see also Proposition 3.3.7).

The ∞ -category $\operatorname{Mix}_{\infty}$ is stable by Proposition 3.1.11. Cofiber sequences of $\operatorname{Mix}_{\infty}$ [Lur14, Def. 1.1.1.6] are detected in its homotopy category, *i.e.*, in $\mathcal{D}(\Lambda)$. By Theorem 3.2.5 (3), the composition loc $\circ C$ (and, analogously, composition with the functor Mix) sends sequences of exact categories inducing equivalences up to factors (between the associated bounded derived categories) to cofiber sequences of the stable ∞ -category $\operatorname{Mix}_{\infty}$.

3.3 The equivariant coarse homology theory $\widetilde{\mathcal{X}C}_k^G$

The goal of this section is to construct an equivariant Mix_{∞} -valued coarse homology theory (see Definition 1.2.1)

$$\widetilde{\mathcal{XC}}_k^{\mathrm{G}} \colon G\mathbf{BornCoarse} o \mathbf{Mix}_\infty$$

from the category of G-bornological coarse spaces to the cocomplete stable ∞ -category of mixed complexes (Proposition 3.1.11) that uses Keller's functor $C: \mathbf{Ex}_k \to \mathbf{Mix}$ of Definition 3.2.1. In fact, the functor $\widetilde{\mathcal{XC}}_k^G$ is defined (see Definition 3.3.1) as Keller's cone C of the k-linear category $V_k^G(X)$ of G-equivariant X-controlled finite dimensional k-vector spaces (Definition 2.1.2) equipped with the exact structure given by the short split exact sequences.

The main result of the section is Theorem 3.4.2, where we prove that the functor $\widetilde{\mathcal{XC}}_k^G : GBornCoarse \to Mix_{\infty}$ is a *G*-equivariant coarse homology theory; this is achieved by showing that the functor $\widetilde{\mathcal{XC}}_k^G$ satisfies:

- (i) coarse invariance, see Proposition 3.3.4;
- (ii) vanishing on flasque spaces, see Proposition 3.3.6;
- (iii) u-continuity, see Proposition 3.3.7;
- (iv) coarse excision, see Theorem 3.3.8;
- *i.e.*, the axioms describing an equivariant coarse homology theory (see Definition 1.2.1).

We start with the definition of the functor $\widetilde{\mathcal{XC}}_k^G : GBornCoarse \to Mix_{\infty}$.

By Remark 2.1.11, the category $V_k^G(X)$ of *G*-equivariant *X*-controlled finite dimensional *k*-vector spaces is a *k*-linear category. It is also exact with the exact structure given by the short split exact sequences by Example A.1.6. Hence, the functor V_k^G (2.1.3) can be seen as a functor

 $V_k^G : G\mathbf{BornCoarse} \to \mathbf{Ex}_k$

from the category of G-bornological coarse spaces to the category of small exact k-linear categories. By abuse of notation, with denote the functor V_k^G from **GBornCoarse** to **Add** or to **Ex**_k with the same symbol.

Let loc be the localization functor loc: $\mathbf{Mix} \to \mathrm{N}(\mathbf{Mix}) \to \mathrm{N}(\mathbf{Mix}[W_{mix}^{-1}]) = \mathbf{Mix}_{\infty}$ [Cis, Def. 7.1.2], [Lur14, Def. 1.3.4.1]. Let $C: \mathbf{Ex}_k \to \mathbf{Mix}$ denote Keller's cone construction of Definition 3.2.1.

Definition 3.3.1. We denote by $\widetilde{\mathcal{XC}}_k^G$ the following composition of functors

$$\widetilde{\mathcal{XC}}_k^G\colon G\mathbf{BornCoarse} \xrightarrow{V_k^G} \mathbf{Ex}_k \xrightarrow{C} \mathbf{Mix} \xrightarrow{\mathrm{loc}} \mathbf{Mix}_\infty$$

from the category of bornological coarse spaces with G-action (Definition 1.1.21) to the ∞ -category of mixed complexes (Definition 3.1.1 and (3.1.7)).

Theorem 3.3.2. The functor

$$\widetilde{\mathcal{XC}}_k^G : G\mathbf{BornCoarse} \longrightarrow \mathbf{Mix}_\infty$$

is a G-equivariant Mix_{∞} -valued coarse homology theory.

The proof of Theorem 3.3.2 follows the ideas of [BEKW17, BC17] and it is a combination of the following Proposition 3.3.4, Proposition 3.3.6, Proposition 3.3.7 and Theorem 3.3.8.

Let split: $\operatorname{Cat}_k \to \operatorname{Ex}_k$ denote the functor that sends a k-linear category to the same category equipped with the split exact structure.

Remark 3.3.3. We have decided to provide a complete proof for the composition

$$\widetilde{\mathcal{XC}}_k^G \colon G\mathbf{BornCoarse} \xrightarrow{V_k^G} \mathbf{Cat}_k \xrightarrow{\mathrm{split}} \mathbf{Ex}_k \xrightarrow{C} \mathbf{Mix} \xrightarrow{\mathrm{loc}} \mathbf{Mix}_{\infty},$$

(that uses the functor $C: \mathbf{Ex}_k \to \mathbf{Mix}$ of Definition 3.2.1 instead of the functor Mix of Definition 3.4.1), because the proof of Theorem 3.3.2 works in the more general context in which the k-linear category $V_k^G(X)$ is equipped with other (non-split) exact structures.

We proceed with the proof of the axioms of Definition 1.2.1.

Proposition 3.3.4. The functor $\widetilde{\mathcal{XC}}_k^G$: **GBornCoarse** \to **Mix**_{∞} (3.3.1) satisfies coarse invariance.

Proof. If $f: X \to Y$ is a coarse equivalence of *G*-bornological coarse spaces, then it induces a natural equivalence $f_*: V_k^G(X) \to V_k^G(Y)$ by Lemma 2.1.14. Keller's cone functor 3.2.1 sends equivalences of *k*-linear exact categories to equivalences of mixed complexes by Theorem 3.2.5 (2). Hence, the functor f_* induces the equivalence

$$\widetilde{\mathcal{X}C}_k^G(X) \xrightarrow{\sim} \widetilde{\mathcal{X}C}_k^G(Y)$$

in \mathbf{Mix}_{∞} . The functor $\widetilde{\mathcal{XC}}_k^G$ satisfies then coarse invariance by Remark 1.2.2.

We recall that the homotopy category of Mix_{∞} is the derived category $\mathcal{D}(\Lambda)$. The following is [Sch11, Theorem 2.3.11], reinterpreted in our setting:

Theorem 3.3.5. Let \mathcal{E} and \mathcal{E}' be exact categories and let $F' \to F \to F''$ be a sequence of exact functors $\mathcal{E} \to \mathcal{E}'$ such that $F'(A) \to F(A) \to F''(A)$ is a short exact sequence for all objects $A \in \mathcal{E}$. Then, the functors F and $F' \oplus F''$ induce an equivalence

$$C(F) \simeq C(F' \oplus F'') \colon C(\mathcal{E}) \to C(\mathcal{E}')$$

in Mix_{∞} .

Proof. Recall the exact functors I, P, R, S defined in Theorem 3.2.7; the induced functors D^bI and D^bP are left adjoint to D^bR and D^bS , with unit and counit the isomorphisms $\mathrm{id} \to D^bR \circ D^bI$ and $D^bP \circ D^bS \to \mathrm{id}$.

The sequence of functors $F' \to F \to F''$ induces an exact functor $F_{\bullet} \colon \mathcal{E} \to \operatorname{Ex} \mathcal{E}'$ sending an object A to the sequence $F'(A) \to F(A) \to F''(A)$. By Theorem 3.2.7, the composition of the functors $I \oplus S$ and (R, P)

$$G\colon \operatorname{Ex} \mathcal{E}' \xrightarrow{(R,P)} \mathcal{E}' \times \mathcal{E}' \xrightarrow{I \oplus S} \operatorname{Ex} \mathcal{E}'$$

being mutual inverses in $\mathcal{D}(\Lambda)$, induces a map $C(G): C(\operatorname{Ex} \mathcal{E}') \to C(\operatorname{Ex} \mathcal{E}')$ that is isomorphic to the identity in $\mathcal{D}(\Lambda)$.

Let $M: \operatorname{Ex} \mathcal{E}' \to \mathcal{E}'$ be the functor sending a short exact sequence $A \to B \to C$ to the object B. Then, the functors F and $F' \oplus F''$ can be written as the following compositions

$$\mathcal{E} \xrightarrow{F_{\bullet}} \operatorname{Ex} \mathcal{E}' \xrightarrow{M} \mathcal{E}' \text{ and } \mathcal{E} \xrightarrow{F_{\bullet}} \operatorname{Ex} \mathcal{E}' \xrightarrow{G} \operatorname{Ex} \mathcal{E}' \xrightarrow{M} \mathcal{E}'$$

and they induce isomorphic maps $C(\mathcal{E}) \to C(\mathcal{E}')$ in $\mathcal{D}(\Lambda)$.

Recall the definition of a flasque category Definition A.1.4. As a corollary of Schlichting's theorem, we get that the functor $\widetilde{\mathcal{XC}}_{k}^{G}$ vanishes on flasque spaces:

Proposition 3.3.6. The functor $\widetilde{\mathcal{XC}}_k^G$: **GBornCoarse** \to **Mix**_{∞} (3.3.1) vanishes on flasque spaces.

Proof. Let X be a flasque space. By Lemma 2.1.15, the category $V_k^G(X)$ is a flasque category, hence there exists an endofunctor $S: V_k^G(X) \to V_k^G(X)$ such that $\operatorname{id}_{V_k^G(X)} \oplus S \cong S$. By Theorem 3.3.5, $C(\operatorname{id}) \oplus C(S)$ and $C(\operatorname{id} \oplus S) \cong C(S)$ induce equivalent morphisms in $\operatorname{\mathbf{Mix}}_{\infty}$. This means that the morphism

$$\widetilde{\mathcal{XC}}_{k}^{G}(\mathrm{id}) \colon \widetilde{\mathcal{XC}}_{k}^{G}(X) \to \widetilde{\mathcal{XC}}_{k}^{G}(X)$$

is equivalent to the 0-morphism, hence that $\widetilde{\mathcal{XC}}_k^G(X)$ is equivalent to the object 0 in $\operatorname{Mix}_{\infty}$, concluding the proof.

Proposition 3.3.7. The functor $\widetilde{\mathcal{XC}}_k^G$: **GBornCoarse** \to **Mix**_{∞} (3.3.1) is u-continuous. *Proof.* Let X be a G-bornological coarse space, and let \mathcal{C}^G be the poset of G-invariant controlled sets. By Remark 2.1.13, there is an equivalence $V_k^G(X) \simeq \operatorname{colim}_{U \in \mathcal{C}^G} V_k^G(X_U)$ of k-linear exact categories.

The canonical maps

$$\operatorname{colim}_U \operatorname{Ch}^b_{\operatorname{dg}}(V^G_k(X_U)) \to \operatorname{Ch}^b_{\operatorname{dg}}(V^G_k(X))$$

and

$$\operatorname{colim}_U \operatorname{Acy}_{\operatorname{dg}}^b(V_k^G(X_U)) \to \operatorname{Acy}_{\operatorname{dg}}^b(V_k^G(X))$$

are equivalences of k-linear exact categories. The functor C sends equivalences to equivalences of mixed complexes, hence we get the equivalence

$$\widetilde{\mathcal{X}C}_{k}^{G}(X) \simeq \operatorname{colim}_{U \in \mathcal{C}^{G}} \widetilde{\mathcal{X}C}_{k}^{G}(X_{U})$$

in \mathbf{Mix}_{∞} . This proves that the functor $\widetilde{\mathcal{XC}}_{k}^{G}$ is u-continuous.

Theorem 3.3.8. The functor $\widetilde{\mathcal{XC}}_k^G$: **GBornCoarse** \to **Mix**_{∞} (3.3.1) satisfies coarse excision.

The proof of the theorem follows the main ideas of [CP97, BE16, BEKW17]; we first need some more terminology.

Definition 3.3.9. [Kas15] A full additive subcategory \mathcal{A} of an additive category \mathcal{U} is a *Karoubi-filtration* if every diagram

$$X \to Y \to Z$$

in \mathcal{U} , with $X, Z \in \mathcal{A}$, admits an extension

$$\begin{array}{ccc} X & \longrightarrow & Y & \longrightarrow & Z \\ \downarrow & & \downarrow \cong & \uparrow \\ U & \stackrel{i}{\longrightarrow} & U \oplus U^{\perp} & \stackrel{p}{\longrightarrow} & U \end{array}$$

with $U \in \mathcal{A}$.

By [Kas15, Lemma 5.6], this definition is equivalent to the classical one [Kar70, CP97]. If \mathcal{A} is a Karoubi-filtration of \mathcal{U} , we can construct a quotient category \mathcal{U}/\mathcal{A} . Its objects are the objects of \mathcal{U} , and the morphisms sets are defined as follows:

$$\operatorname{Hom}_{\mathcal{U}/\mathcal{A}}(U,V) \coloneqq \operatorname{Hom}_{\mathcal{U}}(U,V)/\sim$$

where the relation identifies pairs of maps $U \to V$ whose difference factorizes through an object of \mathcal{A} .

Let X be a G-bornological coarse space and let $\mathcal{Y} = (Y_i)_{i \in I}$ be an equivariant big family on X (see Definition 1.1.26). The bornological coarse space Y_i is a subspace of X with the induced bornology and coarse structure. The inclusion $Y_i \hookrightarrow X$ induces a functor $V_k^G(Y_i) \to V_k^G(X)$ which is injective on objects. The categories $V_k^G(Y_i)$ and $V_k^G(\mathcal{Y}) \coloneqq \operatorname{colim}_{i \in I} V_k^G(Y_i)$ are full subcategories of $V_k^G(X)$.

Lemma 3.3.10. [BEKW17, Lemma 8.14] Let \mathcal{Y} be an equivariant big family on the *G*-bornological coarse space X. Then, the full additive subcategory $V_k^G(\mathcal{Y})$ of $V_k^G(X)$ is a Karoubi filtration.

Let X be a G-bornological coarse space, and (Z, \mathcal{Y}) be an equivariant complementary pair. Consider the functor

$$a: V_k^G(Z)/V_k^G(Z \cap \mathcal{Y}) \to V_k^G(X)/V_k^G(\mathcal{Y})$$
(3.3.1)

induced by the inclusion of Z in X; on objects, it coincides with $i_* \colon V_k^G(Z) \to V_k^G(X)$, but on morphisms it sends an equivalence class [A] of A in the equivalence $[i_*(A)]$ of $i_*(A)$.

Lemma 3.3.11. [BEKW17, Prop. 8.15] The functor a (3.3.1) is an equivalence of categories.

Remark 3.3.12. By [Sch04, Ex. 1.8, Prop. 2.6], if $\mathcal{A} \subseteq \mathcal{U}$ is a Karoubi filtration with \mathcal{A} idempotent complete, the sequence of exact categories $\mathcal{A} \to \mathcal{U} \to \mathcal{U}/\mathcal{A}$ induces a sequence of triangulated categories

$$D^b(\mathcal{A}) \to D^b(\mathcal{U}) \to D^b(\mathcal{U}/\mathcal{A})$$

that is exact up to factors. When \mathcal{A} is not idempotent complete, this is achieved by first applying [Sch00, Lemma 3.8] to the Karoubi filtration $\mathcal{A} \subseteq U$ and then again [Sch04, Prop. 2.6].

We use this remark and the previous lemmas to prove Theorem 3.3.8:

Proof of Theorem 3.3.8. Let X be a G-bornological coarse space, and let (Z, \mathcal{Y}) be an equivariant complementary pair on X. By Lemma 3.3.10, $V_k^G(Z \cap \mathcal{Y}) \subseteq V_k^G(Z)$ and $V_k^G(\mathcal{Y}) \subseteq V_k^G(X)$ are Karoubi filtrations. Consider the following sequences of k-linear exact categories:

$$V_k^G(Z \cap \mathcal{Y}) \to V_k^G(Z) \to V_k^G(Z)/V_k^G(Z \cap \mathcal{Y})$$

and

$$V_k^G(\mathcal{Y}) \to V_k^G(X) \to V_k^G(X)/V_k^G(X \cap \mathcal{Y}).$$

By Remark 3.3.12, the induced sequences of bounded derived categories are exact up to factors, hence the assumptions of Theorem 3.2.5 are satisfied and we get cofiber sequences of mixed complexes:

$$C(V_k^G(Z \cap \mathcal{Y})) \to C(V_k^G(Z)) \to C(V_k^G(Z)/V_k^G(Z \cap \mathcal{Y})) \to C(V_k^G(Z \cap \mathcal{Y}))[1]$$

and

$$C(V_k^G(\mathcal{Y})) \to C(V_k^G(X)) \to C(V_k^G(X)/V_k^G(X \cap \mathcal{Y})) \to C(V_k^G(\mathcal{Y}))[1]$$

The inclusion $Z \hookrightarrow X$ induces a commutative diagram

$$\begin{array}{ccc} C(V_k^G(Z \cap \mathcal{Y})) & \longrightarrow C(V_k^G(Z)) & \longrightarrow C(V_k^G(Z)/V_k^G(Z \cap \mathcal{Y})) \\ & & \downarrow & & \downarrow^{a_*} \\ C(V_k^G(\mathcal{Y})) & \longrightarrow C(V_k^G(X)) & \longrightarrow C(V_k^G(X)/V_k^G(X \cap \mathcal{Y})) \end{array}$$

where a_* is the map induced by $a: V_k^G(Z)/V_k^G(Z \cap \mathcal{Y}) \to V_k^G(X)/V_k^G(\mathcal{Y})$ (3.3.1). By Lemma 3.3.11, the functor a_* yields an equivalence of categories, hence the left square is a co-Cartesian square in \mathbf{Mix}_{∞} .

In order to conclude the proof, we recall that $\mathcal{X}C_k^G(\mathcal{Y})$ is the filtered colimit $\mathcal{X}C_k^G(\mathcal{Y}) = \operatorname{colim}_i \mathcal{X}C_k^G(Y_i)$ (1.2.1) and that

$$V_k^G(\mathcal{Y}) \coloneqq \operatorname{colim}_{i \in I} V_k^G(Y_i).$$

The functor C commutes with filtered colimits of k-linear categories, hence we have the equivalence $C(V_k^G(\mathcal{Y})) = C(\operatorname{colim}_i V_k^G(Y_i)) \simeq \operatorname{colim}_i C(V_k^G(Y_i))$ and the same holds for $Z \cap \mathcal{Y}$. By using these identifications, we obtain the co-Cartesian square in $\operatorname{Mix}_{\infty}$

$$\begin{array}{ccc} \widetilde{\mathcal{X}C}^G_k(Z\cap\mathcal{Y}) & \longrightarrow & \widetilde{\mathcal{X}C}^G_k(Z) \\ & & & \downarrow \\ & & & \downarrow \\ \widetilde{\mathcal{X}C}^G_k(\mathcal{Y}) & \longrightarrow & \widetilde{\mathcal{X}C}^G_k(X) \end{array}$$

meaning that $\widetilde{\mathcal{X}C}_k^G$ satisfies coarse excision.

Proof of Theorem 3.3.2. By Proposition 3.1.11, the ∞ -category $\operatorname{Mix}_{\infty}$ is stable and cocomplete. By Proposition 3.3.4, Proposition 3.3.6, Proposition 3.3.7 and Theorem 3.3.8, the functor $\widetilde{\mathcal{XC}}_{k}^{G}$ of Definition 3.3.1 is a *G*-equivariant coarse homology theory.

3.4 Coarse Hochschild and coarse cyclic homology

In this section, we define Hochschild homology

 $\mathcal{X}\mathrm{HH}_k^G \colon G\mathbf{BornCoarse} \to \mathbf{Ch}_\infty$

and cyclic homology

 $\mathcal{X}\mathrm{HC}^G_k\colon G\mathbf{BornCoarse} \to \mathbf{Ch}_\infty$

for G-bornological coarse spaces (see Definition 3.4.6). These are equivariant coarse homology theories with values in the cocomplete stable ∞ -category \mathbf{Ch}_{∞} of chain complexes.

By Theorem 3.3.2, Keller's cone functor $C: \mathbf{Ex}_k \to \mathbf{Mix}$ yields an equivariant coarse homology theory $\widetilde{\mathcal{XC}}_k^G: G\mathbf{BornCoarse} \to \mathbf{Mix}_\infty$ with values in the cocomplete stable ∞ -category \mathbf{Mix}_∞ of mixed complexes. In this section, namely in Lemma 3.4.4, we see that the functor C and the functor $\mathbf{Mix}: \mathbf{dgcat}_k \to \mathbf{Mix}$ of Definition 3.1.9 are naturally equivalent when restricted to k-linear categories (equipped with the exact structure given by split short exact sequences). Hence, also the functor \mathbf{Mix} yields a G-equivariant coarse homology theory that we call $\mathcal{X}\mathbf{Mix}_k^G$ (see Definition 3.4.1). Then, we define the functors $\mathcal{X}\mathbf{HH}_k^G$ and $\mathcal{X}\mathbf{HC}_k^G$ as the Hochschild and cyclic homology,

Then, we define the functors $\mathcal{X}HH_k^G$ and $\mathcal{X}HC_k^G$ as the Hochschild and cyclic homology, respectively, of the mixed complex $Mix(V_k^G(X))$ associated to the k-linear category $V_k^G(X)$ of G-equivariant X-controlled finite dimensional k-vector spaces (see Definition 2.1.2).

We start with the definition of the functor $\mathcal{X}\operatorname{Mix}_k^G : G\operatorname{BornCoarse} \to \operatorname{Mix}_{\infty}$.

Let Cat_k be the category of small k-linear categories, let $V_k^G : G\operatorname{BornCoarse} \to \operatorname{Cat}_k$ be the functor of Remark 2.1.12, let Mix: $\operatorname{dgcat}_k \to \operatorname{Mix}$ be the functor of Definition 3.1.9, let $\iota : \operatorname{Cat}_k \to \operatorname{dgcat}_k$ be the functor that associates to a k-linear category the dg-category with the same objects and morphisms complexes concentrated in degree zero (see Example A.2.5) and let loc be the localization functor loc: $\operatorname{Mix} \to \operatorname{N}(\operatorname{Mix}) \to \operatorname{N}(\operatorname{Mix}[W_{mix}^{-1}]) =$ $\operatorname{Mix}_{\infty}$.

Definition 3.4.1. Let k be a field. We denote by $\mathcal{X}Mix_k^G$ the following composition of functors

$$\mathcal{X}\mathrm{Mix}_k^G \colon G\mathbf{BornCoarse} \xrightarrow{V_k^G} \mathbf{Cat}_k \xrightarrow{\iota} \mathbf{dgcat}_k \xrightarrow{\mathrm{Mix}} \mathbf{Mix} \xrightarrow{loc} \mathbf{Mix}_\infty$$

from the category of G-bornological coarse spaces (Definition 1.1.21) to the ∞ -category of mixed complexes (Definition 3.1.1).

Theorem 3.4.2. The functor

$$\mathcal{X}\mathrm{Mix}_k^G \colon G\mathbf{BornCoarse} \longrightarrow \mathbf{Mix}_{\infty}$$

is a G-equivariant Mix_{∞} -valued coarse homology theory.

Proof. The theorem is a consequence of Theorem 3.3.2 and the next Lemma 3.4.4. \Box

Let \mathcal{A} be a small additive category, equipped with the exact structure given by the split short exact sequences (Example A.1.6). Consider the category of bounded chain complexes in \mathcal{A} and recall that a bounded chain complex is acyclic in degree n if the boundary operator factors through an object Z_{n+1}



with p_{n+1} a deflation and i_{n+1} an inflation (see Definition A.1.8).

For an additive category \mathcal{A} , let $\operatorname{Ch}_{\operatorname{dg}}^{b}(\mathcal{A})$ be the associated dg-category of bounded chain complexes and $\operatorname{Acy}_{\operatorname{dg}}^{b}(\mathcal{A})$ be the sub-dg-category of bounded acyclic chain complexes; let Mix: $\operatorname{dgcat}_{k} \to \operatorname{Mix}$ be the functor of Definition 3.1.9.

Lemma 3.4.3. Let \mathcal{A} be an additive category. Then, the mixed complex $\operatorname{Mix}(\operatorname{Acy}_{dg}^{b}(\mathcal{A}))$ associated to the dg-category of bounded acyclic chain complexes in \mathcal{A} is quasi-isomorphic to the mixed complex 0.

Proof. The exact structure on \mathcal{A} is given by the short split exact sequences. Let $K = (K_p)_{p \in \mathbb{Z}}$ be a bounded acyclic chain complex in $\mathrm{Ch}^b(\mathcal{A})$. Let H be another bounded chain complex in $\mathrm{Ch}^b(\mathcal{A})$ and consider the chain complex $\mathrm{Hom}_{\mathrm{Ch}^b_{\mathrm{dg}}(\mathcal{A})}(H, K)$. First, we want to see that this chain complex is acyclic.

Let g be a morphism in $\operatorname{Hom}_{\operatorname{Ch}^{b}_{\operatorname{dg}}(\mathcal{A})}(H, K)$; without loss of generality, we can assume that the degree of g is zero, hence $g = (g_p \colon H_p \to K_p)_{p \in \mathbb{Z}}$. We want to construct an element $f = (f_p)_{p \in \mathbb{Z}}$ of $\operatorname{Hom}_{\operatorname{Ch}^{b}_{\operatorname{dg}}(\mathcal{A})}(H, K)$ of degree 1 whose differential is g. In order to construct $f_n \colon H_n \to K_{n+1}$ we use the split exact structure of \mathcal{A} . In fact, by acyclicity, we have the diagram:



where the *i*'s are inflations and the *p*'s deflations of \mathcal{A} . The split exact structure of \mathcal{A} provides splittings of both the *i*'s and *p*'s, say $j_{n+1} \colon K_n \to Z_{n+1}$ and $q_n \colon Z_n \to K_n$. Hence, a lift of $g_n \colon H_n \to K_n$ is constructed as $f_n \coloneqq q_{n+1} \circ j_{n+1} \circ g_n$, and this for every $n \in \mathbb{Z}$. This shows the existence of a map $f = (f_p)_{p \in \mathbb{Z}}$ of degree 1. By using again the splitting at K_n , we see that the differential $d(f) = d_K \circ f - f \circ d_H$ is the starting map g. This is enough to prove that the chain complex $\operatorname{Hom}_{\operatorname{Ch}^b_{\operatorname{dg}}(\mathcal{A})}(H, K)$ is acyclic.

The additive cyclic nerve associated to the dg-category $\operatorname{Acy}_{dg}^{b}(\mathcal{A})$ is given in degree n by direct sums of complexes

$$\operatorname{Hom}(K_1, K_0) \otimes_k \operatorname{Hom}(K_2, K_1) \otimes_k \cdots \otimes_k \operatorname{Hom}(K_0, K_n)$$

where every bounded chain complex K_i over \mathcal{A} is acyclic, hence the chain complexes $\operatorname{Hom}(K_i, K_{i-1})$ are acyclic. The tensor product of acyclic complexes is again acyclic, hence $\operatorname{CN}_n(\operatorname{Acy}_{\operatorname{dg}}^b(\mathcal{A}))$ is acyclic for each n. The whole complex $\operatorname{CN}_*(\operatorname{Acy}_{\operatorname{dg}}^b(\mathcal{A}))$ is now acyclic because it is a double chain complex with acyclic rows. Therefore, the associated mixed complex $\operatorname{Mix}(\operatorname{Acy}_{\operatorname{dg}}^b(\mathcal{A}))$ is quasi-isomorphic to the zero mixed complex. \Box

Consider the following functors:

- (a) \overline{C} : $\mathbf{Cat}_k \xrightarrow{\text{split}} \mathbf{Ex}_k \xrightarrow{C} \mathbf{Mix} \xrightarrow{\text{loc}} \mathbf{Mix}_{\infty}$ where the first functor sends a k-linear category to the exact k-linear category with split exact structure and C is Keller's cone functor (3.2.2).
- (b) $\overline{\text{Mix}}: \mathbf{Cat}_k \xrightarrow{\iota} \mathbf{dgcat}_k \xrightarrow{\text{Mix}} \mathbf{Mix} \xrightarrow{\text{loc}} \mathbf{Mix}_{\infty}$ where the first functor associates to a k-linear category the dg-category with the same objects and morphisms complexes concentrated in degree zero (Example A.2.5) and Mix is the functor of Definition 3.1.9.
- (c) $\overline{\operatorname{Mix} \circ \operatorname{Ch}_{\operatorname{dg}}^{b}}$: $\operatorname{Cat}_{k} \xrightarrow{\iota} \operatorname{dgcat}_{k} \xrightarrow{\operatorname{Ch}_{\operatorname{dg}}^{b}} \operatorname{dgcat}_{k} \xrightarrow{\operatorname{Mix}} \operatorname{Mix} \xrightarrow{\operatorname{loc}} \operatorname{Mix}_{\infty}$ where the functor $\operatorname{Ch}_{\operatorname{dg}}^{b}$ sends a k-linear category \mathcal{A} to the dg-category $\operatorname{Ch}_{\operatorname{dg}}^{b}(\mathcal{A})$.

Lemma 3.4.4. Let \overline{C} , $\overline{\text{Mix}}$ and $\overline{\text{Mix} \circ \text{Ch}_{dg}^b}$ be the functors defined in (a),(b) and (c), respectively. Then, there are natural isomorphisms:

- (i) $\overline{\operatorname{Mix}} \xrightarrow{\simeq} \overline{\operatorname{Mix} \circ \operatorname{Ch}_{da}^b};$
- (ii) $\overline{\operatorname{Mix}} \xrightarrow{\simeq} \overline{C}$.
- Proof. (i) Let \mathcal{A} be a k-linear category, seen as dg-category by the functor $\iota: \operatorname{Cat}_k \to \operatorname{dgcat}_k$. Let id be the identity functor of dgcat_k and let $\operatorname{Ch}_{\operatorname{dg}}^b: \operatorname{dgcat}_k \to \operatorname{dgcat}_k$ be the functor sending a dg-category \mathcal{A} to the dg-category $\operatorname{Ch}_{\operatorname{dg}}^b(\mathcal{A})$ of bounded chain complexes in \mathcal{A} . Then, the transformation $\overline{\operatorname{Mix}} \to \overline{\operatorname{Mix}} \circ \operatorname{Ch}_{\operatorname{dg}}^b$ is induced by

the transformation $\eta: \operatorname{id} \to \operatorname{Ch}^b_{\operatorname{dg}}$ defined by

$$\eta_{\mathcal{A}} \colon \mathrm{id}(\mathcal{A}) \to \mathrm{Ch}^{b}_{\mathrm{dg}}(\mathcal{A}),$$

the canonical inclusion of \mathcal{A} in $\mathrm{Ch}^{b}_{\mathrm{dg}}(\mathcal{A})$.

The inclusion of \mathcal{A} in the dg-category of bounded chain complexes represents the inclusion in the pretriangulated hull, as remarked in Example A.2.13, and the inclusion in the pretriangulated hull is a Morita equivalence by Example A.2.15. Therefore, the transformation η induces an equivalence $\operatorname{Mix}(\mathcal{A}) \simeq \operatorname{Mix}(\operatorname{Ch}^b_{\operatorname{dg}}(\mathcal{A}))$ of mixed complexes by Theorem 3.2.5 (2). This is enough to prove that the transformation η induces a natural isomorphism.

(ii) Observe that the functor $\operatorname{Cat}_k \xrightarrow{\iota} \operatorname{dgcat}_k \xrightarrow{\operatorname{Mix}} \operatorname{Mix}$ is equivalent to the functor $\operatorname{Cat}_k \xrightarrow{\iota'} \operatorname{dgcat}_k^{\Delta^1} \to \operatorname{Mix}^{\Delta^1} \xrightarrow{\operatorname{cone}} \operatorname{Mix}$ where a k-linear category \mathcal{A} in Cat_k is sent by the functor ι' to the morphism $0 \to \mathcal{A}$ of dgcat_k ; the other functors are the same functors as in (3.2.2).

Let $j: \mathbf{dgcat}_k \to \mathbf{dgcat}_k^{\Delta^1}$ be the functor sending a dg-category \mathcal{A} to the morphism $\operatorname{Acy}_{\operatorname{dg}}^b(\mathcal{A}) \to \operatorname{Ch}_{\operatorname{dg}}^b(\mathcal{A})$ (defined on morphisms of \mathbf{dgcat}_k in the natural way) and let $\rho: \iota' \to j$ be the transformation defined as:

$$\rho_{\mathcal{A}} \colon \iota'(\mathcal{A}) \to j(\mathcal{A}), \quad (0 \to \mathcal{A}) \mapsto (\operatorname{Acy}_{\operatorname{dg}}^b(\mathcal{A}) \to \operatorname{Ch}_{\operatorname{dg}}^b(\mathcal{A})),$$

Then, ρ induces a transformation $\overline{\text{Mix}} \to \overline{C}$. By Lemma 3.4.3, the mixed complex $\text{Mix}(\text{Acy}_{dg}^b(\mathcal{A}))$ is equivalent to 0 and by the equivalence (i), we have $\text{Mix}(\mathcal{A}) \xrightarrow{\sim} \text{Mix}(\text{Ch}_{dg}^b(\mathcal{A}))$. After applying the cone,we get the equivalence

$$\operatorname{cone}(\operatorname{Mix}(0) \to \operatorname{Mix}(\mathcal{A})) \xrightarrow{\simeq} \operatorname{cone}(\operatorname{Mix}(\operatorname{Acy}_{\operatorname{dg}}^b(\mathcal{A})) \to \operatorname{Mix}(\operatorname{Ch}_{\operatorname{dg}}^b(\mathcal{A}))),$$

i.e., $\overline{\operatorname{Mix}}(\mathcal{A}) \xrightarrow{\simeq} \overline{C}(\mathcal{A})$.

Remark 3.4.5. There are two different cyclic homology theories for k-linear exact categories: McCarthy's cyclic homology [McC94] and Keller's cyclic homology [Kel99]. McCarthy's cyclic homology of a k-linear category \mathcal{A} equipped with the exact structure of split short exact sequences is defined as the cyclic homology of the additive cyclic nerve associated to \mathcal{A} . Keller's cyclic homology of an exact category is defined in terms of the functor C (3.2.2). The above lemma says that McCarthy's Hochschild homology (and cyclic homology) in terms of the additive cyclic nerve [McC94] and Keller's Hochschild homology (and cyclic homology) in terms of the cone construction in the category of mixed complexes are equivalent functors when restricted to the category of small k-linear categories (equipped with the split exact structure). This fact is probably known to the experts, but for convenience of the reader we have provided a proof of it.

Let **Ch** be the category of chain complexes (over k). Recall that there are functors forget: $\mathbf{Mix} \to \mathbf{Ch}$ (3.1.3), sending a mixed complex to the underlying chain complex, and $\mathrm{Tot}(\mathcal{B}-)$: $\mathbf{Mix} \to \mathbf{Ch}$ (3.1.4), sending a mixed complex to the total complex of the associated bicomplex. These functors send quasi-isomorphisms of mixed complexes to quasi-isomorphisms of chain complexes. Hence, they induce the functors

$$\operatorname{forget}_{\infty} \colon \operatorname{Mix}_{\infty} \to \operatorname{Ch}_{\infty}$$
 (3.4.1)

and

$$\operatorname{Tot}(\mathcal{B}-)_{\infty} \colon \operatorname{Mix}_{\infty} \to \operatorname{Ch}_{\infty}$$
 (3.4.2)

where \mathbf{Ch}_{∞} is the ∞ -category of chain complexes.

Definition 3.4.6. Let k be a field, G a group and \mathbf{Ch}_{∞} the ∞ -category of chain complexes (over k). The G-equivariant coarse Hochschild homology $\mathcal{X}\mathrm{HH}_{k}^{G}$ (with k-coefficients) is the G-equivariant \mathbf{Ch}_{∞} -valued coarse homology theory defined as the following composition:

$$\mathcal{X}\mathrm{HH}_k^G \colon G\mathbf{BornCoarse} \xrightarrow{\mathcal{X}\mathrm{Mix}_k^G} \mathbf{Mix}_\infty \xrightarrow{\mathrm{forget}} \mathbf{Ch}_\infty.$$

where $\mathcal{X}Mix_k^G$ is the equivariant coarse homology theory of Definition 3.4.1. The composition

$$\mathcal{X}\mathrm{HC}_{k}^{G}\colon G\mathbf{BornCoarse} \xrightarrow{\mathcal{X}\mathrm{Mix}_{k}^{G}} \mathbf{Mix}_{\infty} \xrightarrow{\mathrm{Tot}(\mathcal{B}-)} \mathbf{Ch}_{\infty}.$$

is called *G*-equivariant coarse cyclic homology.

Theorem 3.4.7. The functors

$$\mathcal{X}\mathrm{HH}_{k}^{G}$$
: $G\mathbf{BornCoarse} \to \mathbf{Ch}_{\infty}$ and $\mathcal{X}\mathrm{HC}_{k}^{G}$: $G\mathbf{BornCoarse} \to \mathbf{Ch}_{\infty}$

of Definition 3.4.6 are G-equivariant coarse homology theories.

Proof. By Theorem 3.4.2, the functor $\mathcal{X}\operatorname{Mix}_k^G : G\operatorname{BornCoarse} \to \operatorname{Mix}_{\infty}$ is an equivariant coarse homology theory and satisfies coarse invariance, coarse excision, *u*-continuity, and vanishing on flasques. The functors $\operatorname{forget}_{\infty} : \operatorname{Mix}_{\infty} \to \operatorname{Ch}_{\infty}(3.4.1)$ and the functor $\operatorname{Tot}(\mathcal{B}-)_{\infty} : \operatorname{Mix}_{\infty} \to \operatorname{Ch}_{\infty}(3.4.2)$ commute with filtered colimits, and send cofiber sequences to cofiber sequences. The two compositions with $\mathcal{X}\operatorname{Mix}_k^G$ satisfy coarse invariance, coarse excision, *u*-continuity, and vanishing on flasques, and the functors $\mathcal{X}\operatorname{HH}_k^G$ and $\mathcal{X}\operatorname{HC}_k^G$ are equivariant coarse homology theories as well. \Box

Question 3.4.8. The analogue functors providing negative and periodic cyclic homologies are not *u*-continuous and not excisive, as direct products do not commute with colimits. Hence, we are not able to construct coarse versions of negative and periodic cyclic homology with the same methods. It remains open whether a different definition of coarse versions of negative and periodic cyclic homologies can be given.

By applying coarse excision, we can compute the value of coarse Hochschild (and cyclic) homology on coarse spaces finitely generated by one point (endowed with trivial G-action).

Example 3.4.9. By Example 1.1.20, we get:

$$\mathcal{X}$$
HH $(\mathbb{R}^n) \cong \Sigma^n \mathcal{X}$ HH $(*)$

where \mathbb{R}^n is endowed with the euclidean metric.

3.5 A symmetric monoidal refinement of $\mathcal{X} H H_k^G$

The goal of this section is to refine the functors $\mathcal{X}HH_k^G$ and $\mathcal{X}HC_k^G$ of Definition 3.4.6 to lax symmetric monoidal functors.

This is achieved by providing lax symmetric monoidal refinements of the functors V_k^G , ι and loc \circ Mix in the definition

$$\mathcal{X}\mathrm{Mix}_k^G \colon G\mathbf{BornCoarse} \xrightarrow{V_k^G} \mathbf{Cat}_k \xrightarrow{\iota} \mathbf{dgcat}_k \xrightarrow{\mathrm{Mix}} \mathbf{Mix} \xrightarrow{\mathrm{loc}} \mathbf{Mix}_\infty$$

of \mathcal{X} Mix $_k^G$.

We start with the functor Mix: $\mathbf{dgcat}_k \to \mathbf{Mix}$; recall that Mix induces a functor $\operatorname{Mix}_{\infty}$: $\mathbf{dgcat}_{k,\infty} \to \mathbf{Mix}_{\infty}$ (3.2.4). By [Kas87, Thm. 2.4], the functor $\operatorname{Mix}_{\infty}$ has a lax symmetric monoidal refinement (see also [CT12]):

Theorem 3.5.1. The functor $\operatorname{Mix}_{\infty}$: $\operatorname{dgcat}_{k,\infty} \to \operatorname{Mix}_{\infty}$ has a a lax symmetric monoidal refinement.

The lax symmetric monoidal structure on the functor $\operatorname{Mix}_{\infty}$: $\operatorname{dgcat}_{k,\infty} \to \operatorname{Mix}_{\infty}$ (3.2.4) induced by [Kas87, Thm. 2.4] uses shuffle products, that for convenience of the reader we now recall, before giving the proof of the theorem.

Remark 3.5.2. The category of mixed complexes has a natural symmetric monoidal structure induced by the canonical symmetric monoidal structure of (the underlying) chain complexes: if (C, b, B) and (C', b', B') are two mixed complexes, then their tensor product is the mixed complex

$$(C \otimes C', b \otimes \mathrm{id} + \mathrm{id} \otimes b, B \otimes \mathrm{id} + \mathrm{id} \otimes B)$$

where all the tensor products are taken over k. The unit object is k.

Let M and N two simplicial modules with associated chain complexes M_* and N_* . The product $M \times N$ is the simplicial module $(M \times N)_n := M_n \otimes N_n$ with face and degeneracies maps $d_i(m \otimes n) := d_i^M(m) \otimes d_i^N(n)$ and $s_j(m \otimes n) := s_j^M(m) \otimes s_j^N(n)$, where d_i^M and s_j^M are the face and degeneracy maps of M and d_i^N and s_j^N are the face and degeneracy maps of N.

Let p, q be two non-negative integer numbers; a (p, q)-shuffle (μ, ν) is a partition of the set of natural numbers $\{0, \ldots, p+q-1\}$ into two disjoint subsets

$$\mu_1 < \mu_2 < \cdots < \mu_p$$
 and $\nu_1 < \nu_2 < \cdots < \nu_q$

The tuple $(\mu_1, \ldots, \mu_p, \nu_1, \ldots, \nu_q)$ is the result of a permutation of the ordered set $(0, \ldots, p + q - 1)$. Let $sgn(\mu, \nu)$ be the sign of this permutation.

For simplicial modules M and N, define a homomorphism (*a shuffle map*) [Lod98, Sec. 1.6.8]

$$\operatorname{sh}: M_* \otimes N_* \to (M \times N)_*$$

in the product $M \times N$ by extending linearly

$$\operatorname{sh}(m \otimes n) = \sum_{(\mu,\nu)} \operatorname{sgn}(\mu,\nu)(s_{\nu_q} \dots s_{\nu_1}(m) \otimes s_{\mu_p} \dots s_{\mu_1}(n))$$
(3.5.1)

where m belongs to M_p , n to N_q and (μ, ν) runs over all the (p, q)-shuffles. The Eilenberg-Zilber Theorem says that the shuffle map sh is a quasi-isomorphism.

Proof of Theorem 3.5.1. The category \mathbf{dgcat}_k has a natural symmetric monoidal structure induced by the object-wise tensor product of Definition A.3.3. By Remark A.3.6, we get a symmetric monoidal ∞ -category $N(\mathbf{dgcat}_k^{\otimes}) \rightarrow N(\mathbf{Fin}_*)$ whose underlying ∞ -category is $N(\mathbf{dgcat}_k)$. As k is a field, this tensor product preserves the Morita equivalences of the Morita model structure on \mathbf{dgcat}_k . Hence, we can consider the symmetric monoidal localization [Hin16, Prop. 3.2.2]

$$\mathbf{dgcat}_{k,\infty}^{\otimes} \coloneqq \mathrm{N}(\mathbf{dgcat}_{k}^{\otimes})[W_{\mathrm{Morita}}^{\otimes,-1}] \to \mathrm{N}(\mathbf{Fin}_{*})$$

whose underlying ∞ -category is $\mathbf{dgcat}_{k,\infty} = \mathrm{N}(\mathbf{dgcat}_k)[W_{\mathrm{Morita}}^{-1}]$ (3.2.3).

By [Hin16, Prop. 3.2.2], we also get a symmetric monoidal ∞ -category

$$\mathbf{Mix}_{\infty}^{\otimes} \coloneqq \mathrm{N}(\mathbf{Mix}^{\otimes})[W_{mix}^{\otimes,-1}] \to \mathrm{N}(\mathbf{Fin}_{*})$$

because the tensor product of mixed complexes preserves equivalences. Its underlying ∞ -category is $\mathbf{Mix}_{\infty} = \mathrm{N}(\mathbf{Mix})[W_{mix}^{-1}]$ (3.1.7).

We observe that one can define the shuffle map (3.5.1) in the category of mixed complexes as well, but the resulting map sh does not commute with the operator B, hence it is not a morphism of mixed complexes. However, thanks to the work of Kassel [Kas87], we know that the shuffle map can be extended to a map commuting with B up to higher homotopies. Therefore, the functor Mix has a lax symmetric monoidal refinement (by [Kas87, Thm. 2.4])

$$\operatorname{Mix}_{\infty}^{\otimes} : \operatorname{dgcat}_{k,\infty}^{\otimes} \to \operatorname{Mix}_{\infty}^{\otimes}$$

induced by the shuffle map.

As a consequence of Kassel's theorem, we also have lax symmetric monoidal refinements of equivariant coarse Hochschild and cyclic homology:

Proposition 3.5.3. Coarse Hochschild homology $\mathcal{X}HH_k^G : GBornCoarse \to Ch_{\infty}$ and coarse cyclic homology $\mathcal{X}HC_k^G : GBornCoarse \to Ch_{\infty}$ admit lax symmetric monoidal refinements:

$$\mathcal{X}\mathrm{HH}_{k}^{G,\otimes} \colon \, \mathrm{N}(G\mathbf{BornCoarse}^{\otimes}) \to \mathbf{Ch}_{\infty}^{\otimes}$$

and

$$\mathcal{X}\mathrm{HC}_{k}^{G,\otimes}\colon\,\mathrm{N}(G\mathbf{BornCoarse}^{\otimes})\to\mathbf{Ch}_{\infty}^{\otimes}.$$

where $\mathbf{Ch}_{\infty}^{\otimes}$ is the ∞ -category of chain complexes with the induced symmetric monoidal structure.

Proof. Consider the functor

 $\mathcal{X}\mathrm{Mix}_k^G\colon G\mathbf{BornCoarse} \xrightarrow{V_k^G} \mathbf{Cat}_k \xrightarrow{\iota} \mathbf{dgcat}_k \xrightarrow{\mathrm{Mix}} \mathbf{Mix} \xrightarrow{\mathrm{loc}} \mathbf{Mix}_\infty$

By Remark 2.2.13, the functor V_k^G : **GBornCoarse** $\to \mathbf{Cat}_k$ admits a lax symmetric monoidal refinement. The inclusion ι : $\mathbf{Cat}_k \to \mathbf{dgcat}_k$ is evidently symmetric monoidal, due to Definition A.3.2, and we get a functor ι_{∞}^{\otimes} : $\mathbf{Cat}_{k,\infty}^{\otimes} \to \mathbf{dgcat}_{k,\infty}^{\otimes}$ of symmetric monoidal ∞ -categories. By Theorem 3.5.1, the functor Mix has a lax symmetric monoidal refinement $\mathrm{Mix}_{\infty}^{\otimes}$: $\mathbf{dgcat}_{k,\infty}^{\otimes} \to \mathbf{Mix}_{\infty}^{\otimes}$, hence so does $\mathcal{X}\mathrm{Mix}_k^G$ (3.4.1):

 $\mathcal{X}\mathrm{Mix}_{k}^{G,\otimes} \colon \mathrm{N}(G\mathbf{BornCoarse}^{\otimes}) \to \mathbf{Mix}_{\infty}^{\otimes}.$

Finally, as the functors forget (3.4.1) and $\text{Tot}(\mathcal{B}-)$ (3.4.2) are lax symmetric monoidal, also coarse Hochschild and cyclic homology admit lax symmetric monoidal refinements. \Box

Chapter 4

A transformation to coarse ordinary homology and further properties

In this chapter we study some additional properties of the functors $\mathcal{X}HH_k^G$ and $\mathcal{X}HC_k^G$, equivariant coarse Hochschild and cyclic homology, of Definition 3.4.6. The main goal is to define a natural transformation from equivariant coarse algebraic *K*-homology $\mathcal{X}K_k^G$ (see Definition 2.3.1) to equivariant coarse ordinary homology $\mathcal{X}H_k^G$ (1.5.6), factoring through coarse Hochschild homology $\mathcal{X}H_k^G$.

The chapter is organized as follows. In the first section, Section 4.1, we give some comparison results: we compare G-equivariant coarse Hochschild and cyclic homology with the classical Hochschild and cyclic homology of algebras. In Section 4.2 we analyze the forget-control map for coarse Hochschild homology and we show an equivalence between the forget-control map for equivariant coarse Hochschild homology and the associated generalized assembly map.

In the third section, Section 4.3, we construct a natural transformation $\Phi_{\mathcal{X}\mathrm{HH}_k^G}$ from equivariant coarse Hochschild homology to equivariant coarse ordinary homology, which is the main result of the chapter. In Section 4.4, we see that the classical Dennis trace map extends to a transformation from *G*-equivariant coarse algebraic *K*-homology to equivariant coarse Hochschild homology. By composition of this natural transformation with the transformation $\Phi_{\mathcal{X}\mathrm{HH}_k^G} : \mathcal{X}\mathrm{HH}_k^G \to \mathcal{X}\mathrm{H}^G$, we get a transformation

$$K\mathcal{X}_k^G \to \mathcal{X}HH_k^G \to \mathcal{X}H^G$$

of spectra-valued coarse homology theories from equivariant coarse algebraic K-homology to equivariant coarse ordinary homology. We conclude the chapter, in Section 4.5, with a Segal-type localization result for coarse Hochschild homology, in the spirit of Section 1.4.

4.1 Comparison results

In this section, we compare equivariant coarse Hochschild homology with the classical version of Hochschild homology for algebras. We show that the evaluation at the one point (bornological coarse) space, endowed with the trivial action, agrees with the Hochschild homology of k (see Proposition 4.1.3) and that the evaluation at the group G, seen as a G-bornological coarse space, is equivalent to the Hochschild homology of the group algebra k[G] (see Proposition 4.1.4).

Let A be a k-algebra, where k is a field. By Example A.4.5, the cyclic nerve $Z_*(A)$ of A is a cyclic module. By Remark 3.1.4, we get a mixed complex that we have denoted by C(A) (consistently with Keller's functor C of Definition 3.2.1).

Notation 4.1.1. Let A be a k-algebra. We denote by

 $C^{\mathrm{HH}}_{*}(A;k)$ and $C^{\mathrm{HC}}_{*}(A;k)$

the chain complexes computing the Hochschild and cyclic homology of the mixed complex C(A) (with k coefficients) as defined in Definition 3.1.3.

Notation 4.1.2. We denote by C_{∞} the composition $C_{\infty} \colon \mathbf{Ex}_k \xrightarrow{C} \mathbf{Mix} \xrightarrow{\mathrm{loc}} \mathbf{Mix}_{\infty}$, where C is the cone construction of Definition 3.2.1 and loc the localization functor to the ∞ -category of mixed complexes (3.1.7).

Observe that the homologies of $C_*^{\text{HH}}(A;k)$ and $C_*^{\text{HC}}(A;k)$ compute the Hochschild and cyclic homology of the k-algebra A (see Example 3.1.5). We shall omit the field of coefficients k, when clear from the context.

Let $\mathcal{X}HH_k$ and $\mathcal{X}HC_k$ be the non-equivariant coarse Hochschild and coarse cyclic homology defined in Definition 3.4.6. Let $\{*\}$ be the one point bornological coarse space, endowed with a trivial *G*-action.

Proposition 4.1.3. There are equivalences of chain complexes

 $\mathcal{X}\mathrm{HH}_k(*) \simeq C^{\mathrm{HH}}_*(k;k) \quad and \quad \mathcal{X}\mathrm{HC}_k(*) \simeq C^{\mathrm{HC}}_*(k;k)$

between the coarse Hochschild (cyclic) homology of the point and the classical Hochschild (cyclic) homology of k.

Proof. Let A be a k-algebra. By Theorem 3.2.5 (1), the mixed complex C(A) associated to A is equivalent to the mixed complex associated to the exact category of finitely generated projective A-modules. When X is a point endowed with a trivial G-action and k is a field, the k-linear category $V_k(X)$ is isomorphic to the category $\mathbf{Vect}_k^{f.d.}$ of finite dimensional k-vector spaces; hence we get the equivalence of mixed complexes

$$C(V_k(\{*\})) = C(\mathbf{Vect}_k^{f.d.}) \simeq C(k),$$

i.e., $\widetilde{\mathcal{XC}}_{k}^{G}(*) \simeq C_{\infty}(k)$; this proves the statement for both coarse Hochschild and coarse cyclic homology by Lemma 3.4.4 and the fact that both the forgetful functor and the functor $\operatorname{Tot}(\mathcal{B}_{-})$ preserve equivalences.

Let G be a group. By Example 1.1.22, there is a canonical G-bornological coarse space $G_{\text{can,min}} = (G, \mathcal{C}_{\text{can}}, \mathcal{B}_{\text{min}})$ associated to it.

Proposition 4.1.4. There are equivalences of chain complexes:

$$\mathcal{X}\mathrm{HH}_{k}^{G}(G_{\mathrm{can,min}}) \simeq C_{*}^{\mathrm{HH}}(k[G];k)$$

and

$$\mathcal{X}\mathrm{HC}_{k}^{G}(G_{\mathrm{can,min}}) \simeq C_{*}^{\mathrm{HC}}(k[G];k)$$

between the G-equivariant coarse Hochschild and cyclic homologies of $G_{\text{can,min}}$ and the classical Hochschild and cyclic homologies of the group algebra k[G].

Proof. The category $V_k^G(G_{\operatorname{can,min}})$ of *G*-equivariant $G_{\operatorname{can,min}}$ -controlled finite dimensional k-vector spaces is equivalent to the category $\operatorname{Mod}^{\operatorname{fg,free}}(k[G])$ of finitely generated free k[G]-modules [BEKW17, Proposition 8.24] (see also Example 2.3.7). By Remark 3.2.6, Keller's mixed complex $C(\operatorname{Mod}^{\operatorname{fg,free}}(k[G]))$ of the category of finitely generated free k[G]-modules is equivalent to the mixed complex associated to the category $\operatorname{Mod}^{\operatorname{fg,proj}}(k[G])$ of finitely generated projective modules. Therefore, the result follows from the chain of equivalences of mixed complexes

$$C(V_k^G(G)) \simeq C(\mathbf{Mod}^{\mathrm{fg,free}}(k[G])) \simeq C(\mathbf{Mod}^{\mathrm{fg,proj}}(k[G])) \simeq C(k[G]),$$

where the last equivalence is given by Theorem 3.2.5 (1), by using the same reasoning of Proposition 4.1.3. $\hfill \Box$

Let X be a G-set and let $X_{\min,\max}$ denote the G-bornological coarse space with minimal coarse structure and maximal bornology. We conclude the section with the calculations on G-bornological coarse spaces of the form $X_{\min,\max} \otimes G_{\operatorname{can,min}}$:

Lemma 4.1.5. Let X be a G-set. Then, we have an equivalence of mixed complexes

$$\widetilde{\mathcal{X}C}_{k}^{G}(X_{\min,\max}\otimes G_{\operatorname{can,min}})\simeq C_{\infty}\left(\operatorname{\mathbf{Vect}}_{k}^{f.d.}*_{G}X\right)$$

where $\mathbf{Vect}_{k}^{f.d.} *_{G} X$ is the category of Definition 2.3.5.

Proof. The result follows from Proposition 2.3.6.

Recall that, when **A** is the additive category of finitely generated free *R*-modules, where *R* is a ring, then the category $\mathbf{A} *_G (G/H)$ is equivalent to the category of finitely generated free R[H]-modules. As in Remark 2.3.7, we also have:

Remark 4.1.6. Let G be a group, H a subgroup of G and endow the set G/H with the minimal coarse structure and the maximal bornology; then, by Lemma 4.1.5, we get an equivalence of chain complexes:

$$\mathcal{X}\mathrm{HH}_{k}^{G}((G/H)_{\min,\max}\otimes G_{\operatorname{can},\min})\simeq C_{*}^{\operatorname{HH}}(k[H];k);$$

the same holds for equivariant coarse cyclic homology.

4.2 The forget-control map for coarse Hochschild homology

In Section 1.3, we have discussed how to get equivariant homology theories from equivariant coarse homology theories and we have defined the forget-control and the assembly maps. In this section, we give a comparison result between the forget-control maps for equivariant coarse Hochschild and cyclic homology and the associated assembly maps (see Proposition 4.2.9). In order to apply the results of Section 1.3, we first need to prove that coarse Hochschild and cyclic homologies are strong (Lemma 4.2.1) and continuous (Lemma 4.2.3) equivariant coarse homology theories.

Recall the definitions of strongness (Definition 1.2.9) for a coarse homology theory. In the following proofs, we use the equivariant coarse homology theory $\widetilde{\mathcal{XC}}_{k}^{G}$ (3.3.1) (see Remark 3.3.3). As in Notation 4.1.2, we denote by C_{∞} the composition loc $\circ C$.

Lemma 4.2.1. The functor $\mathcal{X}HH_k^G : GBornCoarse \to Ch_{\infty}$ is strong.

Proof. Let X be a weakly flasque space as defined in Definition 1.2.8. Let $f: X \to X$ be a morphism that implements the weak flasqueness of X. Observe that the functor

$$S \coloneqq \bigoplus_{n \in \mathbb{N}} (f^n)_* \colon V_k^G(X) \to V_k^G(X).$$

defined in Lemma 2.1.15 is still well-defined. By Definition 1.2.8, we get the equivalence $\operatorname{id}_{\widetilde{\mathcal{XC}}_k^G(X)} \simeq \widetilde{\mathcal{XC}}_k^G(f).$

The functor S is naturally isomorphic to the functor $\operatorname{id}_{V_k^G(X)} \oplus f_* \circ S$. By Theorem 3.3.5, we get the chain of equivalences of mixed complexes

$$C_{\infty}(\mathrm{id}_{V_{k}^{G}(X)} \oplus f_{*} \circ S) \simeq C_{\infty}(\mathrm{id}_{V_{k}^{G}(X)}) \oplus C_{\infty}(f_{*} \circ S) \simeq \mathrm{id}_{\widetilde{\mathcal{XC}}_{k}^{G}(X)} \oplus \widetilde{\mathcal{XC}}_{k}^{G}(f) \circ C_{\infty}(S)$$

hence,

$$C_{\infty}(S) \simeq \operatorname{id}_{\mathcal{X}C_{k}^{G}(X)} + C_{\infty}(S);$$

this shows that the morphism $\operatorname{id}_{\mathcal{X}C_k^G(X)}$ is equivalent to zero. By Lemma 3.4.4, also the chain complex $\mathcal{X}\operatorname{HH}_k^G(X)$ is equivalent to 0.

As a corollary of the previous lemma, we get:

Corollary 4.2.2. The functor $\mathcal{X}\mathrm{HC}_k^G : G\mathbf{BornCoarse} \to \mathbf{Ch}_\infty$ is strong.

Proof. Equivariant coarse cyclic homology $\mathcal{X}\mathrm{HC}_k^G$ has been defined (Definition 3.4.6) as the composition of the functors $\mathcal{X}\mathrm{Mix}_k^G$ and $\mathrm{Tot}(\mathcal{B}-)$; this last functor preserves equivalences. Hence, the result follows from Lemma 4.2.1.

Recall the definition of continuity (Definition 1.2.13) for a coarse homology theory.

Lemma 4.2.3. The functor $\mathcal{X}HH_k^G$: GBornCoarse \to Ch_{∞} is continuous.

Proof. Let \mathcal{Y} be a trapping exhaustion of X. Let (M, ρ) be a G-equivariant X-controlled finite dimensional k-vector space of $V_k^G(X)$ and denote by F the subset

$$F = \{ x \in X \mid M(\{x\}) \neq 0 \}$$

of X. The set F is locally finite: if B is a bounded set of X, then $F \cap B = \sigma(B)$, where σ is the support function of (M, ρ) (Definition 2.1.5). Therefore, the controlled module (M, ρ) belongs to the subcategory $V_k^G(F)$ of $V_k^G(X)$.

This proves that the category $V_k^G(X)$ is the filtered colimit of subcategories $V_k^G(F)$ indexed on locally finite subsets of X:

$$V_k^G(X) = \operatorname{colim}_{F \subseteq X} V_k^G(F).$$

As the mixed complex functor and the forgetful functor commute with filtered colimits, we get the equivalence $\mathcal{X}HH_k^G(\mathcal{Y}) \simeq \mathcal{X}HH_k^G(X)$. This shows that equivariant coarse Hochschild homology is continuous.

As the functor $\text{Tot}(\mathcal{B}-)$ commutes with filtered colimits as well, the same proof of Lemma 4.2.3 shows the following:

Corollary 4.2.4. The functor $\mathcal{X}\mathrm{HC}_k^G : G\mathbf{BornCoarse} \to \mathbf{Ch}_\infty$ is continuous.

We now proceed with the comparison result of the forget-control map for coarse Hochschild homology and the associated assembly map. Recall, from Section 1.3, the definitions of the cone $\mathcal{O}_{hlg}^{\infty}$ (1.3.3), of the forget-control map β (Definition 1.3.13) and of the coarse assembly map α (Definition 1.3.19). By [BEKW17, Theorem 11.16], the forget-control map for a *G*-equivariant coarse homology theory *E* can be compared with the classical assembly map for the associated *G*-equivariant homology theory $E \circ$ $\mathcal{O}_{hlg}^{\infty}$: *G***Top** \rightarrow **C** (see Remark 1.3.15).

By applying the Eilenberg-MacLane correspondence (1.5.5), we can assume that the equivariant coarse homology theories $\mathcal{X}HH_k^G$ and $\mathcal{X}HC_k^G$ are equivariant spectra-valued coarse homology theories.

Definition 4.2.5. Let $\mathbf{HH}_{k}^{G} \coloneqq \mathcal{X}\mathrm{HH}_{k}^{G} \circ \mathcal{O}_{\mathrm{hlg}}^{\infty}$ be the *G*-equivariant homology theory associated to equivariant coarse Hochschild homology (with *k*-coefficients).

Let $i: GOrb \to GBornCoarse$ be the functor that associates to a transitive *G*-set *S* in *GOrb* the *G*-bornological coarse space $S_{\min,\max}$ with the minimal coarse structure and maximal bornology. Recall that a coarse homology theory can be twisted by a bornological coarse space (see Definition 1.2.5).

Remark 4.2.6. By Lemma 4.1.5, if $\mathcal{X}HH^G_{k,G_{\operatorname{can,min}}}$ is the twist of the coarse homology theory $\mathcal{X}HH^G_k$ by the *G*-bornological coarse space $G_{\operatorname{can,min}}$, then we get

$$\mathbf{HH}_{k}^{G} \simeq \mathcal{X} \mathrm{HH}_{k,G_{\mathrm{can}\,\mathrm{min}}}^{G} \circ i.$$

Let **Fin** be the family of finite subgroups of G.

Proposition 4.2.7. The forget-control map $\beta_{G_{\text{can,min}},G_{\text{max,max}}}$ for $\mathcal{X}\text{HH}_k^G$ is equivalent to the assembly map $\alpha_{E_{\text{Fin}}G,G_{\text{can,min}}}$ for the G-homology theory \mathbf{HH}_k^G .

Proof. The functors $\mathcal{X}HH_k^G$ and $\mathcal{X}HC_k^G$ are strong by Lemma 4.2.1, and continuous by Lemma 4.2.3. Then, the result follows by Theorem 1.3.20.

Remark 4.2.8. The assembly map $\alpha_{E_{\text{Fin}}G,G_{\text{can,min}}}$ for the *G*-homology theory HH_k^G (hence, the forget-control map $\beta_{G_{\text{can,min}},G_{\text{max,max}}}$ for $\mathcal{X}\text{HH}_k^G$) is split injective by [LR06, Thm 1.7].

By using the same reasoning, we get the same result for equivariant coarse cyclic homology:

Proposition 4.2.9. The forget-control map $\beta_{G_{\operatorname{can,min}},G_{\max,\max}}$ for $\mathcal{X}\operatorname{HC}_k^G$ is equivalent to the assembly map $\alpha_{E_{\operatorname{Fin}}G,G_{\operatorname{can,min}}}$ for the G-homology theory $\operatorname{HC}_k^G \coloneqq \mathcal{X}\operatorname{HC}_k^G \circ \mathcal{O}_{hlg}^\infty$.

4.3 A transformation to coarse ordinary homology

Let $\mathcal{X}HH_k^G, \mathcal{X}C^G: GBornCoarse \to Ch_{\infty}$ denote the *G*-equivariant coarse Hochschild homology (see Definition 3.4.6) and the equivariant (chain complexes-valued) ordinary homology (see Section 1.5) with coefficients in a field *k*, respectively. The goal of this section is to define a natural transformation

 $\Phi_{\mathcal{X}\mathrm{HH}_{k}^{G}}\colon \mathcal{X}\mathrm{HH}_{k}^{G}\longrightarrow \mathcal{X}C^{G}$

from equivariant coarse Hochschild homology to the chain valued equivariant coarse ordinary homology $\mathcal{X}C^G$, and a natural transformation $\Phi_{\mathcal{X}\mathrm{HC}^G_k}$ from equivariant coarse cyclic homology.

The idea of our construction uses the fact that, if $A_0 \otimes A_1 \otimes \cdots \otimes A_n$ belongs to the additive cyclic nerve $CN_n(V_k^G(X))$ associated to the category of controlled objects $V_k^G(X)$ (see Definition 2.1.2 and Definition A.4.7), then, after evaluation at (n+1)-tuples (x_0, \ldots, x_n) of points of X, there are well-defined linear operators $(A_0 \circ \cdots \circ A_n)|(x_0, \ldots, x_n)$ (see Notation 4.3.2) which are endomorphisms of finite dimensional k-vector spaces; we associate to a such linear operator its trace, *i.e.*, an element of k. This will give well-defined chains in coarse ordinary homology and a natural transformation $\Phi_{\mathcal{X}HH_k^G}$ (see Theorem 4.3.8) of equivariant coarse homology theories.

Recall that $V_k^G(X)$ denotes the k-linear category of G-equivariant X-controlled (finite dimensional) k-vector spaces (Definition 2.1.8). Recall that the objects of $V_k^G(X)$ are pairs (M, ρ) where M is a functor on the category associated to the poset of bounded sets of X.

The construction of the transformation $\Phi_{\mathcal{X}HH^G}$ proceeds as follows:

- 1. For every G-bornological coarse space X, we consider its associated k-linear category $V_k^G(X)$ of controlled objects, hence the associated additive cyclic nerve $\operatorname{CN}(V_k^G(X))$ (see Example A.4.6);
- 2. for every tensor element $A_0 \otimes \ldots \otimes A_n$ in the additive cyclic nerve of $V_k^G(X)$ and every n + 1 points x_0, \ldots, x_n of X, we define a trace-like map, which gives an element of k (see Notation 4.3.2);
- 3. by letting x_0, \ldots, x_n vary, this yields a *G*-equivariant locally finite controlled chain on *X*, *i.e.*, an element of $\mathcal{X}C_n^G(X)$ (see Definition 4.3.3 and Lemma 4.3.4);
- 4. by letting $A_0 \otimes \ldots \otimes A_n$ vary we get a map $\varphi \colon \operatorname{CN}_*(V_k^G(X)) \to \mathcal{X}C_*^G(X)$ that is a chain map with respect to the differential $d = \sum d_i$ of $\operatorname{CN}(V_k^G(X))$ (see Proposition 4.3.6);
- 5. the additive cyclic nerve $\operatorname{CN}(V_k^G(X))$ yields a mixed complex with the differentials b and B (3.1.6) and the chain map φ extends to a map of mixed complexes $\tilde{\varphi}$ (see Lemma 4.3.7);
- 6. the map $\tilde{\varphi}$ yields a natural transformation of equivariant coarse homology theories $\Phi_{\mathcal{X}HH_{c}^{G}} \colon \mathcal{X}HH_{k}^{G} \longrightarrow \mathcal{X}C^{G}$ (see Theorem 4.3.8).

We now proceed with the construction of the natural transformation $\Phi_{\mathcal{X}HH_k^G} \colon \mathcal{X}HH_k^G \to \mathcal{X}C^G$.

Let $n \ge 0$ be a natural number. The *n*-th component of the additive cyclic nerve of $V_k^G(X)$ (see Definition A.4.7 or Example A.4.6) is :

$$\operatorname{CN}_n(V_k^G(X)) = \bigoplus_{((M_0,\rho_0),\dots,(M_n,\rho_n))} \left(\bigotimes_{i=0}^n \operatorname{Hom}((M_{i+1},\rho_{i+1}),(M_i,\rho_i))\right)$$

where the index *i* runs cyclically in the set $\{0, \ldots, n\}$ and the sum ranges over all the tuples $((M_0, \rho_0), \ldots, (M_n, \rho_n))$ of objects of $V_k^G(X)$.

Remark 4.3.1. For every controlled morphism $A_i: (M_{i+1}, \rho_{i+1}) \to (M_i, \rho_i)$ (see Definition 2.1.6) in Hom $((M_{i+1}, \rho_{i+1}), (M_i, \rho_i))$ and for every pair of points x and y of X, there is a well-defined linear map

$$A_i^{x,y} \colon M_{i+1}(x) \to M_i(y)$$

(see also (2.2.4), where the notation is $A_{i,y,x}$).

In order to simplify the exposition of later proofs, we use the following notation:

Notation 4.3.2. Let $A_0 \otimes \ldots \otimes A_n$ be an element of $\bigotimes_{i=0}^n \operatorname{Hom}((M_{i+1}, \rho_{i+1}), (M_i, \rho_i))$ with $A_i: (M_{i+1}, \rho_{i+1}) \to (M_i, \rho_i)$ and $((M_0, \rho_0), \ldots, (M_n, \rho_n))$ a tuple of objects of $V_k^G(X)$. Let (x_0, \ldots, x_n) be a point of X^{n+1} . The symbol

$$(A_0 \circ \cdots \circ A_n)|(x_0, \ldots, x_n)$$

denotes the linear operator $(A_0 \circ \cdots \circ A_n)|(x_0, \ldots, x_n) \colon M_0(x_n) \to M_0(x_n)$ defined as the composition

$$(A_0 \circ \ldots \circ A_n) | (x_0, \ldots, x_n) \coloneqq M_0(x_n) \xrightarrow{A_n^{x_n, x_{n-1}}} M_n(x_{n-1}) \xrightarrow{A_{n-1}^{x_{n-1}, x_{n-2}}} \ldots \xrightarrow{A_1^{x_1, x_0}} M_1(x_0) \xrightarrow{A_0^{x_0, x_n}} M_0(x_n)$$

of the induced operators $A_i^{x_i,x_{i+1}} \colon M_i(x_i) \to M_{i+1}(x_{i+1})$. It is an endomorphism of $M_0(x_n)$, which is a finite dimensional k-vector space.

Let X be a G-bornological coarse space and let $\mathcal{X}C_n(X)$ be the k-linear vector space generated by the locally finite controlled n-chains on X (see Definition 1.5.1).

Definition 4.3.3. Let X be a G-bornological coarse space and let n be a natural number. We let $\varphi_n \colon \operatorname{CN}_n(V_k^G(X)) \to \mathcal{X}C_n(X)$ be the map defined on elementary tensors as

$$\varphi_n \colon A_0 \otimes \ldots \otimes A_n \longmapsto \sum_{(x_0, \ldots, x_n) \in X^{n+1}} \operatorname{tr} \left((A_0 \circ \cdots \circ A_n) | (x_0, \ldots, x_n) \colon M_0(x_n) \to M_0(x_n) \right) \cdot (x_0, \ldots, x_n)$$

where we use Notation 4.3.2 and the symbol tr denotes the trace map.

The map φ_n is extended to $CN_n(V_k^G(X))$ by linearity.

Lemma 4.3.4. Let X be a G-bornological coarse space and let $A_0 \otimes \ldots \otimes A_n$ be a tensor element of $CN_n(V_k^G(X))$ with $A_i: (M_{i+1}, \rho_{i+1}) \to (M_i, \rho_i)$. Then, the n-chain $\varphi_n(A_0 \otimes \ldots \otimes A_n)$, given by the sum

$$\sum_{(x_0,\ldots,x_n)\in X^{n+1}} \operatorname{tr}((A_0\circ\cdots\circ A_n)|(x_0,\ldots,x_n))\cdot(x_0,\ldots,x_n),$$

of Definition 4.3.3, is locally finite and controlled.

Proof. In order to prove that $\varphi_n(A_0 \otimes \ldots \otimes A_n)$ is locally finite and controlled we show that its support (1.5.1) is locally finite and that there exists an entourage U of X such that every $x = (x_0, \ldots, x_n)$ in $\operatorname{supp}(\varphi_n(A_0 \otimes \ldots \otimes A_n))$ is U-controlled.

We first observe that the operators $A_i: (M_{i+1}, \rho_{i+1}) \to (M_i, \rho_i)$ are U_i -controlled for some entourage U_i of X. By Definition 2.1.6, A_i is given by a natural transformation of functors $M_{i+1} \to M_i \circ U_i[-]$ satisfying an equivariance condition. For every point x in X, by using the Convention 2.2.4, A_i restricts to a morphism

$$M_{i+1}(x) \to M_i(U_i[x]) \cong \bigoplus_{x' \in U_i[x]} M_i(x')$$

where the direct sum has only finitely many non-zero summands.

Let K be a bounded set of X. The set of points $x_n \in K$ for which $M_0(x_n)$ is non-zero is finite (as a consequence of Definition 2.1.8). For such a fixed x_n , there are only finitely many points $x_{n-1} \in U_n[K]$ such that the corresponding map $A_n^{x_n,x_{n-1}} \colon M_0(x_n) \to M_n(x_{n-1})$ is non-zero. The set $U_n[K]$ is a bounded set of X, the morphism $A_i \colon M_{i+1} \to M_i$ is U_i -controlled and we can repeat the same argument for each A_i . This implies that the *n*-chain is locally finite because, for the given bounded set K, we have found only finitely tuples (x_0, \ldots, x_n) in the support of $\varphi_n(A_0 \otimes \ldots \otimes A_n)$ that meet K.

The chain is also U-controlled, where U is the entourage $U := U_0 \circ \cdots \circ U_n$ of X. \Box

Remark 4.3.5. Let X be a G-bornological coarse space. Let (M, ρ) be a G-equivariant X-controlled finite dimensional k-vector space and let g be an element of the group G. Then, $\rho(g)$ (Definition 2.1.2) is a natural isomorphism between the functors M and gM. The morphisms in the category $V_k^G(X)$ satisfy a G-equivariant condition (see Definition 2.1.6). These observations imply that the following diagram is commutative

$$\begin{array}{cccc} M_0(gx_n) \xrightarrow{A_n^{gx_n,gx_n-1}} & M_n(gx_{n-1}) & \longrightarrow & \dots \xrightarrow{A_0^{gx_0,gx_n}} & M_0(gx_n) \\ & & \downarrow \cong & & \downarrow \cong & & \downarrow \cong \\ gM_0(x_n) \xrightarrow{gA_n^{x_n,x_{n-1}}} & gM_n(x_{n-1}) & \longrightarrow & \dots \xrightarrow{gA_0^{x_0,x_n}} & gM_0(x_n) \end{array}$$

for $A_0 \otimes \ldots \otimes A_n$ in $CN_n(V_k^G(X))$ with $A_i: (M_{i+1}, \rho_{i+1}) \to (M_i, \rho_i)$, where the isomorphisms are induced by $\rho_i(g)$. Hence, the image of φ_n is a *G*-invariant locally finite controlled *n*-chain on *X*.

Let $\partial_i \colon \mathcal{X}C_n^G(X) \to \mathcal{X}C_{n-1}^G(X)$ be the *i*-th differential of the chain complex $\mathcal{X}C^G(X)$ and let $d_i \colon \operatorname{CN}_n(V_k^G(X)) \to \operatorname{CN}_{n-1}(V_k^G(X))$ be the *i*-th face map of $\operatorname{CN}(V_k^G(X))$ (see Example A.4.6). Consider the chain complex underlying the additive cyclic nerve $\operatorname{CN}(V_k^G(X))$.

Proposition 4.3.6. Let X be a G-bornological coarse space. The maps $\varphi_n \colon CN_n(V_k^G(X)) \to \mathcal{X}C_n^G(X)$ of Definition 4.3.3 extend to a chain map

$$\varphi \colon (\operatorname{CN}(V_k^G(X)), d) \to (\mathcal{X}C^G(X), \partial).$$

Proof. We prove that the following square is commutative:

$$\begin{array}{cccc}
\operatorname{CN}_{n}(V_{k}^{G}(X)) & \xrightarrow{\varphi_{n}} & \mathcal{X}C_{n}^{G}(X) \\
& & & \downarrow^{d_{i}} & & \partial_{i} \downarrow \\
\operatorname{CN}_{n-1}(V_{k}^{G}(X)) & \xrightarrow{\varphi_{n-1}} & \mathcal{X}C_{n-1}^{G}(X)
\end{array}$$

Consider the case $i \neq n$ and let $A_0 \otimes \ldots \otimes A_n$ be an elementary tensor.

By Definition A.4.7 and by using Notation 4.3.2, we get:

$$\varphi_{n-1}(d_i(A_0 \otimes \ldots \otimes A_n)) =$$

$$= \sum_{(x'_0, \dots, \hat{x}'_i, \dots, x'_n)} \operatorname{tr}(A_0 \circ \cdots \circ (A_i \circ A_{i+1}) \circ \cdots \circ A_n | (x'_0, \dots, \hat{x}'_i, \dots, x'_n)) \cdot (x'_0, \dots, \hat{x}'_i, \dots, x'_n)$$

$$= \sum_{(x'_0, \dots, \hat{x}'_i, \dots, x'_n)} \left(\sum_{x'_i \in X} \operatorname{tr}(A_0 \circ \cdots \circ A_n | (x'_0, \dots, x'_i, \dots, x'_n)) \right) \cdot (x'_0, \dots, \hat{x}'_i, \dots, x'_n)$$

where we used that the trace is additive and that the morphism $A_i \circ A_{i+1}$ factors through all the points x'_i of X (that give contribution zero up to finitely many).

On the other hand, we also get

$$\partial_i(\varphi_n(A_0 \otimes \ldots \otimes A_n)) =$$

$$= \partial_i \left(\sum_{(x'_0, \dots, x'_i, \dots, x'_n)} \operatorname{tr}(A_0 \circ \dots \circ A_n | (x'_0, \dots, x'_i, \dots, x'_n)) \right) \cdot (x'_0, \dots, x'_i, \dots, x'_n)$$

$$= \sum_{(x'_0, \dots, \hat{x}'_i, \dots, x'_n)} \operatorname{tr}(A_0 \circ \dots \circ A_n | (x'_0, \dots, x'_i, \dots, x'_n)) \cdot (x'_0, \dots, \hat{x}'_i, \dots, x'_n)$$

and, by gathering together the preimages, we can write the sum as follows:

$$\partial_i(\varphi_n(A_0 \otimes \ldots \otimes A_n)) =$$

$$= \sum_{(x'_0, \dots, \hat{x}'_i, \dots, x'_n)} \left(\sum_{x'_i \in X} \operatorname{tr}(A_0 \circ \cdots \circ A_n | (x'_0, \dots, x'_i, \dots, x'_n)) \right) \cdot (x'_0, \dots, \hat{x}'_i, \dots, x'_n)$$

Therefore, for the *i*-face maps d_i and ∂_i , with $i \neq n$, the above diagram is commutative.

For i = n, we recall that $d_n(A_0 \otimes \cdots \otimes A_n) = (A_n \circ A_0 \otimes A_1 \otimes \cdots \otimes A_{n-1})$. Hence, the above diagram commutes for i = n as well because the trace map is invariant under cyclic permutations.

Let (M, b) be a cyclic module. By Remark 3.1.4, we get a mixed complex whose differential b is the differential $d = \sum_{i=0}^{n} (-1)^{i} d_{i}$ of above. We recall that the operator B (3.1.6) is defined by $B \coloneqq (-1)^{n+1} (1 - t_{n+1}) sN$, where s denotes the extra degeneracy

Lemma 4.3.7. Let X be a G-bornological coarse space. The chain map $\varphi \colon CN(V_k^G(X)) \to \mathcal{X}C^G(X)$ of Definition 4.3.3 extends to a map

$$\tilde{\varphi} \colon \operatorname{Mix}(V_k^G(X)) \to \mathcal{X}C^G(X)$$

that is a morphism of mixed complexes.

Proof. Let $A_0 \otimes \ldots \otimes A_n$ be an element of $CN_n(V_k^G(X))$. Then, $N(A_0 \otimes \ldots \otimes A_n) = \sum_{i=0}^n A_i \otimes \ldots \otimes A_{i+n}$. We have:

$$sN(A_0 \otimes \ldots \otimes A_n) = (-1)^{n+1} \sum_{i=0}^n 1 \otimes A_i \otimes \ldots \otimes A_{i+n}$$

and then

$$B(A_0 \otimes \ldots \otimes A_n) = \sum_{i=0}^n 1 \otimes A_i \otimes \ldots \otimes A_{i+n} - \sum_{i=0}^n A_{i+n} \otimes 1 \otimes A_i \otimes \ldots \otimes A_{i+n-1}$$
$$= \sum_{i=0}^n (1 \otimes A_i \otimes \ldots \otimes A_{i+n} - A_{i+n} \otimes 1 \otimes A_i \otimes \ldots \otimes A_{i+n-1}).$$

By applying the map φ_n , we see that all the terms of $B(A_0 \otimes \ldots \otimes A_n)$ cancel pairwise, because the trace is invariant under cyclic permutations. This is enough to show that the map φ extends to $\operatorname{Mix}(V_k^G(X))$.

We can finally construct the natural transformation $\Phi_{\mathcal{X}HH_{L}^{G}}$: $\mathcal{X}HH_{k}^{G} \to \mathcal{X}C^{G}$:

Theorem 4.3.8. There are natural transformations

$$\Phi_{\mathcal{X}\mathrm{HH}_{k}^{G}} \colon \mathcal{X}\mathrm{HH}_{k}^{G} \to \mathcal{X}C^{G}.$$

and

$$\Phi_{\mathcal{X}\mathrm{HC}_{k}^{G}} \colon \mathcal{X}\mathrm{HC}_{k}^{G} \to \bigoplus_{n \in \mathbb{N}} \mathcal{X}C^{G}$$

of G-equivariant \mathbf{Ch}_{∞} -valued coarse homology theories.

Proof. The map $\varphi : (\operatorname{CN}(V_k^G(X)), d) \to (\mathcal{X}C^G(X), \partial)$ of Definition 4.3.3 is a chain map by Proposition 4.3.6. Let $f : X \to Y$ be a morphism of *G*-equivariant bornological coarse spaces. Consider the induced chain map $\mathcal{X}C^G(f) : \mathcal{X}C^G(X) \to \mathcal{X}C^G(Y)$ (1.5.2) and the induced functor $f_* = V_k^G(f) : V_k^G(X) \to V_k^G(X')$ (2.1.2). By functoriality of the additive cyclic nerve, f_* induces a morphism $\operatorname{CN}(f_*) : \operatorname{CN}_*(V_k^G(X)) \to \operatorname{CN}_*(V_k^G(Y))$ of cyclic modules (hence, a chain map between the underlying chain complexes as well). The diagram

$$\begin{array}{ccc} \operatorname{CN}_{n}(V_{k}^{G}(X)) & \stackrel{\varphi_{n}}{\longrightarrow} \mathcal{X}C_{n}^{G}(X) \\ & & & & \downarrow^{\operatorname{CN}(f_{*})} & & \downarrow^{\mathcal{X}C^{G}(f)} \\ \operatorname{CN}_{n}(V_{k}^{G}(Y)) & \stackrel{\varphi_{n}}{\longrightarrow} \mathcal{X}C_{n}^{G}(Y) \end{array}$$

is commutative, as we now explain. If $A_0 \otimes \ldots \otimes A_n$ is a tensor element of the additive cyclic nerve associated to $V_k^G(X)$, then $\operatorname{CN}(f_*)(A_0 \otimes \ldots \otimes A_n)$ is the elementary tensor $f_*A_0 \otimes \ldots \otimes f_*A_n$ and belongs to the tensor product $\bigotimes_{i=0}^n \operatorname{Hom}_{V_k^G(Y)}(f_*(M_{i+1}, \rho_{i+1}), f_*(M_i, \rho_i))$. After application of the map φ_n we get:

$$\varphi_n(f_*A_0 \otimes \ldots \otimes f_*A_n) = \sum_{(y_0,\ldots,y_n)} \operatorname{tr} \left((f_*A_0 \circ \cdots \circ f_*A_n) | (y_0,\ldots,y_n) \right) \cdot (y_0,\ldots,y_n).$$

By definition of f_*A_i (2.1.1), the composition $(f_*A_0 \circ \cdots \circ f_*A_n)|(y_0, \ldots, y_n)$ decomposes as a sum over the fibers $(x_0, \ldots, x_n) \in X^n$ of f at (y_0, \ldots, y_n) , *i.e.*,

$$\varphi_n(f_*A_0 \otimes \ldots \otimes f_*A_n) = \sum_{(y_0, \dots, y_n)} \left(\sum_{(x_0, \dots, x_n)} \operatorname{tr} \left((A_0 \circ \cdots \circ A_n) | (x_0, \dots, x_n) \right) \right) \cdot (y_0, \dots, y_n)$$

where the second sum runs over the fibers of (y_0, \ldots, y_n) . But this last term is in fact $\mathcal{X}C^G(f)(\varphi_n(A_0 \otimes \ldots \otimes A_n))$, proving that the diagram of above is commutative.

The map φ extends to the associated mixed complexes by Lemma 4.3.7 and this extension preserves the commutative diagram (of associated mixed complexes). After localization and application of the forgetful functor (recall the definition of equivariant coarse Hochschild homology in terms of $\mathcal{X}\text{Mix}_k^G$, Definition 3.4.6), the map φ yields a natural transformation of equivariant coarse homology theories

$$\Phi_{\mathcal{X}\mathrm{HH}_{k}^{G}} \colon \mathcal{X}\mathrm{HH}_{k}^{G} \to \mathcal{X}C^{G}.$$

To every mixed complex C, we associate the chain complex $\text{Tot}(\mathcal{B}C)$ (3.1.2) defined by $\text{Tot}_n(\mathcal{B}C) = \bigoplus_{i\geq 0} C_{n-2i}$ with differential $d(c_n, c_{n-2}, \dots) = (bc_n + Bc_{n-2}, \dots)$. By Lemma 4.3.7, we conclude that the map φ extends to a chain map on the total complex as well, and to a natural transformation of coarse homology theories

$$\Phi_{\mathcal{X}\mathrm{HC}_k^G} \colon \, \mathcal{X}\mathrm{HC}_k^G \to \bigoplus_{n \in \mathbb{N}} \mathcal{X}C^G$$

where the sum is indexed on the natural numbers because the (mixed complex associated to) the additive cyclic nerve of $V_k^G(X)$ is positively graded.

The following result implies that the transformation $\Phi_{\mathcal{X}HH_k}$: $\mathcal{X}HH_k \to \mathcal{X}C$ is non-zero.

Proposition 4.3.9. If X is the one point space $\{*\}$, then the transformation

$$\Phi_{\mathcal{X}\mathrm{HH}_k} \colon \mathcal{X}\mathrm{HH}_k(*) \to \mathcal{X}C(*)$$

induces an equivalence of chain complexes.

Proof. Let $c: \{*\}^{n+1} \to k$ be an *n*-chain in $\mathcal{X}C_n(*)$; we identify this chain with the element $c \in k$ that is its image. Let $\iota_n: \mathcal{X}C_n(*) \to \operatorname{CN}_n(V_k(*))$ be the map sending c to the element $(\cdot c) \otimes (\cdot 1_k) \otimes \ldots \otimes (\cdot 1_k)$. This extends to a chain map that gives a section of the trace map, *i.e.*, $\varphi \circ \iota = \operatorname{id}$.

As coarse Hochschild homology and coarse ordinary homology of the point are both isomorphic to the Hochschild homology of the ground field k (by Example 1.5.5 and Proposition 4.1.3), we get equivalences of chain complexes

$$\mathcal{X}$$
HH_k(*) $\simeq C_*^{\text{HH}}(k) \simeq \mathcal{X}C_k(*).$

By using these equivalences and the section $\varphi \circ \iota = \text{id}$, we obtain that, when X is the one point space, the transformation $\Phi_{\mathcal{X}\text{HH}_k}$ induces an equivalence of chain complexes $\mathcal{X}\text{HH}_k(*) \to \mathcal{X}C(*)$.

4.4 A transformation from coarse algebraic *K*-homology

In this section, we construct natural transformations $K\mathcal{X}_k^G \to \mathcal{X}\mathrm{HH}_k^G$ and $K\mathcal{X}_k^G \to \mathcal{X}\mathrm{HC}_k^G$ from equivariant coarse algebraic K-homology (see Definition 2.3.1) to equivariant coarse Hochschild and cyclic homology (Definition 3.4.6). In order to construct these natural transformations we use the trace maps constructed by McCarthy [McC94, Sec. 4.4] and Keller [Kel99, Theorem 1.13]. As equivariant coarse algebraic K-homology takes values in spectra, by applying the Eilenberg-MacLane correspondence \mathcal{EM} (1.5.5), we assume in this section that also equivariant coarse Hochschild and cyclic homology take values in the ∞ -category **Sp** of spectra.

Let $K\mathcal{X}_k^G$ be the equivariant coarse algebraic K-homology (Definition 2.3.1). Let HH^{McC} and HH^{Kel} denote McCarthy's [McC94] and Keller's [Kel99] Hochschild homology, respectively. Classically, there are trace maps from algebraic K-theory to (topological) Hochschild homology and its cyclic variants. For example, the Dennis trace map is a transformation from the algebraic K-theory of rings to Hochschild homology. McCarthy has extended the Dennis trace map to a transformation from the algebraic K-theory (of exact categories) to (McCarthy's) Hochschild homology of exact categories [McC94, Sec. 4.4]. We use this extension to construct a transformation from equivariant coarse algebraic K-homology to equivariant coarse Hochschild homology:

Proposition 4.4.1. There are natural transformations

$$\mathrm{K}\mathcal{X}_k^G \to \mathcal{X}\mathrm{HH}_k^G \quad and \quad \mathrm{K}\mathcal{X}_k^G \to \mathcal{X}\mathrm{HC}_k^G$$

induced by the Dennis trace maps from algebraic K-theory to Hochschild homology.

Proof. When k is a field, Keller constructs a natural transformation $\text{HH}^{\text{McC}} \rightarrow \text{HH}^{\text{Kel}}$ between McCarthy's Hochschild homology of exact categories and Keller's Hochschild homology [Kel99, Theorem 1.13]. Trace maps between algebraic K-theory of exact categories and Keller's cyclic homology

$$K \to \mathrm{HH}^{\mathrm{McC}} \to \mathrm{HH}^{\mathrm{Kel}}$$
 (4.4.1)

are then constructed by composing the transformation $\text{HH}^{\text{McC}} \rightarrow \text{HH}^{\text{Kel}}$ with the trace maps $K \rightarrow \text{HH}^{\text{McC}}$ defined by McCarthy [McC94, Sec. 4.4].

Let $V_k^G: GBornCoarse \to Add$ be the functor (2.1.3), seen as a functor to the category of small exact categories (by equipping them with the split exact structure). Then, composition of the transformation $K \to HH^{Kel}$ (4.4.1) with the functor V_k^G , induces a transformation $K\mathcal{X}_k^G \to \mathcal{X}HH_k^G$ of equivariant coarse homology theories, as equivariant coarse algebraic K-homology $K\mathcal{X}_k^G$ and equivariant coarse Hochschild homology $\mathcal{X}HH_k^G$ are defined as the algebraic K-theory and Keller's Hochschild homology, respectively, of the k-linear category of controlled objects $V_k^G(X)$. In the same way, we have a transformation to equivariant coarse cyclic homology.

When restricted to k-linear categories with split exact structure, McCarthy's and Keller's cyclic homologies provide equivalent complexes (in the sense of Lemma 3.4.4 and Remark 3.4.5), hence equivalent spectra after applying the Eilenberg-MacLane correspondence \mathcal{EM} (1.5.5). In particular, when X is the G-bornological coarse space $G_{\text{can,min}}$, the induced map

$$K\mathcal{X}_k^G(G_{\operatorname{can,min}}) \to \mathcal{X}\operatorname{HH}_k^G(G_{\operatorname{can,min}})$$

is the classical Dennis trace map $K(k[G]) \to HH(k[G])$ by McCarthy's agreement result [McC94, Sec. 4.5], by Proposition 2.3.6 and Remark 2.3.7, and by Proposition 4.1.4.

A symmetric monoidal natural transformation is a natural transformation between lax symmetric monoidal functors that respects the symmetric monoidal structures.

Corollary 4.4.2. The transformation $K\mathcal{X}_k^G \to \mathcal{X}HH_k^G$ is symmetric monoidal.

Proof. Both functors $K\mathcal{X}_k^G$ and $\mathcal{X}HH_k^G$ admit lax symmetric monoidal refinements by Theorem 2.3.4 and Proposition 3.5.3. By [McC94, Prop. 4.4.3 & Sec. 4.5], the trace map $K \to HH^{McC}$ is lax symmetric monoidal, and the transformation $K\mathcal{X}_k^G \to \mathcal{X}HH_k^G$ respects the symmetric monoidal structures and is symmetric monoidal.

Consider the transformations $K\mathcal{X}_k^G \to \mathcal{X}HH_k^G$ and $\mathcal{EM} \circ \Phi_{\mathcal{X}HH_k^G} \colon \mathcal{X}HH_k^G \to \mathcal{X}H_k^G$ of spectra-valued equivariant coarse homology theories. By combining Remark 2.3.7, Proposition 4.1.4, Example 1.5.7, Proposition 4.4.1 and Theorem 4.3.8, we get the following evaluations:

where the map $K(k[G]) \to \operatorname{HH}(k[G])$ is the Dennis trace map and the induced map $\operatorname{HH}(k[G]) \to H(G;k)$ can be described as follows:

- 1. the k-algebra k[G] can be seen as a category with a single object. The inclusion of this category in the category of finitely generated free k[G]-modules induces an equivalence of cyclic sets between the associated additive cyclic nerves.
- 2. By Proposition 2.3.6, the category of finitely generated free k[G]-modules is equivalent to the category $V_k^G(G_{\text{can,min}})$ and this equivalence of additive categories induces an equivalence between the associated additive cyclic nerves; then, we get an equivalence of cyclic sets $\operatorname{CN}(k[G]) \to \operatorname{CN}(V_k^G(G_{\text{can,min}}))$.
- 3. We apply the map φ of Definition 4.3.3 to the additive cyclic nerve associated to the category $V_k^G(G_{\text{can,min}})$;
- 4. by [BEKW17, Prop. 7.5] there is an equivalence $\psi \colon \mathcal{X}C^G(G_{\operatorname{can,min}}) \to C_*(G;k)$ of chain complexes, where $\mathcal{X}C^G(G_{\operatorname{can,min}})$ is the equivariant (\mathbf{Ch}_{∞} -valued) coarse ordinary homology of X and $C_*(G;k)$ is the chain complex computing the homology of G with k-coefficients.
- 5. All the maps extend to morphisms of mixed complexes, and their composition (after localization and application of the Eilenberg-MacLane correspondence \mathcal{EM}) yields the map $HH(k[G]) \to H(G;k)$ in the diagram.

We believe that further studies of the natural transformation

$$K\mathcal{X}_k^G \to \mathcal{X}\mathrm{HH}_k^G \xrightarrow{\mathcal{EM} \circ \Phi_{\mathcal{X}\mathrm{HH}_k^G}} \mathcal{X}\mathrm{H}_k^G$$

from equivariant coarse algebraic K-homology to (the spectra-valued) coarse ordinary homology, may be useful for a better understanding of equivariant coarse algebraic K-homology and for detecting coarse K-theory classes.

4.5 A Segal-type localization theorem for coarse Hochschild homology

In this section we apply the abstract localization results of Section 1.4 to coarse Hochschild and cyclic homology; we also assume that the field k is the field of complex numbers $k = \mathbb{C}$. Hence, the additive category with strict *G*-action of Definition 2.1.2 is the category $\operatorname{Vect}_{\mathbb{C}}^{\mathrm{f.d.}}$ of finite dimensional \mathbb{C} -vector spaces. For simplicity of notation, we will omit the subscript \mathbb{C} in the symbol $\mathcal{X}\operatorname{HH}^{G}_{\mathbb{C}}$ denoting equivariant coarse Hochschild homology.

In this section G is always supposed to be a finite group.

By Remark 1.1.28, the one point space $* \in GBornCoarse$ is a commutative algebra object in the category GBornCoarse. By Proposition 3.5.3, equivariant coarse Hochschild homology is a lax symmetric monoidal functor, hence the evaluation at the one point space

$$R_{\mathbb{C}} \coloneqq \mathcal{X} \mathrm{HH}^{G}(*)$$

is a commutative algebra object in the ∞ -category of chain complexes and the functor \mathcal{X} HH^G refines to a $\mathbf{Mod}(R_{\mathbb{C}})$ -valued equivariant coarse homology theory (where $\mathbf{Mod}(R_{\mathbb{C}})$) is the cocomplete stable ∞ -category of modules over $R_{\mathbb{C}}$).

The functor $\mathcal{X}HH^G$, being a coarse homology theory, factors through the category of motivic coarse spaces over the Yoneda functor $\operatorname{Yo}_G : \operatorname{\mathbf{GBornCoarse}} \to \operatorname{\mathbf{GSpc}} \mathcal{X}$ (1.2.3). We denote with the same symbol $\mathcal{X}HH^G$ the factorization $\mathcal{X}HH^G : \operatorname{\mathbf{GSpc}} \mathcal{X} \to \operatorname{\mathbf{Mod}}(R_{\mathbb{C}})$, which is still lax symmetric monoidal (Yo_G being lax symmetric monoidal as described in Section 1.2).

Remark 4.5.1. The twist of \mathcal{X} HH^G by a *G*-bornological coarse space *T*, as in Definition 1.2.5, gives an equivariant $\mathbf{Mod}(R_{\mathbb{C}})$ -valued coarse homology theory, *i.e.*, an element \mathcal{X} HH^G_T of $\mathbf{Fun}^{\mathrm{colim}}(G\mathbf{Spc}\mathcal{X}.\mathbf{C})$.

Remark 4.5.2. Equivariant coarse Hochschild homology is defined as the Hochschild homology of the category $V_{\mathbb{C}}^G(X)$. When X is a point, the category $V_{\mathbb{C}}^G(*)$ is equivalent to the category of finitely generated free $\mathbb{C}[G]$ -modules that are of finite dimension over \mathbb{C} . Hence, by Theorem 3.2.5, Hochschild homology of this category is equivalent to the Hochschild homology of $\mathbb{C}[G]$. An easy computation shows that the Hochschild homology of the group algebra $\mathbb{C}[G]$ is given by the representation ring R(G) of G and it is concentrated in degree 0. It then follows that $H_*(R_{\mathbb{C}})$ is isomorphic to the representation ring R(G).

Let \mathfrak{p} be a prime ideal of R(G). Then, we can form the lax symmetric monoidal localization functor (a Bousfield localization)

$$(-)_{\mathfrak{p}} \colon \mathbf{Mod}(R_{\mathbb{C}}) \to L_{\mathfrak{p}}\mathbf{Mod}(R_{\mathbb{C}})$$

generated by the set of morphisms $(f \colon R_{\mathbb{C}} \to R_{\mathbb{C}})_{f \in R_{\mathbb{C}} \setminus \mathfrak{p}}$ and a lax symmetric monoidal functor

$$\operatorname{Mod}(R_{\mathbb{C}}) \to \operatorname{Mod}(R_{\mathbb{C}})$$

sending an $R_{\mathbb{C}}$ -module M to the localization $M_{\mathfrak{p}}$ (see [BCa, Sec. 2.3] or [BCL18]). For every G-bornological coarse space X, the twisted equivariant coarse Hochschild homology $\mathcal{X}HH^G_T(X)$ is an $R_{\mathbb{C}}$ -module, and we denote by $\mathcal{X}HH^G_{T,\mathfrak{p}}(X)$ its localization at the prime ideal \mathfrak{p} of R(G) (by using the isomorphism between R(G) and the homology of $R_{\mathbb{C}}$).

We can apply the construction of Remark 1.4.7 to the functor $\mathcal{X}HH^G_{T,\mathfrak{p}}$, getting the functor

$$\underbrace{\mathcal{X}\mathrm{HH}_{T,\mathfrak{p}}^{G}}: G\mathbf{Orb} \to \mathbf{Fun}^{\mathrm{colim}}(\mathbf{Spc}\mathcal{X}, \mathbf{Mod}(R_{\mathbb{C}})).$$
(4.5.1)

We now proceed with proving the coarse localization theorem for coarse Hochschild homology.

Recall that a family of subgroups is invariant under conjugation and taking subgroups (Definition 1.3.16). Let γ be a conjugacy class of G. Then, the family $\mathcal{F}(\gamma) = \{H < G \mid H \cap \gamma = \emptyset\}$ is a family of subgroups (see Definition 1.4.19 and Lemma 1.4.20).

Definition 4.5.3. Let γ be a conjugacy class of the group G. The ideal of R(G) given by the representations ρ with $\operatorname{tr}(\rho(g)) = 0$ for all $g \in \gamma$ is denoted by (γ) .

Remark 4.5.4. We observe that the ideal (γ) of Definition 4.5.3 is a prime ideal of the representation ring R(G).

Let H be a subgroup in $\mathcal{F}(\gamma)$. Then Segal [Seg68b] has shown that there exists an element η in the representation ring R(G) of G such that:

- 1. $\eta_{|H} = 0$ and
- 2. $\operatorname{tr}(\eta(g)) \neq 0$ for all g in γ .

Hence, if H is any subgroup of G, we can form the localization $R(H)_{(\gamma)}$ of the R(G)-module R(H) at the prime ideal (γ) .

Let **C** be a stable ∞ -category. Recall (Definition 1.4.14) that a functor $E: \mathbf{GOrb} \to \mathbf{C}$ vanishes on the family of subgroups $\mathcal{F}(\gamma)$ of G if E(S) is equivalent to 0 for all S in $G_{\mathcal{F}(\gamma)}\mathbf{Orb}$.

Lemma 4.5.5. Let $\mathcal{X}HH_{T,(\gamma)}^G$ be the functor (4.5.1) localized at the prime ideal (γ) associated to the conjugacy class γ of G. Then, the functor $\mathcal{X}HH_{T,(\gamma)}^G$ vanishes on the family $\mathcal{F}(\gamma)$ of Definition 1.4.19.

Proof. We want to show that

$$\mathcal{X}\mathrm{HH}^{G}_{(\gamma)}((G/K)_{\min,\max}\otimes X)\simeq 0$$

for every K in $\mathcal{F}(\gamma)$ and every X in G**BornCoarse**. We first need to explicitly understand the R(G)-module structure of \mathcal{X} HH^G(X) for a G-bornological coarse space. We proceed as in [BCb, Prop. 4.11 & 4.15].

The module structure is induced by the morphism of C-linear categories

$$V^G_{\mathbb{C}}(*) \otimes V^G_{\mathbb{C}}(X) \to V^G_{\mathbb{C}}(X)$$

that sends $((V,\pi), (H,\rho))$ to $(V \otimes H, \pi \otimes \rho)$. Here (V,π) is a representation of G on a finitely dimensional \mathbb{C} -vector spaces, and (H,ρ) is an equivariant X-controlled $\mathbf{Vect}^{\mathrm{f.d.}}_{\mathbb{C}}$ -object. This construction extends to morphisms in the natural way and we get an exact functor between additive categories

$$(V,\pi)\otimes (-): V^G_{\mathbb{C}}((G/K)_{min,max}\otimes X) \to V^G_{\mathbb{C}}((G/K)_{min,max}\otimes X)$$
.

~

We assume now that (V^{\pm}, π^{\pm}) are two representations of G such that there exists an isomorphism $W: (V^+, \pi^+)_{|K} \to (V^-, \pi^-)_{|K}$. Then there is a controlled isomorphism [BCb, Prop. 4.15]

$$\widetilde{W}_{(H,\rho)}: (V^+ \otimes H, \pi^+ \otimes \rho) \to (V^- \otimes H, \pi^- \otimes \rho)$$

providing a natural isomorphism of functors

The existence of the isomorphism \tilde{W} implies that the multiplication by the classes $[V^+, \pi^+]$ and $[V^-, \pi^-]$ in $R(G) \cong H_*(R_{\mathbb{C}})$ on the homology of $\mathcal{X}\operatorname{HH}^G((G/K)_{\min,\max} \otimes X)$ coincide. Hence, if η is an element of R(G) which restricts to 0 in K, then η acts trivially on the homology of $\mathcal{X}\operatorname{HH}^G((G/K)_{\min,\max} \otimes X)$.

Let now H be a subgroup of G in $\mathcal{F}(\gamma)$. Let $\eta \in R(G)$ be the representation constructed by Segal satisfying (1) and (2). By (2), the multiplication by η acts as an isomorphism on the homology of $\mathcal{X}\mathrm{HH}^{G}_{(\gamma)}((G/H)_{min,max} \otimes X)$, but by (1) it vanishes. This implies that the homology of $\mathcal{X}\mathrm{HH}^{G}_{(\gamma)}((G/H)_{min,max} \otimes X)$ is zero, hence

 $\mathcal{X}\mathrm{HH}^G_{(\gamma)}((G/H)_{min,max}\otimes X)\simeq 0.$

The lemma is proved for any T by setting $X = T \otimes Y$ in the above argument. \Box

Let X be a G-bornological coarse space and γ a conjugacy class of G. In Definition 1.4.10, we have defined a category $G\mathbf{Sp}\mathcal{X}\langle G\mathbf{Orb}\otimes\mathbf{Sp}\mathcal{X}_{bd}\rangle$; we consider here also the category $G\mathbf{Sp}\mathcal{X}\langle G\mathbf{Orb}\otimes\mathbf{Sp}\mathcal{X}\rangle$ defined analogously [BCb, Def. 3.43]. Recall Definition 1.4.16 of X^{γ} and Definition 1.4.17 of a nice coarse space. Let T be an object in $G\mathbf{Sp}\mathcal{X}$. The following is the coarse localization theorem for coarse Hochschild homology.

Theorem 4.5.6. Let G be a finite group and let γ be a conjugacy class of G. Let X be a G-bornological coarse space and assume that the G-bornological coarse space of γ -fixed points X^{γ} is a nice subset of X. Assume one of:

- 1. G is finite and both X^{γ} and X belong to $GSp\mathcal{X}(GOrb \otimes Sp\mathcal{X})$;
- 2. $T = G_{\text{can,min}}$ and both the G-bornological coarse spaces X^{γ} and X belong to $G\mathbf{Sp}\mathcal{X}\langle G\mathbf{Orb}\otimes\mathbf{Sp}\mathcal{X}_{bd}\rangle$.

Then, the inclusion $X^{\gamma} \to X$ induces an equivalence

$$\mathcal{X}\mathrm{HH}^{G}_{T,(\gamma)}(X^{\gamma}) \to \mathcal{X}\mathrm{HH}^{G}_{T,(\gamma)}(X).$$

The same theorem holds if one replaces $\mathcal{X}HH^G$ with coarse cyclic homology $\mathcal{X}HC^G$ with \mathbb{C} -coefficients.

Proof. Assume (1); then, by [BCb, Cor. 3.43], we can replace the functor $\mathcal{X}HH^G_{T,(\gamma)}$ with its Bredon-style approximation. In the case (2), we use [BCb, Cor. 3.47], Lemma 4.2.3 and Example 1.4.11.

By Lemma 4.5.5, the functor $\underline{\mathcal{X}HH^G_{T,(\gamma)}}$ vanishes on $\mathcal{F}(\gamma)$. By Corollary 1.4.22, the morphism

$$\mathcal{X}\mathrm{HH}^{G}_{T,(\gamma)}(X^{\gamma}) \to \mathcal{X}\mathrm{HH}^{G}_{T,(\gamma)}(X)$$

is then an equivalence.

The localization theorem for coarse Hochschild homology implies a localization theorem for the associated equivariant Hochschild homology $\mathbf{HH}^G \coloneqq \mathcal{X}\mathrm{HH}^G_{k,G_{\mathrm{can,min}}} \circ \mathcal{O}^{\infty}_{\mathrm{hlg}}$:

Corollary 4.5.7. Let G be a finite group and let γ be a conjugacy class of G. Let W be a finite G CW-complex and let W^{γ} be the sub-complex of γ -fixed points. Then, the inclusion $W^{\gamma} \to W$ induces an equivalence

$$\mathbf{HH}^{G}_{(\gamma)}(W^{\gamma}) \to \mathbf{HH}^{G}_{(\gamma)}(W)$$

of chain complexes.

The proof is the same as [BCb, Cor. 4.14 & Rem. 4.20], where the same result has been obtained in the case of equivariant coarse topological and equivariant coarse algebraic K-homology. The result translates in the case of equivariant coarse cyclic homology as well.

89

90 4. A TRANSFORMATION TO COARSE ORDINARY HOMOLOGY AND FURTHER PROPERTIES
Appendix A

In this appendix we collect some information about additive categories, dg-categories, symmetric monoidal categories and cyclic objects. In this way we introduce the notation and language that we need in this thesis, and specially in Chapter 3 in order to define coarse Hochschild and cyclic homology.

In this appendix we let k be a commutative ring; we write \otimes for the tensor product over k when clear from the context.

A.1 Additive categories

In this section we recollect definitions and properties of additive categories, k-linear categories and exact categories.

We start with the definition of an additive category.

Definition A.1.1. [Mac71, Section VIII.2] An *additive category* is a category enriched over (the category of) abelian groups, that has a zero object and that admits a biproduct for each pair of objects.

Let \mathcal{A} and \mathcal{B} be additive categories and let $F: \mathcal{A} \to \mathcal{B}$ be a functor. The functor F is called *additive* if it preserves finite biproducts; hence it sends the 0-object to the 0-object and for every two objects A, B in \mathcal{A} there is a canonical isomorphism $F(A \oplus B) \cong F(A) \oplus F(B)$, where \oplus denotes the biproduct of \mathcal{A} . If, in addition, F preserves also finite limits and colimits, then it is called *exact*.

We denote by Add the category of small additive categories and exact functors.

Definition A.1.2. A k-category is a category enriched on k-modules; a k-linear category is a k-category that is also additive.

An example of k-linear categories is given by k-algebras: a k-algebra is a k-linear category with a single object. Morphisms of k-linear categories are additive functors that preserve the k-linear structure as well. We denote by \mathbf{Cat}_k the category of small k-linear categories and exact functors.

Let \mathcal{A} be a small additive category and let G be a group.

Definition A.1.3. The additive category \mathcal{A} is an *additive category with strict G-action* (on the right) if:

- 1. for every $g \in G$ we have an additive isomorphism $F_g: \mathcal{A} \to \mathcal{A}$;
- 2. if e is the unit element of G, then $F_e = id_{\mathcal{A}}$;
- 3. for every g, h in G we have $F_h \circ F_g = F_{gh}$ satisfying strictly associative relations.

The equalities in the definition express the strictness of the *G*-action. In general, they are replaced by natural isomorphisms subject to coherence associative relations. Observe that, if *BG* denotes the category with a single object and endomorphisms the given group G, then an additive category with strict *G*-action can be equivalently described as a functor $BG \rightarrow \mathbf{Add}$ from the category *BG* to the category of small additive categories.

Let \mathcal{A} be an additive category and denote by \oplus its biproduct.

Definition A.1.4. An additive category \mathcal{A} is called *flasque* if it admits an endofunctor $S: \mathcal{A} \to \mathcal{A}$ and a natural isomorphism $\mathrm{id}_{\mathcal{A}} \oplus S \cong S$.

Following Keller, we now recall the definition of an exact category [Kel96, Sec. 4]. Let \mathcal{A} be an additive category. A pair (i, p) of composable morphisms

$$A \xrightarrow{i} B \xrightarrow{p} C$$

in \mathcal{A} is called *exact* if *i* is a kernel for *p* and *p* is a cokernel for *i*. We use Keller's terminology: exact pairs $(i, p), A \xrightarrow{i} B \xrightarrow{p} C$, are called *conflations*, the morphism *i* is called an *inflation* and the morphism *p* a *deflation*.

Definition A.1.5. An *exact category* is an additive category equipped with a class of exact pairs satisfying the following axioms:

- 1. The identity morphism of the zero object is a deflation.
- 2. A composition of two deflations is a deflation.
- 3. A composition of two inflations is an inflation.
- 4. The pull-back of a deflation exists and is a deflation.
- 5. The push-out of an inflation exists and is an inflation.

These set of axioms are equivalent to the Quillen's axioms [Qui73], as proved by Keller [Kel90]. If \mathcal{A} and \mathcal{B} are exact categories, an additive functor $F: \mathcal{A} \to \mathcal{B}$ is *exact* if it sends conflations to conflations.

Example A.1.6. Let \mathcal{A} be an additive category. The family of all split short exact sequences is an exact structure on \mathcal{A} . We call this exact structure on \mathcal{A} the *split exact structure*.

Example A.1.7. Let \mathcal{A} be an additive category and let $Ch(\mathcal{A})$ be the category of chain complexes in \mathcal{A} , whose objects are chain complexes in \mathcal{A} and morphisms are chain maps. The category $Ch(\mathcal{A})$, equipped with the class of exact pairs (i, p) for which (i_n, p_n) is a split short exact sequence in \mathcal{A} for all $n \in \mathbb{Z}$, is an exact category.

We now give the definition of acyclic chain complexes:

Definition A.1.8. [Kel90] Let \mathcal{E} be an exact category. A chain complex $K \in Ch(\mathcal{E})$ is called *acyclic in degree* n if the boundary operator d_K^{n+1} factors as follows:



where p_{n+1} is a deflation, and i_{n+1} is an inflation. The complex K is called *acyclic* if it is acyclic in degree n for every n.

We denote by $\operatorname{Ch}^{b}(\mathcal{E})$ the category of bounded chain complexes over \mathcal{E} and by $\operatorname{Acy}^{b}(\mathcal{E})$ the category of bounded acyclic chain complexes over \mathcal{E} .

Remark A.1.9. Let \mathcal{E} be an exact category and let K be an acyclic chain complex. This means that we have conflations

$$Z_{n+1} \xrightarrow{i_{n+1}} K_n \xrightarrow{p_n} Z_n$$

for every $n \in \mathbb{N}$.

Let \mathcal{E} be a small exact k-linear category. Following [Kel96, Sec. 11], we now explain how to define the bounded derived category $\mathcal{D}^{b}(\mathcal{E})$. Let $\mathcal{K}^{b}(\mathcal{E})$ be the homotopy category of bounded chain complexes. The objects are the bounded chain complexes in \mathcal{E} and the morphisms are given by the chain maps up to chain homotopy. The category $\mathcal{K}^{b}(\mathcal{E})$ is a triangulated category. Let $\mathcal{K}^{b}_{acy}(\mathcal{E})$ be the sub-category of acyclic chain complexes, and observe that this is closed under taking cones and shifts in $\mathcal{K}^{b}(\mathcal{E})$. Hence, it is a full triangulated sub-category of $\mathcal{K}^{b}(\mathcal{E})$. The bounded derived category $\mathcal{D}^{b}(\mathcal{E})$ is then defined as the Verdier quotient

$$\mathcal{D}^b(\mathcal{E}) \coloneqq \mathcal{K}^b(\mathcal{E}) / \mathcal{K}^b_{acy}(\mathcal{E}).$$

Definition A.1.10. The category $\mathcal{D}(\mathcal{E})$ ($\mathcal{D}^b(\mathcal{E})$) is called the *(bounded) derived category* of \mathcal{E} .

We refer to [Kel96, Kel06] for more details.

A.2 Differential graded categories

In this section we recall the main notions concerning differential graded categories. We give the definitions of dg-algebras, dg-modules and dg-categories. We also review the notions of (strngly) pretriangulated categories and of Morita equivalences. Our main references are [Kel06, Toë11].

We start with the definition of a dg-algebra:

Definition A.2.1. A differential graded k-algebra A (shortly, a dg-algebra), is a \mathbb{Z} -graded k-algebra $A = \bigoplus_{p \in \mathbb{Z}} A^p$ endowed with a differential d either of degree -1 (chain complex convention) or of degree 1 (cochain complex convection) that satisfies the Leibniz rule

$$d(ab) = d(a)b + (-1)^p a d(b)$$

for all $a \in A^p$ and $p \in \mathbb{Z}$, and all $b \in A$.

A left dg-module M over a dg-algebra A is a left graded module $M = \bigoplus_{p \in \mathbb{Z}} M^p$ endowed with a differential d (of the same degree as the differential of A) such that $d(ma) = d(m)a + (-1)^p m d(a)$ for every $m \in M^p$ and $a \in A$. A morphism of dg-modules is a homogeneous morphism of degree 0 of the underlying graded modules commuting with the differentials.

The category of dg-modules over the dg-algebra A and morphisms of dg-modules is denoted by A-Mod.

A dg-category over a ring k is a category enriched on (the category of) chain complexes of k-modules. We spell it out:

Definition A.2.2. A small differential graded category \mathcal{A} (shortly, a dg-category) consists of the following data:

- a small set of objects $obj(\mathcal{A})$ (denoted \mathcal{A} as well);
- for each pair of objects A, B of \mathcal{A} , a chain complex of k-modules $\operatorname{Hom}_{\mathcal{A}}(A, B)$;
- for each triple of objects A, B, C in \mathcal{A} compositions

 $\operatorname{Hom}_{\mathcal{A}}(A, B) \otimes \operatorname{Hom}_{\mathcal{A}}(B, C) \to \operatorname{Hom}_{\mathcal{A}}(A, C)$

satisfying the associativity relations;

• for each object A of \mathcal{A} , a morphism $k \to \operatorname{Hom}_{\mathcal{A}}(A, A)$ satisfying the usual unit condition with respect to the compositions.

Example A.2.3. Every dg-algebra is a dg-category with a single object.

Definition A.2.4. Let \mathcal{A} and \mathcal{A}' be small dg-categories. A *dg-functor* $F: \mathcal{A} \to \mathcal{A}'$ consists of a map $F: \operatorname{obj}(\mathcal{A}) \to \operatorname{obj}(\mathcal{A}')$ of sets and, for each pair of objects A, B in \mathcal{A} , of a morphism $F(A, B): \operatorname{Hom}_{\mathcal{A}}(A, B) \to \operatorname{Hom}_{\mathcal{A}'}(F(A), F(B))$ of chain complexes satisfying the usual unit and composition conditions.

We denote by \mathbf{dgcat}_k the category of small dg-categories (over k) and dg-functors.

Example A.2.5. An additive category \mathcal{A} is a dg-category: for every object A in \mathcal{A} , $\operatorname{Hom}_{\mathcal{A}}(A, A)$ is a chain complex concentrated in degree 0. This gives a functor

$$\iota \colon \mathbf{Add} \to \mathbf{dgcat} \tag{A.1}$$

that sends an additive category to the corresponding dg-category.

Example A.2.6. Let \mathcal{E} be a small k-linear exact category. The category $\operatorname{Ch}(\mathcal{E})$ of chain complexes over \mathcal{E} has the structure of a dg-category: if K and K' are two chain complexes, then $\operatorname{Hom}_n(K, K')$ consists of the homogeneous morphisms f of degree n of graded objects: $(f: K_p \to K'_{p+n})_p$. The differential d_n : $\operatorname{Hom}_n(K, K') \to \operatorname{Hom}_{n-1}(K, K')$ is defined by the Leibniz rule

$$d(f) \coloneqq d_{K'} \circ f - (-1)^n f \circ d_K.$$

We denote by $\operatorname{Ch}_{dg}(\mathcal{E})$ this dg-structure of the category of chain complexes over \mathcal{E} . Analogously, the category of bounded chain complexes over \mathcal{E} is a dg category that we denote by $\operatorname{Ch}_{dg}^{b}(\mathcal{E})$.

The category $\operatorname{Acy}^{b}(\mathcal{E})$ of bounded acyclic chain complexes (see Definition A.1.8) is a sub-category of the category of bounded chain complexes; we denote by $\operatorname{Acy}_{dg}^{b}(\mathcal{E})$ the category of bounded acyclic chain complexes with the induced dg-structure.

Example A.2.7. If \mathcal{A} is a dg-category, then the dg-category \mathcal{A}^{op} defined as the category with the same objects as \mathcal{A} and morphisms $\text{Hom}_{\mathcal{A}^{\text{op}}}(A, B) := \text{Hom}_{\mathcal{A}}(B, A)$, is a dg-category.

We have defined dg-modules over a dg-algebra; we now generalize the definition of dg-modules to dg-categories.

Definition A.2.8. Let \mathcal{A} be a dg-category. A left (right) *dg-module* over \mathcal{A} is a dg-functor

$$\mathcal{A}^{\mathrm{op}} \to \mathrm{Ch}_{dq}(k)$$

where $Ch_{dq}(k)$ is the dg-category of chain complexes over the dg-algebra k.

There is a natural category of dg-modules over a dg-category \mathcal{A} , whose objects are the dg-modules over \mathcal{A} and whose morphisms are the natural transformations of dg-functors $F: M \to N$ such that $F(A): M(A) \to N(A)$ is a morphism of complexes for every object A of \mathcal{A} .

Remark A.2.9. When \mathcal{A} is a dg-category with one object (*i.e.*, a dg-algebra), the category of modules over \mathcal{A} just described is equivalent to the category of dg-modules (over a dg-algebra) defined in the text after Definition A.2.1.

Let \mathcal{A} be a dg-category and let M, N be two dg-modules over \mathcal{A} . Then, the evaluations of M and N at an object $A \in \mathcal{A}$ are chain complexes.

Definition A.2.10. A morphism of dg-modules $M \to N$ is a *quasi-isomorphism* if it induces a quasi-isomorphism of chain complexes $M(A) \to N(A)$ for every object A in A.

Remark A.2.11. The category of dg-modules (over a dg-algebra or a dg-category) admits two Quillen model structures where the weak equivalences are the object-wise quasi-isomorphisms of dg-modules; these are the injective and the projective model structure induced from the injective and projective model structure on chain complexes, respectively. We remark that the category of dg-modules over a dg-algebra, equipped with the projective model structure (hence the fibrations are the object-wise epimorphisms), is a combinatorial model category; see, for example [Coh, Rem. 2.14].

In analogy with chain complexes, we can define shifts and cones of objects in a dg-category, as we now explain.

Let \mathcal{A} be a dg-category and let A be an object of \mathcal{A} . The *n*-translation of A, for n an integer in \mathbb{Z} , is an object A[n] of \mathcal{A} representing the dg-functor

$$B \mapsto \operatorname{Hom}_{\mathcal{A}}(B, A)[n]$$

where $\operatorname{Hom}_{\mathcal{A}}(B, A)[n]$ is the *n*-shifted chain complex.

If A and B are objects of a dg-category \mathcal{A} , and $\varphi \colon A \to A'$ is a morphism of \mathcal{A} , then its cone cone(φ) is the object of \mathcal{A} that represents the dg-functor

$$B \mapsto \operatorname{cone}(\operatorname{Hom}_{\mathcal{A}}(B, A) \xrightarrow{\operatorname{Hom}(B,\varphi)} \operatorname{Hom}_{\mathcal{A}}(B, A'))$$

where the cone between the Hom-complexes here denotes the usual mapping cone of chain complexes.

Definition A.2.12. [BV08, Dri04] The dg-category \mathcal{A} is strongly pretriangulated if it admits a zero object 0 and the translations A[n] and cones $\operatorname{cone}(\varphi)$ exist for every object $A \in \mathcal{A}$, $n \in \mathbb{Z}$, and morphism φ .

For every dg-category \mathcal{A} there exist a dg-category \mathcal{A}^{pretr} called the *pretriangulated hull* of \mathcal{A} and a universal dg-functor $\mathcal{A} \to \mathcal{A}^{pretr}$, *i.e.*, a dg-functor inducing an equivalence $\operatorname{Hom}_{\operatorname{dgcat}}(\mathcal{A}, \mathcal{B}) \simeq \operatorname{Hom}_{\operatorname{dgcat}}(\mathcal{A}^{pretr}, \mathcal{B})$ for every strong pretriangulated dg-category \mathcal{B} [Kel06, Sec. 4.5] and [BK91].

We observe here that a weaker notion, namely of a pretriangulated category, is usually considered (see *e.g.*, [Coh, Kel06]). However, the pretriangulated hull \mathcal{A}^{pretr} of \mathcal{A} is always strongly pretriangulated [BK91, Dri04].

Example A.2.13. [BV08, BK91] If \mathcal{A} is an additive category, seen as a dg-category by $\iota(\mathcal{A})$ (A.1), then its pretriangulated hull \mathcal{A}^{pretr} is the dg-category $\mathrm{Ch}^{b}_{\mathrm{dg}}(\mathcal{A})$ of bounded chain complexes in \mathcal{A} .

If \mathcal{A} is a dg-category, we can define an associated derived category:

Definition A.2.14. [Kel06, Sec. 3.2] Let \mathcal{A} be a dg-category. The *derived category* $\mathcal{D}(\mathcal{A})$ of \mathcal{A} is the localization of the category of dg-modules over \mathcal{A} at the class of quasi-isomorphisms.

The objects of $\mathcal{D}(\mathcal{A})$ are the dg-modules over \mathcal{A} and the morphisms are obtained from morphisms of dg-modules by inverting the quasi-isomorphisms. It is a triangulated category with shift functor induced by the 1-translation and triangles coming from short exact sequences of complexes.

Let \mathcal{A} and \mathcal{B} be two small dg-categories. A dg-functor $F: \mathcal{A} \to \mathcal{B}$ is called a *Morita* equivalence if it induces an equivalence of derived categories. For a precise definition of Morita equivalences we refer to [Kel06, Sec. 3.8], or [Coh, Def. 2.29]. The most important example to us is the following:

Example A.2.15. [Kel06, Sec. 4.6] Let \mathcal{A} be a dg-category. Then, the canonical morphism $\mathcal{A} \to \mathcal{A}^{pretr}$ in the pretriangulated hull is a Morita equivalence.

Theorem A.2.16. [Tab05, Thm. 5.1] The category dgcat_k of small dg-categories over k admits the structure of a combinatorial model category whose weak equivalences are the Morita equivalences.

For a description of fibrations and cofibrations we refer to [Tab05, Thm. 5.1], or [Kel06, Thm. 4.1].

The localization [Lur14, Def. 1.3.4.1] of the category of small dg-categories **dgcat** at the class of Morita equivalences is the ∞ -category of (idempotent-complete) dg-categories:

$$\mathbf{dgcat}_{\infty} \coloneqq \mathrm{N}(\mathbf{dgcat})[W_{\mathrm{Morita}}^{-1}]$$

where W_{Morita} denotes the class of Morita equivalences of small dg-categories.

Remark A.2.17. The Morita model structure on **dgcat** presents the ∞ -category of stable ∞ -categories [Coh, Fao17]. The bridge between the two theories is the dg-nerve functor N_{dg} that sends a dg-category to a stable ∞ -categories [Lur14, Sec. 1.3.1].

We refer to Robalo's thesis [Rob14, Sec. 6.1 & 6.2] for a comparison between the theory of dg-categories and the theory of stable ∞ -categories.

A.3 Symmetric monoidal structures

In this section we recall the definitions of a symmetric monoidal category and (lax) symmetric monoidal functors.

A symmetric monoidal structure on a category \mathcal{C} consists of a functor $\otimes_{\mathcal{C}} : \mathcal{C} \times \mathcal{C} \to \mathcal{C}$, a tensor unit $1_{\mathcal{C}}$, and the associator, symmetry and unit-transformations that must satisfy some compatibility relations. We review these relations in the following definition:

Definition A.3.1. [Mac71, Sec. VII.1.] A symmetric monoidal structure on a category C is given by the following data:

- 1. a bifunctor $(-\otimes_{\mathcal{C}} -): \mathcal{C} \times \mathcal{C} \to \mathcal{C}$,
- 2. an object $1_{\mathcal{C}}$ (the *tensor unit*),
- 3. a natural isomorphism (the associativity constraint)

$$\alpha^{\mathcal{C}}: (-\otimes_{\mathcal{C}} -) \circ ((-\otimes_{\mathcal{C}} -) \times \mathrm{id}_{\mathcal{C}}) \to (-\otimes_{\mathcal{C}} -) \circ (\mathrm{id}_{\mathcal{C}} \times (-\otimes_{\mathcal{C}} -)) ,$$

- 4. a natural isomorphism $\eta^{\mathcal{C}} : 1_{\mathcal{C}} \otimes_{\mathcal{C}} \rightarrow \mathrm{id}_{\mathcal{C}}$ (the unit constraint),
- 5. a natural isomorphism (the symmetry) $\sigma^{\mathcal{C}} : (- \otimes_{\mathcal{C}} -) \circ T \to (- \otimes_{\mathcal{C}} -)$, where $T : \mathcal{C} \times \mathcal{C} \to \mathcal{C} \times \mathcal{C}$ is the flip functor.

This data has to satisfy the following relations:

1. the pentagon relation;



2. the triangle relation;



3. the inverse relation; $\sigma_{C,C'} \circ \sigma_{C',C} = \mathrm{id}_{C' \otimes C}$,

4. the associativity coherence.

A symmetric monoidal category is a category with a symmetric monoidal structure.

The category \mathbf{Cat}_k of small k-linear categories has a symmetric monoidal structure:

Definition A.3.2. Let \mathcal{A} and \mathcal{B} be small k-linear categories. The *tensor product* $\mathcal{A} \otimes_{\mathbf{Cat}_k} \mathcal{B}$ is the k-linear category whose objects are pairs (A, B) of objects with A in \mathcal{A} and B in \mathcal{B} and whose morphisms are defined by

$$\operatorname{Hom}_{\mathcal{A}\otimes_{\operatorname{Cat}_k}\mathcal{B}}((A_0, B_0), (A_1, B_1)) \coloneqq \operatorname{Hom}_{\mathcal{A}}(A_0, A_1) \otimes_k \operatorname{Hom}_{\mathcal{B}}(B_0, B_1).$$

Analogously, when k is a field, the category of small dg-categories has a natural symmetric monoidal structure:

Definition A.3.3. Let \mathcal{A} and \mathcal{B} be small dg-categories over k. The *tensor product* $\mathcal{A} \otimes_{\mathbf{dgcat}_k} \mathcal{B}$ is the dg-category whose objects are pairs (A, B) of objects with A in \mathcal{A} and B in \mathcal{B} and whose morphisms are defined by

$$\operatorname{Hom}_{\mathcal{A}\otimes_{\operatorname{dgcat}}\mathcal{B}}((A_0, B_0), (A_1, B_1) \coloneqq \operatorname{Hom}_{\mathcal{A}}(A_0, A_1) \otimes_k \operatorname{Hom}_{\mathcal{B}}(B_0, B_1),$$

where \otimes_k denotes the tensor product of chain complexes over k.

When k is not a field, the tensor product just described does not respect, in general, the Morita model structure on the category \mathbf{dgcat}_k of Theorem A.2.16 because it is not compatible with Morita equivalences. However, if k is a field, a dg-category over k is also flat, and the tensor product induces a symmetric monoidal structure on \mathbf{dgcat}_k

$$-\otimes_k -: \operatorname{dgcat}_k \times \operatorname{dgcat}_k \to \operatorname{dgcat}_k$$
 (A.1)

that preserves the Morita equivalences. We denote by $\mathbf{dgcat}_k^{\otimes}$ the symmetric monoidal category obtained in this way.

Definition A.3.4. [Mac71, Sec. VII.1.] Let $F : \mathcal{C} \to \mathcal{D}$ be a functor between symmetric monoidal categories. A symmetric monoidal structure on F is given by the following data:

- 1. an isomorphism $\epsilon^F : \mathbf{1}_{\mathcal{D}} \to F(\mathbf{1}_{\mathcal{C}}),$
- 2. a natural isomorphism $\mu^F : (-\otimes_{\mathcal{D}} -) \circ (F \times F) \to F \circ (-\otimes_{\mathcal{C}} -).$

This data has to satisfy the following relations:

- 1. associativity;
- 2. unitality;
- 3. symmetry.

We refer to [Mac71, Sec. VII.1.] for further details.

Remark A.3.5. If we weaken the assumption and only require that that ϵ^F and μ^F are natural transformations, then we get the definition of a *lax symmetric monoidal functor*.

Remark A.3.6. Let C be a symmetric monoidal category. Then, it gives rise to a category C^{\otimes} [Lur14, Constr. 2.0.0.1] and to a symmetric monoidal ∞ -category

$$N(\mathcal{C}^{\otimes}) \to N(\mathbf{Fin}_*)$$

[Lur14, Ex. 2.1.2.21] with the property that the underlying ∞ -category of $N(\mathcal{C}^{\otimes})$ is the nerve $N(\mathcal{C})$.

In Section 2.2, we need to refine functors with target the category of small additive categories **Add**. Let \mathcal{C} be an ordinary category. A functor $F : \mathcal{C} \to \mathbf{Add}$ gives rise to a functor between ∞ -categories $F_{\infty} : \mathcal{N}(\mathcal{C}) \to \mathbf{Add}_{\infty}$ in the natural way, *i.e.*, as the composition

$$F_{\infty}: \mathcal{N}(\mathcal{C}) \xrightarrow{\mathcal{N}(F)} \mathcal{N}(\mathbf{Add}) \to \mathcal{N}(\mathbf{Add})[W_{\mathbf{Add}}^{-1}] = \mathbf{Add}_{\infty}$$

where W_{Add} is the class of equivalences of additive categories.

Recall that a map of ∞ -operads [Lur14, Def. 2.1.2.7] can be thought of as a (lax) symmetric monoidal functor [Lur14, Def. 2.1.3.7] between the underlying categories.

Definition A.3.7. Let $F : \mathcal{C} \to \mathbf{Add}$ be a functor to the category of small additive categories. A *lax symmetric monoidal refinement* of F is a morphism

$$F^{\otimes}: \mathcal{N}(\mathcal{C}^{\otimes}) \to \mathbf{Add}_{\infty}^{\otimes}$$

of ∞ -operads that induces a functor equivalent to F_{∞} between the underlying ∞ categories.

A.4 Cyclic objects and the additive cyclic nerve

Following [Con83, Lod98, NS17], we recall in this section the definition of cyclic objects in a category C; then, we recall the definition of additive cyclic nerve of a dg-category.

A cyclic object X is a simplicial object with extra structure: in addition to face and degeneracy maps, there is an action of the cyclic group of order i + 1 acting on the objects X_i , for every *i*. There are different equivalent definitions of cyclic objects; we define them as controvariant functors on Connes' cyclic category, as we now explain.

We start with the definition of the paracyclic category Λ_{∞} [NS17, Appendix B].

The paracyclic category Λ_{∞} is a full subcategory of the category $\mathbb{Z}\mathbf{PoSet}$ of partially ordered sets equipped with a \mathbb{Z} -action and non-decreasing equivariant maps. The objects of Λ_{∞} are all the objects of $\mathbb{Z}\mathbf{PoSet}$ that are isomorphic to

$$\frac{1}{n}\mathbb{Z} \coloneqq \left\{\frac{k}{n} \mid k \in \mathbb{Z}\right\}$$

for $n \ge 1$, equipped with its natural ordering and the \mathbb{Z} action by addition. We use the notation $[n]_{\Lambda_{\infty}} := \frac{1}{n}\mathbb{Z}$.

There is an action of \mathbb{Z} on the morphism sets of Λ_{∞} : if τ is a generator of \mathbb{Z} and $f: [n]_{\Lambda_{\infty}} \to [m]_{\Lambda_{\infty}}$ is a morphism of Λ_{∞} , then $\tau(f): [n]_{\Lambda_{\infty}} \to [m]_{\Lambda_{\infty}}$ is defined as $\tau(f) = f + 1$.

Definition A.4.1. The cyclic category Λ is the full subcategory of Λ_{∞} with the same objects as Λ_{∞} ; the morphisms between two objects of Λ_{∞} are given by the morphisms of Λ_{∞} with the \mathbb{Z} -action by τ divided out.

It is a standard fact that the category Λ is self-dual [NS17, Appendix B]. If Δ is the category of totally ordered non-empty finite sets, there is a functor $\Delta \to \Lambda$ sending $[n] = \{0, \ldots, n\} \in \Delta$ to $[n+1]_{\Lambda}$ inducing faces and degeneracy maps. The cyclic operator in Λ descends from the \mathbb{Z} action by the generator τ described above. Then, the previous definition of Λ is equivalent to the following more explicit one:

Definition A.4.2. [Lod98, Def. 6.1.1] The cyclic category Λ has objects [n], for every $n \in \mathbb{N}$, and morphisms generated by:

- faces $\delta_i \colon [n-1] \to [n]$ for $i \leq n$;
- degeneracies $\sigma_j \colon [n+1] \to [n]$ for $j \leq n$;
- cyclic operators $\tau_n \colon [n] \to [n];$

satisfying the following relations:

(i)
$$\delta_j \delta_i = \delta_i \delta_{j-1}$$
 for $i < j$;
 $\sigma_j \sigma_i = \sigma_i \sigma_{j+1}$ for $i \le j$;
 $\sigma_j \delta_i = \begin{cases} \delta_i \sigma_{j-1} & \text{for } i < j \\ \mathrm{id}_{[n]} \sigma_{j-1} & \text{for } i = j, i = j+1 \\ \delta_{i-1} \sigma_j & \text{for } i > j+1 \end{cases}$

(ii) $\tau_n \delta_i = \delta_{i-1} \tau_{n-1}$ for $1 \le i \le n$ and $\tau_n \delta_0 = \delta_n$;

$$\tau_n \sigma_i = \sigma_{i-1} \tau_{n+1}$$
 for $1 \le i \le n$ and $\tau_n \sigma_0 = \sigma_n \tau_{n+1}^2$;

(iii) $\tau_n^{n+1} = \operatorname{id}_{[n]}.$

Definition A.4.3. Let C be a category. A *cyclic object* in C is a functor

$$X \colon \Lambda^{\mathrm{op}} \to \mathcal{C}.$$

Morphisms of cyclic objects are given by natural transformations of functors.

Notation A.4.4. We denote by X_n the value of X at [n].

There is a category of cyclic objects in C and morphisms of cyclic objects. When C is the category k-Mod of k-modules over a ring k, then a cyclic object in k-Mod is called a cyclic module. Hence, observe that a cyclic module C is a simplicial k-module endowed for all n with an action of the cyclic group of order n + 1 on C_n . Morphism of cyclic modules are morphisms of simplicial k-modules that commute with the cyclic structure.

Example A.4.5. Let k be a commutative ring with identity 1 and let A be a k-algebra. Then, for each $n \in \mathbb{N}$ define

$$Z_n(A) \coloneqq A^{\otimes_k n+1}$$

The following degeneracies s_i , face maps d_i and cyclic maps t

$$d_i(a_0, \dots, a_n) = \begin{cases} (a_0, a_1, \dots, a_i a_{i+1}, \dots, a_n) & \text{if } 0 \le i \le n-1 \\ (a_n a_0, a_1, \dots, a_n) & \text{if } i = n \end{cases}$$
$$s_i(a_0, \dots, a_n) = (a_0, a_1, \dots, a_i, 1, a_{i+1}, \dots, a_n) & \text{if } 0 \le i \le n \\ t_{n+1}(a_0, \dots, a_n) = (a_n, a_0, \dots, a_{n-1}) \end{cases}$$

make $Z_*(A)$ a cyclic module [Goo85, Sec II.1].

Example A.4.6. [McC94, Def. 2.1.1] Let \mathcal{A} be an additive category and k-Mod be the category of k-modules, k being a ring. The previous construction generalizes to additive categories, describing a cyclic module, called the *additive cyclic nerve*,

$$\operatorname{CN}_*(\mathcal{A}) \colon \Lambda^{\operatorname{op}} \to k\operatorname{-Mod}$$

that in degree n is given by:

$$\operatorname{CN}_n(\mathcal{A}) \coloneqq \bigoplus \operatorname{Hom}_{\mathcal{A}}(A_1, A_0) \otimes \operatorname{Hom}_{\mathcal{A}}(A_2, A_1) \otimes \cdots \otimes \operatorname{Hom}_{\mathcal{A}}(A_0, A_n)$$

where the sum runs over all the objects (A_0, A_1, \ldots, A_n) in \mathcal{A}^{n+1} and tensor products \otimes are tensor products over k. We give a description of the face maps, degeneracies and

cyclic maps:

$$d_{i}(f_{0} \otimes \dots \otimes f_{n}) = \begin{cases} f_{0} \otimes f_{1} \otimes \dots \otimes f_{i} \circ f_{i+1} \otimes \dots \otimes f_{n} & \text{if } 0 \leq i \leq n-1 \\ f_{n} \circ f_{0} \otimes f_{1} \otimes \dots \otimes f_{n} & \text{if } i = n \end{cases}$$
$$s_{i}(f_{0} \otimes \dots \otimes f_{n}) = \begin{cases} f_{0} \otimes f_{1} \otimes \dots \otimes f_{i} \otimes \operatorname{id}_{A_{i+1}} \otimes f_{i+1} \otimes \dots \otimes f_{n} & \text{if } 0 \leq i \leq n-1 \\ f_{0} \otimes f_{1} \otimes \dots \otimes f_{n} \otimes \operatorname{id}_{A_{0}} & \text{if } i = n \end{cases}$$
$$t(f_{0} \otimes \dots \otimes f_{n}) = (f_{n} \otimes f_{0} \otimes \dots \otimes f_{n-1})$$

The additive cyclic nerve is a covariant functor from the category of small k-linear categories to the category of cyclic k-modules.

By [McC94, Ex. 2.2.1], when applied to a k-algebra A seen as a k-linear category with a single object, the additive cyclic nerve $CN_*(A)$ is the cyclic module Z_*A of Example A.4.5.

We can do the same for dg-categories. For reference, we spell this out:

Definition A.4.7. The additive cyclic nerve of a differential graded k-linear category \mathcal{C} is the cyclic k-module defined for each n by

$$\operatorname{CN}_n(\mathcal{C}) \coloneqq \bigoplus \operatorname{Hom}_{\mathcal{C}}(C_1, C_0) \otimes \operatorname{Hom}_{\mathcal{C}}(C_2, C_1) \otimes \cdots \otimes \operatorname{Hom}_{\mathcal{C}}(C_0, C_n)$$

where the sum runs over all the objects (C_0, C_1, \ldots, C_n) in \mathcal{C}^{n+1} . The face and degeneracy maps, and the cyclic action, are defined as follows:

$$d_i(f_0 \otimes \dots \otimes f_n) = \begin{cases} f_0 \otimes f_1 \otimes \dots \otimes f_i \circ f_{i+1} \otimes \dots \otimes f_n & \text{if } 0 \le i \le n-1\\ (-1)^{n+\sigma} f_n \circ f_0 \otimes f_1 \otimes \dots \otimes f_n & \text{if } i=n \end{cases}$$
$$s_i(f_0 \otimes \dots \otimes f_n) = \begin{cases} f_0 \otimes f_1 \otimes \dots \otimes f_i \otimes \operatorname{id}_{C_{i+1}} \otimes f_{i+1} \otimes \dots \otimes f_n & \text{if } 0 \le i \le n-1\\ f_0 \otimes f_1 \otimes \dots \otimes f_n \otimes \operatorname{id}_{C_0} & \text{if } i=n \end{cases}$$
$$t(f_0 \otimes \dots \otimes f_n) = (-1)^{n+\sigma} (f_n \otimes f_0 \otimes \dots \otimes f_{n-1})$$

where $\sigma = (\deg f_n)(\deg f_{n-1} + \dots + \deg f_0).$ The differential b is defined as $b = (-1)^n d_n + \sum_{i=0}^{n-1} d_i.$

Observe that the definition of additive cyclic nerve for dg-categories generalizes both Example A.4.5 and Example A.4.6. In particular, when applied to an additive category, seen as a dg-category by means of the functor $\iota: \mathbf{Cat}_k \to \mathbf{dgcat}_k$ (A.1), we get (up to sign) the additive cyclic nerve for additive categories of Example A.4.6 (because the maps f_i of the above definition have only degree 0).

Bibliography

- [Ati64] M. F. Atiyah, Lectures on k-theory, 1964. mimeographed, Harvard. ↑xiii
- [BC17] U. Bunke and D.-C. Cisinski, A universal coarse k-theory, ArXiv e-prints (2017). https://arxiv.org/abs/1705.05080. ↑viii, 11, 44, 47, 53, 58
- [BCa] U. Bunke and L. Caputi, Controlled objects as a symmetric monoidal functor. In preparation. ↑xv, 29, 36, 37, 38, 39, 40, 42, 43, 45, 87
- [BCb] U. Bunke and L. Caputi, Localization for coarse homology theories. In preparation. ↑xiv, xv, 1, 19, 20, 21, 22, 23, 24, 25, 88, 89
- [BCL18] C. Braun, J. Chuang, and A. Lazarev, Derived localisation of algebras and modules, Adv. Math. 328 (2018), 555–622. MR3771137 ↑87
- [BE16] U. Bunke and A. Engel, Homotopy theory with bornological coarse spaces, ArXiv e-prints (2016). https://arxiv.org/abs/1607.03657. ↑vii, viii, 1, 2, 3, 4, 5, 8, 9, 11, 12, 26, 27, 44, 60
- [BE17a] U. Bunke and A. Engel, Coarse assembly maps, ArXiv e-prints (2017), available at 1706.02164. https://arxiv.org/abs/1706.02164. ↑viii
- [BE17b] U. Bunke and A. Engel, Coarse cohomology theories, ArXiv e-prints (2017), available at 1711.08599. https://arxiv.org/abs/1711.08599. ↑viii
- [BE17c] U. Bunke and A. Engel, The coarse index class with support, arXiv e-prints (2017), arXiv:1706.06959, available at 1706.06959. https://arxiv.org/abs/1706.06959. ↑viii
- [BEKW17] U. Bunke, A. Engel, D. Kasprowski, and C. Winges, Equivariant coarse homotopy theory and coarse algebraic k-homology, ArXiv e-prints (2017). https://arxiv.org/abs/1710.04935. ↑vii, viii, x, xi, xiii, xv, 1, 6, 7, 8, 9, 10, 11, 12, 13, 14, 15, 16, 17, 18, 19, 26, 27, 28, 29, 30, 31, 32, 33, 34, 35, 44, 45, 46, 47, 53, 58, 60, 61, 73, 75, 85
- [BEKW18] U. Bunke, A. Engel, D. Kasprowski, and C. Winges, *Injectivity results for coarse homology theories*, arXiv e-prints (2018), arXiv:1809.11079, available at 1809.11079. https://arxiv.org/abs/1809.11079. ↑vii, xiii, 1, 14, 16, 17, 18, 19, 44
 - [BK91] A. I. Bondal and M. M. Kapranov, Enhanced triangulated categories, Mathematics of the USSR-Sbornik 70 (1991), no. 1, 93. ↑96, 97
 - [BKW18] U. Bunke, D. Kasprowski, and C. Winges, Split injectivity of A-theoretic assembly maps, arXiv e-prints (2018), arXiv:1811.11864, available at 1811.11864. https://arxiv.org/abs/1811.11864. ↑viii, xiii, 11
 - [BLR07] A. Bartels, W. Lueck, and H. Reich, On the Farrell-Jones Conjecture and its applications, arXiv Mathematics e-prints (March 2007), math/0703548, available at math/0703548. https://arxiv.org/abs/math/0703548. ↑xiii, 13
 - [Blu17] A. Blumberg, Equivariant homotopy theory, 2017. https://web.ma.utexas.edu/users/a. debray/lecture_notes/m392c_EHT_notes.pdf. ↑18

- [BR07] A. Bartels and H. Reich, Coefficients for the farrell-jones conjecture, Advances in Mathematics 209 (2007), no. 1, 337 –362. ↑45
- [Bun18] U. Bunke, Coarse homotopy theory and boundary value problems, arXiv e-prints (2018), arXiv:1806.03669, available at 1806.03669. https://arxiv.org/abs/1806.03669. ↑viii
- [BV08] A. Beilinson and V. Vologodsky, A DG guide to Voevodsky's motives, Geom. Funct. Anal. 17 (2008), no. 6, 1709–1787. MR2399083 ↑96, 97
 - [Cis] D.-C. Cisinski, Higher categories and homotopical algebra. http://www.mathematik. uni-regensburg.de/cisinski/CatLR.pdf. ↑xvi, 51, 58
- [Coh] L. Cohn, Differential graded categories are k-linear stable infinity categories, available at 1308.2587. https://arxiv.org/abs/1308.2587. ↑96, 97
- [Con83] A. Connes, Cohomologie cyclique et foncteurs Extⁿ, C. R. Acad. Sci. Paris Sér. I Math. 296 (1983), no. 23, 953–958. MR777584 ↑100
- [CP97] M. Cárdenas and E. K. Pedersen, On the karoubi filtration of a category, K-theory 12 (1997), 165–191. ↑60, 61
- [CT11] D.-C. Cisinski and G. Tabuada, Non-connective K-theory via universal invariants, Compos. Math. 147 (2011), no. 4, 1281–1320. MR2822869 ↑ix
- [CT12] D.-C. Cisinski and G. Tabuada, Symmetric monoidal structure on non-commutative motives, J. K-Theory 9 (2012), no. 2, 201–268. MR2922389 ↑68
- [DGM13] B. I. Dundas, T. Goodwillie, and R. McCarthy, The local structure of algebraic k-theory, Vol. 18, Springer, 2013. ↑viii
 - [Dri04] V. Drinfeld, *DG quotients of DG categories*, J. Algebra **272** (2004), no. 2, 643–691. MR2028075 ↑96
- [Elm83] A. D. Elmendorf, Systems of fixed point sets, Trans. Amer. Math. Soc. 277 (1983), no. 1, 275–284. MR690052 ↑18, 21
- [Fao17] G. Faonte, Simplicial nerve of an A_{∞} -category, Theory Appl. Categ. **32** (2017), 31–52. MR3607208 \uparrow 52, 97
- [GHN17] D. Gepner, R. Haugseng, and T. Nikolaus, Lax colimits and free fibrations in ∞-categories, Doc. Math. 22 (2017), 1225–1266. MR3690268 ↑20
- [Goo85] T. G. Goodwillie, Cyclic homology, derivations, and the free loopspace, Topology 24 (1985), no. 2, 187 –215. ↑102
- [Hin16] V. Hinich, Dwyer-Kan localization revisited, Homology Homotopy Appl. 18 (2016), no. 1, 27–48. MR3460765 ↑69
- [Kar70] M. Karoubi, Foncteurs derivés et k-theorie, Springer Berlin Heidelberg, Berlin, Heidelberg, 1970. ↑61
- [Kas15] D. Kasprowski, On the k-theory of groups with finite decomposition complexity, Proc. London Math. Soc. 110 (2015), no. 3, 565–592. ↑60, 61
- [Kas87] C. Kassel, Cyclic homology, comodules, and mixed complexes, Journal of Algebra 107 (1987), no. 1, 195–216. ↑ix, 47, 48, 49, 50, 68, 69
- [Kel06] B. Keller, On differential graded categories, International Congress of Mathematicians 2 (2006), 151 –190. ↑xvi, 93, 94, 96, 97
- [Kel90] B. Keller, Chain complexes and stable categories, Manuscripta Mathematica 67 (1990), no. 1, 379–417. ↑92, 93
- [Kel96] B. Keller, Derived categories and their uses, Handbook of Algebra 1 (1996), 671 −701. ↑92, 93

- [Kel98] B. Keller, Invariance and localization for cyclic homology of dg algebras, Journal of Pure and Applied Algebra 123 (1998), no. 1, 223 –273. ↑51
- [Kel99] B. Keller, On the cyclic homology of exact categories, Journal of Pure and Applied Algebra 136 (1999), no. 1, 1 −56. ↑viii, ix, x, xv, 47, 51, 53, 54, 55, 56, 66, 83, 84
- [Lod98] J.-L. Loday, Cyclic homology, Grundlehren der mathematischen Wissenschaften, vol. 301, Springer Berlin Heidelberg, 1998. ↑viii, ix, 47, 49, 69, 100, 101
- [LR06] W. Lück and H. Reich, Detecting K-theory by cyclic homology, Proc. London Math. Soc. (3) 93 (2006), no. 3, 593–634. MR2266961 ↑76
- [Lue18] W. Lueck, Assembly Maps, arXiv e-prints (May 2018), arXiv:1805.00226, available at 1805. 00226. ↑xiii, 13
- [Lur09] J. Lurie, Higher topos theory, Annals of Mathematics Studies, vol. 170, Princeton University Press, Princeton, NJ, 2009. ↑xvi, 24
- [Lur14] J. Lurie, *Higher algebra*, 2014. Available at www.math.harvard.edu/lurie. ↑xvi, 27, 44, 51, 52, 56, 57, 58, 97, 100
- [Mac71] S. MacLane, Categories for the working mathematician, Springer-Verlag, New York-Berlin, 1971. Graduate Texts in Mathematics, Vol. 5. MR0354798 ↑20, 91, 98, 99, 100
- [McC94] R. McCarthy, The cyclic homology of an exact category, Journal of Pure and Applied Algebra 93 (1994), no. 3, 251 –296. ↑viii, x, 66, 83, 84, 102, 103
- [Mit01] P. D. Mitchener, Coarse homology theories, Algebr. Geom. Topol. 1 (2001), 271–297. ↑vii
- [Mos73] G. D. Mostow, Strong rigidity of locally symmetric spaces. (am-78), Princeton University Press, 1973. ↑vii
- [NS17] T. Nikolaus and P. Scholze, On topological cyclic homology, arXiv e-prints (July 2017), arXiv:1707.01799, available at 1707.01799. ↑100, 101
- [Qui73] D. Quillen, Higher algebraic k-theory, Lecture Notes in Mathematics 341 (1973), no. 1, 85–147. ↑92
- [Rob14] M. Robalo, Theorie homotopique motivique des espaces noncommutatifs (2014). PhD thesis. ↑97
- [Roe03] J. Roe, Lectures on coarse geometry, University Lecture Series, vol. 31, American Mathematical Society, Providence, RI, 2003. ↑vii
- [Roe93] J. Roe, Coarse cohomology and index theory on complete riemannian manifolds, Mem. Amer. Math. Soc. 104 (1993), no. 497. ↑vii
- [Roe96] J. Roe, Index theory, coarse geometry, and topology of manifolds, CBMS Regional Conference Series in Mathematics, vol. 90, Published for the Conference Board of the Mathematical Sciences, Washington, DC; by the American Mathematical Society, Providence, RI, 1996. MR1399087 ↑vii
- [RV17] H. Reich and M. Varisco, Algebraic K-theory, assembly maps, controlled algebra, and trace methods, arXiv e-prints (February 2017), arXiv:1702.02218, available at 1702.02218. ↑xiii, 13
- [Sch00] M. Schlichting, Délacage de la K-théorie des catégories exactes et K-groupes négatifs (2000). PhD thesis. ↑61
- [Sch04] M. Schlichting, Delooping the k-theory of exact categories, Topology 43 (2004), no. 5, 1089 -1103. ↑61
- [Sch06] M. Schlichting, Negative k-theory of derived categories, Mathematische Zeitschrift 253 (2006), no. 1, 97–134. ↑44
- [Sch11] M. Schlichting, Higher algebraic k-theory (after quillen, thomason and others), Springer Berlin Heidelberg, Berlin, Heidelberg, 2011. ↑59

- [Seg68a] G. Segal, Equivariant K-theory, Inst. Hautes Études Sci. Publ. Math. 34 (1968), 129–151. MR0234452 ↑xiii, xiv, 1, 19
- [Seg68b] G. Segal, The representation ring of a compact Lie group, Inst. Hautes Études Sci. Publ. Math. 34 (1968), 113–128. MR0248277 ↑19, 87
- [Shi07] B. Shipley, HZ-algebra spectra are differential graded algebras, Amer. J. Math. 129 (2007), no. 2, 351–379. MR2306038 ↑27
- [Tab05] G. Tabuada, Invariants additifs de dg-catégories, Internat. Math. Res. Notices 53 (2005), 3309–3339. ↑97
- [Tab07] G. Tabuada, *Theorie homotopique des DG-categories*, ArXiv e-prints (October 2007), available at 0710.4303. PhD thesis. ↑ix
- [Toë11] B. Toën, Lectures on dg-categories, Topics in algebraic and topological K-theory, 2011, pp. 243–302. MR2762557 ↑94

Acknowledgements

First and foremost, I would like to express my sincere gratitude to my advisers Prof. Dr. Ulrich Bunke and Prof. Dr. Clara Löh, for their support, the patience, the continuous motivation, the comments on the manuscript, the many helpful discussions, and in particular, for trusting in me since the very beginning. I especially thank Prof. Dr. Ulrich Bunke for his guidance, all the contributions of time and ideas, and for the deep support on this project. Besides my advisers, I would like to thank Dr. Alexander Engel, for the many discussions, and Prof. Dr. Denis-Charles Cisinski, for his fruitful and important clarifications; in particular, for pointing me out those results by Keller that eventually have become essential in the thesis.

I gratefully acknowledge the financial and academic support by the Graduiertenkolleg GRK 1692 "Curvature, Cycles, Cohomology" and all the members and staff of the University of Regensburg, and especially Frau Lindner. Among others, special thanks go to Dali, Michael, Kévin, Yuka, Koen and Johanna, and to the (non small) Italian community, to Enrico, Simone, Marta, Federico, Danny, Fritz, Celeste, but also to Martino and César, they have contributed immensely to my personal and professional time, becoming wonderful friends.

I would like to thank the Welcome Center and its staff, for welcoming and helping us, and especially for organizing those beautiful events that lead to so many friendships. My life in Regensburg was made enjoyable in large part due to the many friends and groups that became a significant part of my life. Besides the aforementioned, special thanks go to Ottavia, Vane and Celia, Rebecca and Yih, Holly and Greg, Mathilde and Cagla, Alice and Giulia, Ilknur and Raffaella, Lauisa, Sara and Madi. I am immensely grateful for all the time spent together, in Regensburg and abroad, for all the memorable trips, the long nights, the helles beer and the chats, the various dults and Halloweens, the stammtischs, the various coffees at 11, 14, 16, and sometimes also at 9 and 18, the Sunday's lunches and the delicious dinners, with pizza and Käsespätzle. Living in a different country might be in some occasions difficult and demotivating; their incredible support and their amazing encouragement has been of fundamental importance and they let me feel constantly at home.

I would especially like to thank Enrico, for being the very first friend that I met in this new city and for having shared with me way more then just an office and a flat. And also Edo and Antonio, for having been contributing immensely to my personal life in this last year.

I would like to thank all my friends spread around Europe, their long-distance support never ended. In particular, many thanks to Carlo, Peppe, Ivo, Giuliano, Silvia, Elena, Giovanni, Sanna, Dariuzzo, Isa, Daniele, Roberta, Benny, Caiulo, Santa, Luciana, Frapi, Arianna, Vincenzo, Agnese, Ilaria and Antonio and all the others that in this rush hours I might have forgotten; in particular, countless thanks go to Sabino and Cristina.

Last but not least, to my family. There are no words for describing how much deeply grateful I am; I dedicate this thesis to them. Grazie mamma, papà, Maria e Piera!